Decision Makers' Use of Intuition at the Front End of Innovation

Olli Hyppänen





DOCTORAL DISSERTATIONS

Decision Makers' Use of Intuition at the Front End of Innovation

Olli Hyppänen

A doctoral dissertation completed for the degree of Doctor of Science in Technology to be defended, with the permission of the Aalto University School of Science, at a public examination held at the lecture hall AS1 of the school on 22nd of March 2013 at 12 noon.

Aalto University School of Science Department of Industrial Engineering and Management

Supervising professor

Professor Karlos Artto

Thesis advisor

Professor Miia Martinsuo, Tampere University of Technology, Finland

Preliminary examiners

Professor Juha-Antti Lamberg, University of Jyväskylä, Finland

Professor Harri Haapasalo, University of Oulu, Finland

Opponent

Professor Juha Laurila, University of Turku, Finland

Aalto University publication series **DOCTORAL DISSERTATIONS** 30/2013

© Olli Hyppänen

ISBN 978-952-60-5029-4 (printed) ISBN 978-952-60-5030-0 (pdf) ISSN-L 1799-4934 ISSN 1799-4934 (printed) ISSN 1799-4942 (pdf) http://urn.fi/URN:ISBN:978-952-60-5030-0

Pictures: Cover image: Olli Hyppänen

Unigrafia Oy Helsinki 2013

Finland

Publication orders (printed book): olli.hyppanen@gmail.com



441 697 Printed matter



Author	
Olli Hyppänen	
Name of the doctoral dissertation	
Decision Makers' Use of Intuition at the Fr	ont End of Innovation
Publisher School of Science	
Unit Department of Industrial Engineering	and Management
Series Aalto University publication series	DOCTORAL DISSERTATIONS 30/2013
Field of research International Project-or	riented Business
Manuscript submitted 6 November 2012	Date of the defence 22 March 2013
Permission to publish granted (date) 15	January 2013 Language English
⊠ Monograph	sertation (summary + original articles)

Abstract

With regard to creating new products, the front end of innovation has been judged to be the most important place for decision making. This dissertation studies the decision makers' use of intuition at the innovation front end. The findings in managerial decision making research suggest that decision makers most often use intuition in uncertain situations. Innovation front end is one environment with high uncertainty. The use of intuition in decision making has not been extensively researched at the innovation front end context and the empirical research data in particular is almost nonexistent. Existing research in innovation front end decision making has concentrated on building traditional normative models to deal with uncertainty.

Empirical data for this study was gathered by interviews and a survey questionnaire from 4 ICT companies. Altogether 19 experienced decision makers took part in in-depth interviews. Survey data consisted of 86 survey responses from experienced and inexperienced decision makers. The grounded theory method was used to build a framework about the use of intuition in front end decision making, which was the main contribution of this research. The empirical data was analyzed by looking the data using several encounters which focus on separate aspects over the studied phenomena. Survey data was analyzed using statistical methods.

The findings of this study suggest that intuition plays a major role for experienced decision makers when making innovation front end decisions. Four facets in revealing the use of intuition were discovered: Symptoms, preceding enablers, simultaneous enablers and safeguards. This research also shows what kind of approaches decision makers take when using intuition. Seven different approaches were found: Drifter, thinker, negotiator, tester, discoverer, believer and seer. The difference in the use of intuition between experienced and inexperienced decision makers was also studied. Both (experienced and inexperienced) decision maker groups have same level of belief over the use of intuition in decision making and they also use it in same extent.

This study contributes to the intuition in decision making and innovation front end literature by describing what approaches decision makers have in using intuition and by establishing new methods to study the use of intuition in decision making. The four facets which reveal the use of intuition establish a concrete way to study the use of intuition. The approaches, what decision makers have, bring more understanding on decision makers' intuition use and how that may reflect on managerial activities. The findings also stress that the role intuition plays should receive greater acknowledgement in innovation research.

Keywords Intuition, decision making, uncertainty, front end, innovation, innovation management, New Product Development

ISBN (printed) 978-952-60	-5029-4	ISBN (pdf) 978-	952-60-5030-0	
ISSN-L 1799-4934	ISSN (printed)	1799-4934	ISSN (pdf)	1799-4942
Location of publisher Esp	oo Locat	ion of printing	Helsinki	Year 2013
Pages 204	urn	http://urn.fi/UI	RN:ISBN:978-952	-60-5030-0



🛛 Monografia 🛛 Yhdistelmävä	itöskirja (yhteenveto-osa + erillisartikkelit)
Julkaisuluvan myöntämispäivä 15.01.2013	Kieli Englanti
Käsikirjoituksen pvm 06.11.2012	Väitöspäivä 22.03.2013
Tutkimusala Kansainvälinen projektiliiketoin	minta
${\bf Sarja} \ {\rm Aalto} \ {\rm University} \ {\rm publication} \ {\rm series} \ {\rm DOC}$	CTORAL DISSERTATIONS 30/2013
Yksikkö Tuotantotalouden laitos	
Julkaisija Perustieteiden korkeakoulu	
Päätöksentekijöiden intuition käyttö innovaat	ioprosessin alkupäässä.
Väitöskirjan nimi	
Olli Hyppänen	
Iekija	

Tiivistelmä

Innovaatioprosessin alkupää on todettu tärkeimmäksi päätöksentekopaikaksi koskien uusia innovaatioita. Tämä väitöskirja tarkastelee päätöksentekijöiden intuition käyttöä innovaatioiden alkupään päätöksenteossa. Tulokset johtajien päätöksenteon tutkimuksesta osoittavat, että päätöksentekijät käyttävät intuitiota useimmiten epävarmoissa päätöksentekotilanteissa. Tuotekehityshankkeiden alkuvaiheen päätöksentekoa pidetään yleisesti sellaisena, jossa epävarmuus on korkea. Intuition käyttöä päätöksenteossa ei kuitenkaan ole juurikaan tutkittu tuotekehityshankkeiden alkupään päätöksenteon yhteydessä. Varsinkin empiirinen tutkimus alalta on ollut vähäistä. Nykyinen tutkimus on keskittynyt lähinnä kehittämään uusia rationaaliseen päätöksentekoon liittyviä malleja vähentääkseen epävarmuudesta johtuvaa päätöksenteon vaikeutta.

Empiirinen aineisto tähän tutkimukseen kerättiin haastattelujen ja kyselytutkimuksen avulla. Aineisto kerättiin neljästä tuotekehitystä tekevästä informaatioteknologian alan yrityksestä. Aineisto koostuu 19 kokeneen päätöksentekijän haastattelusta sekä 86 kyselytutkimusvastauksesta. Aineiston analysointiin käytettiin sekä grounded theorymenetelmää että tilastollista analyysiä. Tutkimuksen pääasialliset tulokset intuition käytöstä innovaatioiden alkupään päätöksenteossa saatiin grounded theory- menetelmän avulla.

Tutkimuksen tuloksena syntyi nelitahoinen malli paljastamaan ja kuvaamaan intuition käyttöä innovaatioiden alkupään päätöksenteossa. Tutkimuksen tuloksena syntyi myös malli siitä kuinka kokeneet päätöksentekijät käyttävät intuitiota päätöksenteossa sekä millaisia intuitiiviset päätöksentekijät ovat ominaisuuksiltaan. Lisäksi vertailtiin kokeneiden ja kokemattomien päätöksentekijöiden eroa intuition käytön suhteen. Tulokset osoittavat, että molemmat uskovat intuition käytön hyödyllisyyteen ja myös käyttävät intuitiota omassa päätöksenteossaan.

Tämä tutkimus vie eteenpäin intuitiivisen päätöksenteon tutkimusta määrittelemällä intuitiivisen päätöksentekijän ominaisuuksia sekä esittelemällä uuden tavan tutkia intuition käyttöä päätöksenteossa. Tutkimus kehittää myös innovaatiojohtamisen ja tuotekehityshankkeiden tietämystä kuvaamalla kuinka intuition käyttö tulee esille innovaatioiden alkupään päätöksenteossa ja kuinka intuitiota käytetään tuotekehityshankkeiden alkupään päätöksenteossa. Tutkimuksen tuloksilla on käytännöllistä merkitystä kehitettäessä innovaatioiden alkupään päätöksentekomenetelmiä ja päätöksentekijöiden osaamista.

Avainsanat Intuitio, päätöksenteko, epävarmuus, innovaatioprosessin alkupää, innovaatio, innovaatiojohtaminen, tuotekehitys

ISBN (painettu) 978-952-	60-5029-4	ISBN (pdf) 978-98	52-60-5030-0
ISSN-L 1799-4934	ISSN (painettu	ı) 1799-4934	ISSN (pdf) 1799-4942
Julkaisupaikka Espoo	Pain	opaikka Helsinki	Vuosi 2013
Sivumäärä 204	ur	n http://urn.fi/URN	ISBN:978-952-60-5030-0

Acknowledgements

There are many individuals and parties who have influenced positively on the process that resulted to this dissertation. Unfortunately the space is limited, so here are mentioned only some of the influential persons. I hope the others would not mind too much if they are not mentioned here – my greatest thanks to everyone.

Professor Karlos Artto, my thesis supervisor, guided me into the academic world and showed the right path to follow. His support and encouragement was crucial for this project. Professor Miia Martinsuo is the person I am the most grateful for getting this dissertation done. Her valuable support and advice as an instructor made this dissertation finally possible. Without her spending countless hours on reading my drafts this journey would have never reached the end.

I am grateful for the valuable comments that the preliminary examiners professors Harri Haapasalo and Juha-Antti Lamberg gave to me during the process. Also the opponent, professor Juha Laurila, deserves great thanks for accepting the opportunity to publicly discuss with me about this dissertation.

Leif Hilden, fellow student, was the first one to wake me up that we were about to finish the studies – "hey we are going to have our last exam next week". That sparked me to finish the studies – I noticed that now when the obligatory studies are over, maybe I need to do the dissertation too.

Professor Henri Schildt, our principal lecturer in the Helsinki University of Technology ExiMa program, made the post graduate studies feel such an interesting and easy task to do. Well, it was not that easy at all – but interesting for sure.

Many thanks for checking the thesis language and making the pictures more professional goes to Mrs Carly Jaques and Mrs Laura Raesmaa. Krista Hyppänen deserves also a commendation for transcribing the – sometimes mumbling - interviews. Thanks also to those 19 interviewed persons and 86 survey respondents who made up the empirical data for this dissertation.

This dissertation received financial support from the Academy of Finland. Without it I would have never started to build my initial research thoughts to a form of a dissertation. When they, in the Academy, found the subject of this dissertation as "a good idea" to research further it encouraged me to go for it.

I'm also grateful to my previous employer Basware Oyj, especially CEO Ilkka Sihvo. He had faith in my endeavor and accepted my numerous study leaves. Also by giving me interesting and challenging tasks at work he motivated me to finish the dissertation. I owe many thanks to my Basware colleagues and team members who supported and encouraged me during this process.

Finally the greatest thanks and thoughts go to the persons who are closest to me. Valtteri, Sallamaija, Heli-Minttu and Petteri. Now I finally have more time to spend with them. With my wife Minna, we have had a wonderful time together, so many achievements (this dissertation being one of those) and experiences that I would have never dreamed of when we started our lifelong learning journey. I want to thank her for her continuous support also during this peculiar challenge – I guess new challenges will come. I sometimes have interesting ideas and dreams and that adventure continues many decades to come.

Unfortunately my parents, late Ella and Ilmari, were not able to see this work finalized. At least one thing I have learned from them and their Karelian roots – one has to strive persistently whatever obstacles are on the way, never give up - "sortumatta souta vaik' ois vastatuulta". This dissertation is dedicated to them.

N 6664866 E 375927, January 2013

Olli Hyppänen

Table of Contents

1	Int	rodu	uction	1
	1.1	Mot	tivation	1
	1.2	Res	earch questions and research scope	4
	1.3	Stru	acture of the dissertation	7
	. .			
2	Lite	erat		9
	2.1	Inti	uition in decision making	.10
	2.	1.1	Intuition definitions	.10
	2.	1.2	Intuition as a phenomenon from three different viewpoints	.12
	2.	1.3	Use of intuition in decision making	.17
	2.2	Sett	ting the scene – Innovation decision making	24
	2.	2.1	Innovation definitions	24
	2.	2.2	Research on innovation decision making	27
	2.3	The	e front end of innovation	33
	2.	3.1	Front end activities	33
	2.	3.2	Decision making at the front end	36
	2.4	Intı	uition in new product development front end	39
	2.5	Sun	nmary and conclusions of literature review	43
3	Res	sear	ch design and methods	.47
Č	3.1	Obi	ective and research questions	47
	3.2	Res	earch approach	.48
	3.3	Res	earch procedure	
	3.4	Inte	erview data collection and analysis	55
	3.	4.1	Company demographics	56
	3.	4.2	Interviewees	.58
	3.	4.3	Interview data collection	59
	3.	4.4	Interview data analysis using grounded theory approach	. 62
	3.5	Que	estionnaire data collection and analysis	65
	3.	5.1	Questionnaire data description and collection	65
	3.	5.2	Questionnaire and variables	67
	3.6	Que	estionnaire data analysis	. 69
	3.7	Vali	idity, reliability and the researcher's role	. 69
	3.	7.1	Validity and reliability	69

3.7.2	Researcher's role72
⊿ Result	5
4.1 Co	mpany descriptions
4.1.1	Company Alfa
4.1.2	Company Beta
4.1.3	Company Gamma
4.1.4	Company Delta
4.1.5	Summary: Innovative companies80
4.2 Fin innovatio	rst encounter with the data: Decision makers' experiences of on management
4.2.1	Innovation enablers82
4.2.2	Compelling forces86
4.2.3	Decision making elements89
4.2.4 decisio	Summary after the first encounter with the data: Innovation n making92
4.3 Se front end	cond encounter with the data: Facets on using intuition at the of innovation
4.3.1	Symptoms of the use of intuition94
4.3.2	Preceding enablers98
4.3.3	Simultaneous enablers 100
4.3.4	Safeguards104
4.3.5 facets o	Summary after the second encounter with the data: Four of intuition frame105
4.4 Th makers' u	ird encounter with the data: Alternative approaches to decision use of intuition
4.4.1	Drifter110
4.4.2	Thinker112
4.4.3	Negotiator114
4.4.4	Tester115
4.4.5	Discoverer 117
4.4.6	Believer118
4.4.7	Seer
4.4.8 makers	Summary after the third encounter with the data: Decision a approaches in using intuition
4.5 Su	rvey questionnaire results 123
4.5.1	Variable structure 124
4.5.2	Descriptive statistics and correlations127
4.5.3	Results127
4.6 Sy	nthesis 129
5 Discus	ssion
5.1 Re	psponses to research questions
0	

5.2 The	oretical contribution147
5.2.1	Contribution to intuitive decision making research147
5.2.2	Contribution to intuition research instruments149
5.2.3	Contribution to innovation management research150
5.2.4	Contribution to decision making and sense-making research152
5.3 Pra	ctical implications153
5.4 Eva	luation of the study156
5.5 Ide	as for further research161
6 Refere	nces164
7 Appen	lices
7 Appen 7.1 Apj	lices
7 Appen 7.1 Apj 7.1.1	lices
7 Appen 7.1 Apj 7.1.1 7.1.2	lices
7 Appen 7.1 Apj 7.1.1 7.1.2 7.2 Apj experience	lices
7 Append 7.1 Apj 7.1.1 7.1.2 7.2 Apj experience 7.3 Apj encounter	dices

List of Figures

Figure 2.1: Intuition definition (adapted from Kahneman (2003b)) 12	
Figure 2.2: Information gathering and processing styles (based on Scott &	
Bruce, 1995)	
Figure 3.1: Research procedure. Interview data	
Figure 3.2: Research procedure. Survey data	
Figure 4.1: Network of codes for decision making	
Figure 4.2: Illustrative figure of the use of intuition - "Four facets of	
intuition frame"107	
Figure 4.3 Illustration of the role descriptions and their connections (based	
on commonalities)121	
Figure 4.4: Scree plot of cleaned factor model126	
Figure 4.5: Synthesis and relations of intuition frame and three perspectives	
to innovation management131	
Figure 5.1: The appearance of intuition (in Italic) in innovation	
management perspectives150	
Figure 5.2: The ways to foster use of intuition156	
Figure 7.1: Co-citation network of "decision making AND innovation	
management"	
Figure 7.2: Keyword relatedness in search pattern "decision making AND	
innovation management"	
Figure 7.3: Co-citation network of "strategic decision making AND	
innovation management"	
Figure 7.4: Co-citation network of "decision making AND intuition" 185	

List of Tables

Table 2.1: Experience based intuition definitions
Table 2.2: Affective/sensory based intuition definitions15
Table 2.3: Holistic intuition definitions16
Table 2.4: Use of intuition in decision making
Table 2.5: Selected references of innovation management research (NPD) in
environmental, individual and organizational categories
Table 2.6: Front end process models
Table 2.7: Decision making models used in NPD front end to handle
uncertainty
Table 2.8: Recent intuition studies in NPD (NBD, TBS) context
Table 2.9: Summary of present research in connection with the research
questions
Table 3.1: Company interview demographics 57
Table 3.2: Demographics of interviewed persons
Table 3.3: Demographics of the control group (inexperienced decision
makers)
Table 3.4: Demographics of the whole survey group
Table 4.1: Most appeared codes per interviewed person (CxPy) 109
Table 4.2: Role descriptions matrix123
Table 4.3: Final factor model used in the survey data analysis125
Table 4.4: REI variable structure126
Table 4.5: Descriptive statistics
Table 4.6: Results, Chi-Square (x²) and t-test 128
Table 4.7: Role descriptions matrix129

1 Introduction

"We know more than we can tell."

Michael Polanyi

1.1 Motivation

Innovation matters, not only at the level of individual enterprises but also at the level of nations and their economic growth. Business Week regularly lists the top innovative firms in the world. It found that between 1995 and 2005 the median profit margin of the 25 most innovative firms increased almost ten times more than other companies in the S&P Global Index (BusinessWeek, 2009). In Finland the national innovation strategy was published in 2008 (Aho et al., 2008). The strategy resulted in the Innovation Policy Report by The Council of State to the Parliament which was given to the Parliament in 2008. The report states that an innovation system in the Finnish economy is important for surviving in the global competition.

The ability to innovate is a key ingredient in the survival and prosperity of companies, especially in companies connected to product development. Without innovations companies are unable to launch new offerings to the market, develop new business models or define totally new businesses and are therefore not able to grow their business (Dougherty & Hardy, 1996; Porter, 1990). It is evident that the management of innovations is important, however research has shown that the majority of innovation processes or product development projects fail (Cooper, 1999; Liberatone & Stylianou, 1995). New product launches have only a 60% success rate (Hultink et al., 2000). This success rate has not changed at all in over 50 years (Booz et al., 1968). The reasons for failure are numerous: Mistakes in

innovation process execution, failures in idea screening and selection, timing in the launching the products, delays in development and failures in decision making. The difficulty in selecting the best ideas to be further developed is one of the issues that can be solved in innovation management. The selection is about evaluating the available options and then making decisions.

The importance of selection is highest in the front end phase of innovation (Calantone et al., 1999; Cooper & Kleinschmidt, 1987; Stevens & Burley, 2003; Verganti, 1997). Innovation front end is also the focus of this study. In this study, the front end refers to those tasks and activities which are accomplished before the formal development project is launched (Reinertsen, 1985, 1999). Decisions about whether or not to invest in a new idea for further development are made at the front end. The decision making situation at the front end is challenging in many ways because the results of the development can be seen only when the product is ready. At the decision making point there is uncertainty about market, competition, technology etc. (Kim & Wilemon, 2002b; Nobelius & Trygg, 2002; Zhang & Doll, 2001).

Decision making at the front end has been regarded as one of the key factors in successful innovation management (Calantone et al., 1999; Cooper & Kleinschmidt, 1995; Khurana & Rosenthal, 1998; Tzokas et al., 2004). However, the essence of decision making at the front end of innovation management still required more clarification: How are decisions made or do they just happen, is there a thorough rational analysis behind every important decision or are the decision makers just using their gut feeling? There is limited research available about decision making in innovation management, especially concerning the early phases (front end) of the innovation process where uncertainty and ambiguity are highest (Brentani & Reid, 2012; Kim & Wilemon, 2002b; Zhang & Doll, 2001). In addition innovation front end decision making has mainly been studied from a rational analysis and decision making perspective (Calantone et al., 1999; Hart et al., 2003).

Intuition has recently received considerable research attention in strategic management (Dane & Pratt, 2007; Khatri & Ng, 2000), project management (Dayan & Di Benedetto, 2011; Leybourne & Sadler-Smith, 2006), hu-

man resource management (Hodgkinson & Sadler-Smith, 2003a) and in manager decision making in general (Kahneman, 2010; Matzler et al., 2007; Sinclair et al., 2009). The definition of intuition has varied substantially among studies. I have based my definition of intuition on recent studies (see e.g. Dane & Pratt, 2007; Hodgkinson et al., 2008; Sinclair & Ashkanasy, 2005; Sinclair et al., 2009). Thus, intuition in this study is defined as:

"Intuition is an unconscious process involving holistic associations that are produced rapidly, which result in affectively charged judgments".

The literature suggests that many managers use intuition in turbulent environments where the decisions must still be made (see e.g. Burke & Miller, 1999; Sadler-Smith & Shefy, 2004; Shapiro & Spence, 1997; Simon, 1987). Research of the innovation front end has concentrated on describing decision makers' work based on a formal rational analysis in the formal screening and gate decisions (Cooper, 2008; Hart et al., 2003). However, there are also studies which stress that formal rational analysis is not the only component of innovation decision making (Sinclair et al., 2009; Tzokas et al., 2004). One of the components used in decision making is the use of intuition.

This study brings new empirical evidence to the innovation decision making discussion, from a non-rational (use of intuition) perspective, concentrating on the early phases (front end) of innovation in New Product Development (NPD). The division between rational and non-rational decision making is not clear cut. In order to clarify the terminology used in this study, intuition is considered to be non-rational. Traditional normative decision making methods are considered rational. It is acknowledged that the use of intuition in decision making can also be a rational act by the decision maker.

Even though there are some studies that report that intuitive judgment is used in a NPD context (Dayan & Di Benedetto, 2011; Hart et al., 2003; Stevens & Burley, 2003; Tzokas et al., 2004; Yahaya & Abu-Bakar, 2007), the results are varied. This is due to the variety of definitions of intuition and of intuitive decision making and because of a lack in available research instruments needed to study the phenomenon (Sinclair et al., 2009). Introduction

The justifiable motivation for this dissertation is as discussed in the previous paragraphs. The empirical evidence about the decision makers' use of intuition at the innovation front end is currently a clear research gap. This study also adds to the previous research on intuition in decision making and contributes to the research instrument discussion on intuition in decision making (see e.g. Sinclair et al., 2009). The approach taken by experienced decision makers when they use intuition as examined in this study, supplements the role descriptions in innovation management and decision making literature.

This study brings new evidence of the use of intuition in the innovation front end and help organizations further understand and develop their innovation management practices. The use of intuition at the front end of innovation is revealed by symptoms, enablers and safeguards, which either make the use of intuition visible in decision making or ease the use of intuition in the decision process. The results of this study complement descriptions of how decision makers use intuition and raises intuitive decision making to a noteworthy position alongside prevailing rational decision making models. Decision makers use the following approaches when using intuition: drifter, thinker, negotiator, tester, discoverer, believer and seer. The results of this study also help in clarifying the intuition concept and its significance in decision making in general.

1.2 Research questions and research scope

The purpose of this research was to explore the front end of innovation, to study how innovation projects are decided upon, who makes the decisions and how those people make decisions. The aim of this research was to gain a deeper understanding of decision makers' use of intuition and to offer new theoretical end empirical insights in the use of intuition in decision making at the front end of innovation. This is achieved by studying the experiences of innovation decision makers and by describing innovation management, decision making in the innovation front end and the decision making styles of the persons acting at the front end of innovation. This is important because by understanding how decision making at the innovation front end works and how the actors – decision makers – operate, helps in further developing innovation management practices.

This research was conducted using Grounded Theory (GT) method. In the context of the grounded theory method the above mentioned broad description of the purpose and aim enabled the first setting of the research in the field. The research method stipulates that the researcher should not conduct the research with pre-specified research questions in mind, because of possible biases of the researcher himself or biases arising from prior literature (Glaser & Strauss, 1967). The more specific content of the research and the research questions arose during the data collection phase as the interviewed persons described what they experienced in the innovation projects. The aim of grounded theory research is to develop an understanding that is grounded in the reality of the persons involved with the studied phenomenon.

The first batch of empirical data of this research consists of 19 interviews with experienced decision makers from 4 ICT companies. The second batch of empirical data consists of the results from 86 questionnaires from innovation decision makers (from one of the previously mentioned companies). During the first interview and the data collection phase the interviewees spontaneously described their experiences and activities on the innovation front end. They spoke about what made innovations possible in the first place, what was exciting, what had forced them to move forward in innovation projects and how the ideas and innovation projects were decided upon. The decisions were about which ideas and "seeds" to follow through with i.e. very early front end phase decisions.

The results of the initial data collection and data analysis phase (first encounter with the data) resulted in list of categories and related properties relevant in the development of the grounded theory framework for innovation front end decision making. Decision making emerged as one of the core categories in the first data analysis phase. At the beginning of the second encounter with the data, the decision making was the main focus of the data analysis. Soon the existence of non-rational elements in decision making emerged as a core category. This category was named as intuition. This prompted the first research question:

1. How does intuition reveal itself in innovation front end decision making?

The data collection and analysis continued simultaneously and resulted in new categories and properties. The properties emerging from the use of intuition in decision making became apparent when the full interview data was examined. This sparked interest in reanalyzing the research data based on the viewpoint of the interviewed decision makers. A second research question then surfaced:

2. What approaches do decision makers have when using intuition at the innovation front end?

As a result of the third encounter with the data the approaches that decision makers have when using intuition and role descriptions were resulted. Knowing that the interviewed persons were all experienced decision makers and innovators, the third research question explored the differences between experienced and inexperienced decision makers that had surfaced.

3. How do experienced decision makers differ from inexperienced decision makers in their use of intuition?

The research approach concerning the first two research questions is qualitative. The third question is studied using the survey questionnaire and quantitative analysis. The answers to the research questions are investigated based both on the previous literature and on the empirical findings of this study.

This study is limited to ICT companies and method of managing innovation front end. The ICT industry has generally been regarded as innovative and suitable for purposeful sampling (see e.g. Christensen & Raynor, 2003; Tidd & Bessant, 2009) and thus it is a suitable target when studying innovation management. ICT companies were also selected as a research target because of the author's interest in that industry.

The unit of analysis in this dissertation is the use of intuition in product development project front end decision making. Emphasis is on front end decision making and individual decision makers. The use of intuition is approached from the individual decision maker's perspective. A single project and individual level approach enables rich data and thus more truthful results, since the interview data described front end activities in detailed and focused way (Charmaz, 2006).

The empirical data is limited to Finnish and American owned ICT companies that conduct product development activities mainly in Finland. The decision making context studied in this dissertation is strategic decisions, not normal operative decisions, since the decisions at the innovation front end are more strategic. Decisions were whether an idea was worth forwarding into the development funnel or not. The interview data consists of a sample from interviews with experienced decision makers. The main argument in this study is based on the data from experienced decision makers' description of the innovation front end. Survey questionnaire data is used to compare experienced decision makers use of intuition in decision making with the process used by inexperienced decision makers.

1.3 Structure of the dissertation

Since this dissertation follows the grounded theory method, where the aim is to develop new grounded perspectives on the studied phenomenon, the focus of the dissertation is on the theory development based on empirical data. This is different than an approach which is based on extensive literature review followed by theory and hypothesis development and then verified by empirical data analysis. In order to make the reader familiar with the existing literature, a thorough literature review on the topic of intuition is presented.

Chapter 1 contains the introduction to this study with the motivation and presentation of the research problem and scope.

Chapter 2 describes the theoretical background in the intuition, innovation and new product development and decision making research fields, discusses the main definitions and theoretical concepts concerning innovation, decision making and the use of intuition and presents relevant literature and previous research. The current research gap is also brought visible. Chapter 3 describes the research design, the methods used and the data that was used in the study. The validity and reliability of the research methods and the data is also discussed as well as the effect of researchers' role in the results of the study.

Chapter 4 presents the empirical findings of the study by first explaining the interview data and analysis and then the survey questionnaire data and analysis.

Chapter 5 discusses the contribution of the study by answering the presented research questions in detail and presents an evaluation of the study. Practical implications of the results of this study and ideas for further research are also discussed.

The final chapters of the thesis are the references and appendices. The appendices are: A: Bibliometric analysis, B: Survey questionnaire used to collect data on experienced and inexperienced decision makers' use of intuition, C: Coding network used in the first encounter of the data, and D: Cross tabulation of the interviewed persons and the codes.

2 Literature review

"Everything is vague to a degree you do not realize till you have tried to make it precise"

Bertrand Russell

The purpose of the literature review is to map the scene with regard to previous research in intuition, innovation, NPD front end, decision making and the use of intuition in decision making and to address the research questions. Innovation and NPD literature complement each other in this literature review and provide broader overview of the existing research. It is to be noted that innovation process front end and NPD process front end are very much alike, especially when we look at the decision making at the front end. That's why this literature review uses literature from both sources. The literature review starts with a thorough analysis of intuition as a phenomenon including the terminology and a review of the use of intuition in decision making, especially in new product development. The review continues with the description of innovation and innovation front end literature. Those sections concentrate on presenting innovation and innovation front end literature in the light of decision making and how the use of intuition is presented in existing literature. Finally the summary presents highlights of the literature review and addresses the research questions, also explaining the current research gap.

A bibliometric analysis was conducted to support the literature review and to verify that relevant theory discussions has been taken into account (see Appendix A). The bibliometric analysis showed that the discussion of innovation management, decision making and intuition is related to a firm's competitive strategy and competitive advantages (Barney, 1991; Porter, 1990), firm's and person's competencies (Cohen & Levinthal, 1990; Eisenhardt & Martin, 2000; Grant, 1996; Prahalad & Hamel, 1990; Teece et al., 1997) and judgment under uncertainty (Benner & Tanner, 1987; Kahneman, 2003b; Kahneman et al., 1982). This research concentrates more on the person's competencies and judgment under uncertainty than on strategy and competitive advantages research streams. Literature on intuition has also strong connections towards psychology research fields. This research was purposely limited to describe intuition and use of intuition more on innovation management perspective and not go into human psychology research.

2.1 Intuition in decision making

2.1.1 Intuition definitions

Intuition has recently received considerable research attention in strategic management (Dane & Pratt, 2007; Khatri & Ng, 2000), project management (Dayan & Di Benedetto, 2011; Leybourne & Sadler-Smith, 2006), human resource management (Hodgkinson & Sadler-Smith, 2003a) and in manager decision making in general (Kahneman, 2010; Matzler et al., 2007; Sinclair et al., 2009). The literature suggests that many managers use intuition as an approach in turbulent environments where the decisions still have to be made (Burke & Miller, 1999; Sadler-Smith & Shefy, 2004; Shapiro & Spence, 1997; Simon, 1987). The research has reported the various benefits of using intuition in decision making, for example: accelerated decision making, a higher quality of decision making, and solving creative and unstructured problems (Glaser, 1995; Miller & Ireland, 2005; Sadler-Smith & Shefy, 2004).

Intuition as a concept can be viewed as one kind of information processing which differs (however is not the opposite) from analytical and rational processes. Carl Jung was one of the first scientists to theorize about intuition. He characterized intuition as a primary mode of perception which operates subconsciously (Jung, 1926,1989). Chester Barnard made a distinction between logical and non-logical processes (Barnard, 1938). By logical he defines conscious thinking as thinking that can be expressed with words or other symbols, that is, reasoning. By non-logical he meant thoughts that cannot be expressed using words or reason, which consist only of judgments, decisions or actions (based on Barnard, 1938). In recent studies the human information processing system has been categorized as part of two different cognitive systems. This is also referred as a dual processing system (Sloman, 1996; Sloman, 2006). One of the systems, which is believed to be older from an evolutionary perspective, involves the automatic and effortless learning and processing of information. This system allows people learn from the experience and is thus also referred to as experiential, automatic, tacit, natural, associative or system 1 (Gilovich et al., 2006; Kahneman et al., 1982; Sloman, 2006; Tversky & Kahneman, 1974). Intuition is often referred to in this system. The second system enables people to deliberately learn and develop ideas. It has been referred to as a rational, intentional, deliberate, extensional, rule-based and system 2 (Gilovich et al., 2006; Kahneman et al., 1982; Sloman, 2006; Tversky & Kahneman, 1974). Traditional rational decision making models use this system.

In recent research the notion of split-brain or highly separated information processing systems is rated as an over simplified model of presenting different information processing modes (Sinclair et al., 2009). Instead a model were two information processing systems operate in parallel and interact with each other is considered to have more support both in theoretical and empirical research (Burke & Miller, 1999; Epstein, 1994; Epstein et al., 1996; Gilovich et al., 2006; Hodgkinson et al., 2008; Sinclair et al., 2009).

Daniel Kahneman presented in his Nobel lecture (Kahneman, 2003a) a figure of a three cognitive system theory where perception played a role in both the intuitive and rational systems (Figure 2.1). System 1 processes function in the same way as perception. They are fast, automatic, effortless, associative and emotionally charged. They are also difficult to control and modify. System 2 processes are slower, serial, and effortful and can be de-liberately controlled. They are also flexible and rule-governed. The difference in effort provides usually an indication of which system is used. Effortful System 2 processes tend to disrupt each other whereas System 1 processes do not suffer when done in parallel with other activities.

All the process characteristics attributed to System 1 are also attributed to perceptions. The difference between perception and System 1 is that System 1 is not restricted to processing using only current stimulations as percep-

tion is. However like System 2, System 1 deals with stored concepts and can be evoked with language. System 2 monitors System 1 by using reasoning to verify that our intuitions are correct (Kahneman, 2003b). This is a personally dependent function – which may be stronger in some and weaker in others. My argument is that since individual personalities differ there are many different approaches when it comes to the use of intuition in decision making. How experience – learning from the past – affects to the development of intuition and the use of intuition can also vary among individuals. This is clarified in this study by presenting different descriptions of intuitive decision makers along with the approaches they have taken when using intuition at the front end of innovation.



Figure 2.1: Intuition definition (adapted from Kahneman (2003b)).

2.1.2 Intuition as a phenomenon from three different viewpoints

In order to understand the differences between the definitions of intuition and to build up the definition of intuition which is used in this study it is necessary to look at the available intuition definitions from the previous literature in more detailed level. When looking at the intuition phenomenon in more detail, the broad categorization of definitions based on Sinclair et al. (Sinclair et al., 2009) put intuition definitions into two mutually reinforcing categories (which then makes up the third category):

- Intuition as an experience-based phenomenon which relies on tacit knowledge accumulated through experience and quick retrieval of that knowledge using pattern recognition (e.g. Banning, 2008; Frantz, 2003; Simon, 1987).
- Intuition as a phenomenon incorporating sensory and affective elements (e.g. Agor, 1985, 1989b; Epstein et al., 1996; Parikh, 1994; Vaughan, 1979).

In summary, intuition definitions can be categorized into three groups: the first consists of definitions based on *experience*, the second is based on *affective and sensory* elements and the third category includes more *holistic* definitions of intuition.

The experience based category relies on knowledge, explicit and tacit, which has been acquired through years of practice. Mental shortcuts, pattern recognition and heuristics are associated with this category. Intuition is understood to be based on a deep understanding of the situation, but the decisions happen quickly. In this approach, there is a danger of irrational biases. A summary of experience based intuition definitions are presented in Table 2.1.

Table 2.1: Experience based intuition definitions

Source	Experience based definition of intuition /	
	Intuitive decision making	
(Tversky & Kahneman, 1974,	Mental shortcuts – heuristics – that are highly susceptible	
p. 1124)	to irrational biases.	
(Simon, 1987, p. 60), (Frantz, 2003, p. 267)	Non-rational experience based activity. Pattern recognition.	
(Burke & Miller, 1999, p. 92)	A cognitive conclusion based on the decision makers'	
	previous experiences and emotional inputs.	
(Khatri & Ng, 2000, p. 62)	Intuition is based on a deep understanding of a situation. It	
	is a complex phenomenon that draws from a store of	
	knowledge in our subconscious and is rooted in past expe-	
	rience. It is quick, but not necessarily biased as is pre-	
	sumed in previous research on rational decision making.	
(Gilovich et al., 2006, pp. 16-	Intuitive judgment is not simpler but different than rational	
17)	judgment and is a result of the use of a limited number of heuristics.	
(Miller & Ireland, 2005, p. 21)	Intuition can be conceptualized to 1. holistic hunch 2. au-	
	tomated expertise. Holistic hunch refers to a subconscious	
	gut feeling. Automated expertise refers to a recognition of	
	familiar situations and partially subconscious action.	
(Matzler et al., 2007, p. 13)	Intuition is a highly complex and highly developed form of	
	reasoning that is based on years of experience and learn-	
	ing, on facts, patterns, concepts, procedures and abstrac-	
	tions stored in one's mind.	
(Banning, 2008, p. 190)	Intuitive-Humanist model is based on experience and pat-	
	tern recognition.	

The affective category links intuition to emotions and feelings. Intuition is characterized as a judgment achieved with a strong feeling of certainty in its truth. Intuition relies on emotions and affect. Intuitive judgments may also develop slowly over time, so feelings and judgments do not have to be immediate. Intuition is based more on sensory elements than on experience. Psychological functions are highlighted in affective descriptions of intuition. The summary of affective/sensory based definition of intuition is presented in the following Table 2.2.

Source	Affective/sensory based definition of intui-
	tion / intuitive decision making
(Vaughan, 1979, p. 46)	Knowing without being able to explain how we know.
(Benner & Tanner, 1987, p. 23)	Intuition is understanding without a rationale.
(Agor, 1989a, p. 15)	Rational and logical brain skill that can be used to help
	guide decision making. Affectively charged.
(Jung, 1926,1989, pp. 567-	A psychological function that unconsciously transmits per-
568)	ceptions.
(Scott & Bruce, 1995, p. 820)	Intuitive decision making style is characterized by a reli-
	ance on hunches and feelings.
(Epstein et al., 1996, p. 401)	Two information processing modes which operate inde-
	pendently but contribute jointly to behavior. Intuition relates
	to an intuitive-experiential thinking style.
(Allinson & Hayes, 1996, p.	Immediate judgment based on feeling and the adoption of
122)	a global perspective.
(Ferguson, 1999, p. 12)	Intuition is a sensory process in human beings triggered by
	particular kinds of interactions both inside and outside our
	bodies that influences our stability and optimal perfor-
	mance.
(Rew, 2000, p. 95)	The deliberate application of knowledge or understanding
	that is gained immediately as a whole and that is inde-
	pendently distinct from the usual, linear and analytic rea-
	soning process.
(Kahneman, 2003b, p. 697)	Thoughts and preferences that come to mind quickly and
	without much reflection.
(Myers, 2004, p. 1)	Intuition is our capacity for direct knowledge, for immediate
	insight without observation or reason.

Table 2.2: Affective/sensory based intuition definitions

A holistic approach combines cognitive – experience based and affective elements of intuition. Judgments are made without direct knowledge but they require holistic associations from diverse sources of information. Intuition differs from analysis in that it is not transparent and cannot be justified by articulating the logical steps behind the judgment process. The holistic approach acknowledges both sides of intuition – experience based and affective elements – and presents a more complex set of inter-related processes also called dual-process theory (see e.g. Hodgkinson et al., 2008; Kahneman, 2003b). A summary of holistic definitions of intuition is presented in Table 2.3.

Table 2.3: Holistic intuition definitions

Source	Holistic definition of intuition /
	Intuitive decision making
(Barnard, 1938)	Non-logical processeswhich are only made known by a
	judgment, decision or action.
(Shapiro & Spence, 1997, p.	Intuition is an unconscious, holistic processing mode in
64)	which judgments are made with no awareness of the rules
	or knowledge used for the inference and can feel right
	despite one's inability to articulate the reason.
(Sadler-Smith & Shefy, 2004,	Intuition is the capacity for attaining direct knowledge or
p. 77)	understanding without the apparent intrusion of rational
	thought or logical inference.
(Sinclair & Ashkanasy, 2005,	Intuition is a non-sequential information processing mode,
p. 355-356)	which comprises both cognitive and affective elements and
	results in direct knowledge without the use of conscious
	reasoning.
(Pretz & Totz, 2007, p. 1248)	Intuition is the product of a tacit – experiential – system:
	affective, heuristic and holistic aspects.
(Gigerenzer, 2007, p. 16)	Intuition is a judgment that appears quickly, we are not
	fully aware of the underlying reasons but the feelings are
	strong enough to act upon.
(Dane & Pratt, 2007, p. 40)	Intuitions are affectively charged judgments that arise
	through rapid, unconscious and holistic associations.
(Hodgkinson et al., 2008, p.	Intuition is a complex set of inter-related cognitive, affec-
4,19)	tive and somatic processes for which there is no apparent
	intrusion of deliberate, rational thought. Intuition is related
	to the dual-process theory of cognition.
(Sinclair et al., 2009, p. 393)	Intuition is a process leading to a recognition or judgment
	that is arrived rapidly at, without deliberate or rational
	thought, and difficult to articulate verbally.

The definition of intuition has varied substantially across studies. One reason is that the term intuition itself has been treated as a process, output of the process of predisposition to use intuition. Many studies, especially those from the 1980s and 90s, did not define the term intuition clearly, which makes it difficult to compare the findings of the research (see e.g. Agor, 1985, 1986; Hayashi, 2001).

Some intuitive definitions relate to spiritual, paranormal and supernatural definitions (Leners, 1992). I have not included that path of research in this study since it is an unstructured arena of science and it conflicts with my ontological standpoint since I am trying to understand and interpret what is happening in innovation front end decision making.

I do not see intuition as a mystic, unexplained phenomenon. In this study, intuition and the use of intuition is regarded as a single component of decision making, a component which has not been clearly understood and accepted as a serious contender in the innovation front end field. My definition of intuition follows closely with the holistic interpretation of intuition as described by Dane and Pratt (2007), Hodgkinson et al.(2008), Kahneman (2003), Sinclair and Ashkanasy (2005) and Sinclair et al. (2009). The definition of intuition used in this study is the following:

> Intuition is an unconscious process involving holistic associations that are produced rapidly, which result in affectively charged judgments.

Intuition is sometimes confused with terms insight or instinct. Instincts are more like biological hard wired reflexes (Epstein, 1990; Hogarth, 2001). Insight is similar to intuition but it is often a lengthy process that begins with deliberate analytical thinking even though the end result may appear to be intuition (Koriat, 1993; Lieberman, 2000; Shirley & Langan-Fox, 1996). With insight the logics and the path to the end result is known at the end of the process, which is not the case in intuition.

Heuristics is often compared to intuition, however they are not the same thing. The use of heuristics is related to low-effort rational strategies where decision makers rely on the presented data to make a conscious guess (which may result in a biased estimate) (Sinclair & Ashkanasy, 2005; Tversky & Kahneman, 1974). Intuition assumes the absence of any awareness of the process used to reach the decision (Epstein et al., 1996; Shapiro & Spence, 1997).

2.1.3 Use of intuition in decision making

In order to study and understand the use of intuition in decision making, intuition and the intuitive process have to be apparent to the researcher. As discussed in previous chapters, the use of intuition is not necessarily a deliberate process. Therefore direct observation cannot be used as a method of observation. Previous research has suggested that intuition is difficult to describe but easy to recognize (Sadler-Smith & Shefy, 2004). This is in part true, but since definitions of intuition are so varied it is also difficult to state are we seeing "similar use(s) of intuition" when comparing previous reLiterature review

search. Previous studies have also been inclined to provide more indirect evidence for the use of intuition (Shapiro & Spence, 1997).

Individuals who rely on an intuitive information processing style tend to rely on information filtering based on perception in the information gathering phase (see Figure 2.2). The use of intuition in the information gathering and processing phases has been characterized by Scott and Bruce 1995, (based on Keen, 1973). Those who use the analytical and rational approach, focus more on details in the information gathering phase. This is consistent with long standing managerial tradition in using a Myers-Briggs Type Indicator (MBTI), which has concluded that intuitive decision makers are more likely than others to favor abstract information and perceptual processes and are more inclined to use unconventional and creative behaviors (Gardner & Martinko, 1996).



Figure 2.2: Information gathering and processing styles (based on Scott & Bruce, 1995).

The use of intuition has also been described as case dependent. A person who is intuitive by nature does not use intuition in every situation. (Kahneman, 2003b). A judgment or choice is made by:

- 1. An intuitive judgment or choice is initiated and
 - a. endorsed by System 2
 - b. adjusts for other features that are recognized as relevant
 - c. corrects (sometimes overcorrects) for an explicitly recognized bias
 - d. is identified as violating a subjectively valid rule and blocked from of overt expression

2. No intuitive response comes to mind and the judgment is computed by System 2.

The use of intuition has been found to play a major role in effective decision making in organizations (Burke & Miller, 1999; Gigerenzer, 2007; Hodgkinson et al., 2008; Kahneman, 2003b; Kahneman & Klein, 2009; Sinclair & Ashkanasy, 2005) and in strategic decision making (Eisenhardt, 1989c; Khatri & Ng, 2000; Miller & Ireland, 2005; Sinclair et al., 2009). These results from these studies do not universally support the use of intuition. Miller and Ireland (2005) state that intuition is a troublesome decision making tool whereas Khatri and Ng (2000) found that the use of intuition had a positive influence on organization performance in an unstable environment but is negative in a stable environment. Others have just noted the existence of intuition or presented the available tools required to analyze the use of intuition.

The use of intuition among senior managers has also been addressed in numerous studies (Agor, 1986; Dane & Pratt, 2007; Isenberg, 1984; Sadler-Smith & Shefy, 2004; Simon, 1987). Typically the use of intuition is in those studies related to the experience of senior managers or executives. Junior managers' use of intuition has not received the same amount of attention in these studies even though they are commonly mentioned in the results.

The studies by Weston Agor (Agor, 1985, 1986) state that top managers in all organizations differ significantly from lower level managers in their use of intuition in decision making (top managers use more), his empirical evidence is mainly based on studies of the experienced executives. Agor points out numerous techniques to activate the use of intuition and to develop managers' intuitive skills. The results of his studies are so complicated, that the main arguments remain vague and difficult to compare with other studies.

After studying dozens of senior managers, Isenberg found it hard to determine when senior managers actually made a decision and what were the rational methods employed in making those decisions (Isenberg, 1984). The study does not take into account junior managers, but he concludes that "the higher you go in a company, the more important it is that you combine intuition and rationality, act while thinking, and see problems as interrelated" (ibid, p.81). According to Isenberg, there are five different ways in how senior managers use intuition (sensing the problems existence, rapid reactions, building integrated pictures, check for rational analysis, by-pass analysis). These can be used as a structure for studying intuition and its use among senior managers.

Dane and Pratt (2007) have found that the use of intuition in decision making is effective in complex domain-relevant situations. Their study also proposes that experts, who have years of experience (implicit and explicit learning), are superior in their use of intuition when compared to less experienced decision makers. Their study still lacks empirical data and evaluation, and is only based on the analysis of previous research. They also propose that intuition "use factors" that can be uncovered by observing, could be found out by studying the role of body factors – how people react when they use intuition.

Sadler-Smith and Shefy (2004) present a set of guidelines for developing intuitive awareness (open up the closet, don't mix your I's, elicit good feedback, get a feel of your batting average, use imagery, play devil's advocate, capture and validate your intuitions). Their study is limited to executives and experienced decision makers and did not present the ways in which the use of intuition can be made more apparent. Instead it is stated that intuition is self-evident, without empirical sound evidence.

Entrepreneurs have also been found to make more use of intuition in their decision making than non-entrepreneurs (Armstrong & Hird, 2009). Entrepreneurs who are involved in the early stages of venture creation were also found to possess more entrepreneurial drive. Armstrong and Hird used a Cognitive Style Index (CSI) to study intuitiveness. The instrument itself (CSI) has been questioned since the results based on its use are varied (Hodgkinson & Sadler-Smith, 2003a, 2003b). How decision makers use intuition is also not answered by their results. They raise this issue and indicate that it is an opening for the future research.

Previous research presents different tools in developing and evaluating the use of intuition, and therefore recommendations often differ significantly from study to study. As a result, there are no clear and explicit means of showing how the intuition is used in decision making.
In addition to contemporary management research in industrial economics and management arena intuition has received major research focus in medical schools with regard to clinical decision making (Benner & Tanner, 1987; King & Appleton, 1997; Rew, 2000). When in contemporary management research analytical decision making approaches are main stream, then the use of intuition in decision making is considered traditional in clinical decision making research. Intuition has thus been found to be one of the main components in decision making among medical doctors and nurses.

In most of the clinical decision making studies the use of intuition have been found to have a positive effect on medical decision making. However, criticism towards the use of intuition has been raised by some clinical researchers (Lamond & Thompson, 2000). They argue that the patients and other health care professionals need to know the basis of decisions about treatment and that this is often not possible when using intuition. They also point out that if the decisions are made based on intuition, then they lose the opportunity to learn and share any findings based on those decisions. So, instead of only using intuition in decision making, they encourage the use of analysis in clinical decision making (this is quite the opposite as in traditional management research) (Lamond & Thompson, 2000; Shamian, 1991).

Since intuition is often linked with creativity, the use of intuition has also been researched in connection with the creative design processes (Polanyi, 1999; Raami et al., 2010). The use of intuition has been noted to have a major impact especially on the work of experienced designers. Moreover, in systems intelligence research the use of intuition – or emotions – in decision making plays a central role (Dufva, 2008; Hämäläinen & Saarinen, 2008b; Saarinen & Hämäläinen, 2004). The proposition from a systems intelligence perspective is that decision making must be seen as a system where many components (one of those is intuition – emotions) operate and have to be managed simultaneously. The use of intuition is visible but how it is used and how decision makers use intuition is not explained.

Even though there has been a number of studies concerning the use of intuition, ability and preference, it has been hard to compare the actual re-

sults of these disperse studies. There is also limited empirically sound evidence of the use of intuition in decision making (Hodgkinson et al., 2008; Sinclair et al., 2009). A large part of the results of these studies were from a quantitative survey analysis (Agor, 1989b; Allinson & Hayes, 1996; Benner & Tanner, 1987; Epstein et al., 1996). The research context – intuition – is still an unknown area and difficult to quantify and therefore quantitative methods are not always the most suitable method for this kind of study. Recent studies of intuition have tried to define the constructs of intuition and methods for studying intuition, but they also urge a more analytical and empirical approach for studies of the components of intuition (Hodgkinson et al., 2008). Table 2.4 summarizes the selected references and the use of intuition in decision making related research and their relevance for this study.

Research area	Selected refer- ences	Contribution from the selected refer- ences	Relevance to this study
Information gather- ing and processing styles	(Gardner & Martinko, 1996; Kahneman, 2003b; Keen, 1973; Scott & Bruce, 1995)	Intuitive decision makers gather infor- mation differently than analytical ones. Use of intui- tion(system 1) can be combined with system 2 processing.	Means to differenti- ate decision makers based on infor- mation gathering. Case dependent use of intuition.
Intuitive decision making	(Burke & Miller, 1999; Gigerenzer, 2007; Hodgkinson et al., 2008; Hodgkinson & Sadler-Smith, 2003a; Kahneman, 2003a; Kahneman & Klein, 2009; Sinclair & Ashkanasy, 2005; Tversky & Kahneman, 1974)	Intuition as an im- portant component in decision making. Use of intuition and uncertainty are relat- ed.	Tools to measure intuition. Different approaches to using intuition in decision making. Decision making styles.
Intuition and strate- gic decision making	(Eisenhardt, 1989c; Eisenhardt & Zbaracki, 1992; Khatri & Ng, 2000; Miller & Ireland, 2005; Sinclair et al., 2009)	Intuition is used in strategic decision making	Intuition in key roles in strategic decision making. How intui- tion reveals itself in strategic decision making.
Clinical decision making, Arts&creativity	(Benner & Tanner, 1987; Lamond & Thompson, 2000; Polanyi, 1999; Raami et al., 2010; Rew, 2000)	Intuition as the main decision making approach. Also crit- ics towards the use of intuition on the clinical side.	Connection to disci- plines other than management sci- ences.
Managerial decision making	(Agor, 1985; Dane & Pratt, 2007; Isenberg, 1984; Sadler-Smith & Shefy, 2004; Shapiro & Spence, 1997; Simon, 1987)	Senior managers use intuition in deci- sion making.	Use of intuition is connected to the decision maker role. Differences between experienced and inexperienced deci- sion makers not studied. Empirical evidence missing.
Systems intelligence	(Dufva, 2008; Saarinen & Hämäläinen, 2004)	Intuition as one crite- rion in systemic decision making.	Intuition is recog- nized as one com- ponent in decision making, but no de- tailed explanation.

2.2 Setting the scene – Innovation decision making

2.2.1 Innovation definitions

Innovation is a broad concept that can be defined in a variety of ways. Even the definition of innovation as a process or as a result of the process is sometimes mixed. Inconsistencies in definitions of innovation, innovation process and front end process models make it difficult to compare the results of the previous research. Therefore the following chapters define the terms used in this study. The literature presented contains both innovation and new product development front end literature. This is to get a broader picture of the existing research. It is acknowledged that innovation process and NPD process are not the same – innovation process has a broader scope than NPD process. They are however much alike when looking at the decision making at the front end. To severely limit the literature only to the NPD front end would have unnecessarily narrowed the scope – NPD is an innovation activity as such.

Innovation is a commercialized invention as defined by Schumpeter (Schumpeter, 1934). That definition is the basis for many subsequent definitions. Typically discussions of innovations only concern product innovation. Innovations can also happen for example in a service, process and business context. The innovation management literature presents several ways to categorize innovations. All of them use at least two basic elements in categorization: <u>scope</u> or type of innovation and the <u>degree of novelty</u>. Categorization by scope or type (essentially change) can use the following four broad categories (Francis & Bessant, 2005):

- <u>Product innovation</u>: Changes to the things (product/services) that an organization offers.
- <u>Process innovation</u>: Changes in the ways in which products are created and delivered
- <u>Position innovation</u>: Changes to the context in which the products/services are introduced.
- <u>Paradigm innovation</u>: Changes to the underlying mental models that frame what the organization does.

All four categories can then de characterized by a degree of novelty from incremental to radical (also called discontinuous) (Christensen, 2003; Garcia & Calantone, 2002; Tidd & Bessant, 2009). In incremental innovations an already existing product is gradually developed to include new features. This can also be said for other types of innovations (service, process, etc.). It is simply doing what we already do but better. Radical or discontinuous innovation is such where a radically new product or service is commercialized which opens up a totally new business. There can also be discontinuity e.g. in product design because of a discovery of new technological possibilities which makes it possible to do the same but with new and more efficient means. Disruptive innovation is even more extreme than radical innovation. It profoundly disrupts the very basics of business and creates new business models or technologies (Christensen & Raynor, 2003; Christensen, 2003; Leifer et al., 2000; Wheelwright & Clark, 1992).

Whatever the categorization or characterization that is used, the line between the different types of innovations is often unclear because it also depends on context, environment and other variables. Something considered radical in one industry could be incremental in another. It is also hard to see the difference between paradigm innovation and radical innovation. If the innovation is at a paradigm/mental model level it is in fact radical. In practice, the important aspect is that different types of innovations require different kinds of mechanisms for the management of innovations because the complexity and uncertainty so greatly differs between different innovation cases.

When defining innovation it is important to distinguish the innovation process from innovation itself. The innovation process leads to innovation and that process needs to be managed. Much of innovation management is the management of the innovation process. Since the innovation process is understood as a process of turning ideas into reality and capturing the value of their use (Drucker, 1985; Porter, 1990; Rothwell & Gardiner, 1985; Tidd & Bessant, 2009; Trott, 2005), there are defined phases in that process. Descriptions of the innovation process vary as do definitions of innovation itself. In many cases the innovation process has four key phases: search, select, implement and capture. Search refers to how we can find the opportunities to breed further. Select is about making strategic choices about future actions. Implementation is about converting the ideas into reality and

capturing is about how to benefit from what just has been implemented (adapted from Brown & Eisenhardt, 1995; Cormican & O'Sullivan, 2004; Khurana & Rosenthal, 1998). Typically the first two phases (search and select) are called the front end of innovation process. Different studies has been done in order to evaluate and develop practices in these different activities (see e.g. Brown & Eisenhardt, 1995; Cooper, 1990). The development of the innovation process has mainly concentrated on the development of the innovation process so that it is more rational and analytical when it comes to decision making.

The best known rationality based innovation process (actually NPD process) description is the stage gate process model (Cooper, 1990). The basic idea behind the stage-gate process is related to the physical – manufacturing – process where the inputs and outputs of the machine are known and can be controlled. This is the major deficiency in this sort of process description – innovation management is not that rational and cannot be planned in a totally analytical way. This is even more the case at the innovation front end. The innovation process seldom follows a purely linear innovation process model. Instead innovation model descriptions have developed from early linear model descriptions to interactive, flexible and more complex models (Cooper, 2008; Rothwell, 1992; Trott, 2005). The flexibility of the process has been introduced by focusing on the need for communication between the parties, the importance of interfaces and the roles operating in those interfaces and by creating iterations in addition to a pure linear process model.

Based on the earlier studies the following definitions for innovation and innovation process are used in this dissertation:

- Innovation = (theoretical) conception + (technical) invention + (commercial) exploitation.
- Innovation is a result of innovation process which needs to be managed.

This definition highlights the conception of new ideas as the starting point for innovations. A new idea is not an invention nor is it an innovation, it is just a concept. The process of converting concepts into a tangible new artifact is invention. The commercial exploitation completes the process and takes it to the process of innovation. Invention occurs typically at an individual level whereas innovations require a social context. Invention is thus a cognitive process and innovation is a social process (Higgins, 1995; Schumpeter, 1939; Trott, 2005; Van de Ven, 1986).

The innovation process is complex and the management of that process involves the management of several different activities (Cormican & O'Sullivan, 2004; Tidd & Bessant, 2009; Trott, 2005; Van de Ven, 1986). Research also shows that a high proportion of new product ideas fail commercially (Cooper, 1999) i.e. product innovation fails. Liberatone and Stylianou found that less than one sixth of the ideas that enter new product development process are commercially successful (Liberatone & Stylianou, 1995). Based on the study from 1990s by Hultink et al. the new product launches have only a 60% success rate (Hultink et al., 2000). It is remarkable that the success rate has remained unchanged for more than 50 years (Booz et al., 1968; Jones, 1958).

2.2.2 Research on innovation decision making

There are numerous studies which have increased the understanding about decision making in innovation management and new product development. Decision making cannot be viewed only from the NPD process (*organizational*) side. External (*environmental*) factors as well as *individual* factors need to be considered when decision making is evaluated (Balachandra & Friar, 1997; Reid & de Brentani, 2004; Trott, 2005; Van de Ven, 1986). This categorization of innovation management research into three categories (*organizational*, *environmental*, *and individual*) is used as a method to organize the existing literature and the empirical findings and to discuss the findings of this research in relation to the previous research. The following Table 2.5 summarizes influential decision making studies in innovation management and the NPD context.

Table 2.5: Selected references of innovation management research (NPD) in environmental, individual and organizational categories.

Component	Selected references	Contribution from the reference research	Relevance to this study
ENVIRONMENTAL			
Life cycle of technol- ogy, industry, etc.	(Abernathy & Utterback, 1978; Tushman & Anderson, 1987; Utterback, 1994)	Different life cycles em- phasize different aspects of innovation	Changing decision making approaches
Degree of novelty – continuous vs. dis- continuous	(Christensen, 2003; Leifer et al., 2000)	Different management approach needed depend- ent on degree of novelty	Different decision making approaches
High velocity envi- ronments	(Brown & Eisenhardt, 1995; Eisenhardt, 1989c)	Fast decision makers use more information and that leads to superior perfor- mance.	Decision makers' use of information and decision mak- ing style.
INDIVIDUAL			
Individual decision making theory	(Dean & Sharfman, 1996; March, 1994; Nutt, 1984)	Individuals have different preferences about the decision making model that they use (rational ap- proach)	Rational decision making models.
Teams and individuals	(Allen, 1970; Day & Schoemaker, 2004; Eisenhardt & Bourgeois, 1988; Kim & Maubourgne, 1998; Park et al., 2009; Rothwell & Gardiner, 1985; Schmidt et al., 2001; Thamhain & Wilemon, 1987) (Allen, 1970; Burgelman & Sayles, 1986; Chakrabarti, 1974; Cooper & Kleinschmidt, 1987; Gemunden et al., 2007; Hippel, 2006; Kim & Wilemon, 2002a; Tushman, 1977; Tushman & Katz, 1980; Zirger & Maidique, 1990)	Innovative team composi- tion fosters innovation. Team vs., individual deci- sion making performance. Roles which facilitate inno- vation.	Team innovation as a composition of individuals. Political decision making. Individuals with different competen- cies and practices. Decision making in relation to individu- al roles. Sense-making and learning
ORGANIZATIONAL			
Size of the company	(Oakey, 1991; Rothwell, 1983, 1993)	Small firms need to devel- op more linkages	Complexity in deci- sion making
Processes	(Adams et al., 2006; Calantone et al., 1999; Cooper, 1990; Cooper & Kleinschmidt, 1995; Khurana & Rosenthal, 1998; Krishnan & Ulrich, 2001; Reid & de Brentani, 2004; Smiths, 2002; Tzokas et al., 2004)	The importance of a pro- cess approach. More empirical research is needed on processes and decision making systems.	The holistic ap- proach. The Im- portance of front end phase. Focus on decision making importance. Research gap identified in deci- sion making sys- tems.

The environmental research stream has introduced decision making as a main component of innovation management and the need for different decision making approaches as dependent on the degree of innovation novelty (Christensen, 2003; Leifer et al., 2000). Technology and industry life cycles also have an effect on how innovations are managed (Abernathy & Utterback, 1978; Tushman & Anderson, 1987; Utterback, 1994). Findings propose that different decision making approaches are needed depending on the industry or the technology maturity phase. High velocity environments and decision making in those environments in particular have been studied by Brown and Eisenhardt (Brown & Eisenhardt, 1995; Eisenhardt, 1989c). They highlight the importance of getting more data and especially accurate data from the situations at hand and having faster and more efficient decision making processes.

Previous research on the environmental side has highlighted the importance of decision making in innovation management, however it lacks an explanation of how the environmental aspects affect decision making and how those characteristics that are present in the decision making change the environmental conditions. In fast changing environments the amount of available information and the pace of decision making situations reach its peak in decision makers' "computing capability" and limits further development of decision making practices. Coping with the uncertainty inherent during the early phases of the innovation process has rarely been researched.

Individual perspective deals with the role of the individual in managing and advancing innovation within a firm. Most of the decision making activities in the early phases of innovation are handled by individuals before the organization gets involved in the process (Reid & de Brentani, 2004). When decision making is viewed only from the individual viewpoint, the two most common explanations of individual decision making are logic of consequence and logic of appropriateness (March, 1994). Logic of consequence (rational procedure) assumes that the decision maker has clear preferences, that there are choices connected to those preferences and that the outcomes or consequences of different actions are known (Dean & Sharfman, 1996; March, 1994; Nutt, 1984). The logic of appropriateness is the rule following procedure. The decision maker follows the rules (for example the innovation process description) which they see as appropriate for the decision making situation at hand (March, 1994). In innovation management, logic of consequence of the stage gate innovation management process and a process descriptions itself (rule following) are good examples of how logic of consequence and logic of appropriateness manifest in innovation decision making. The basic assumption in these models is the rationality of the decision maker.

The study of innovation champions has explained the role of champions, gatekeepers and boundary spanners (Allen, 1970; Chakrabarti, 1974; Cooper & Kleinschmidt, 1987; Kim & Wilemon, 2002a; Tushman, 1977; Tushman & Katz, 1980; Zirger & Maidique, 1990). A champion is a person who makes a contribution to an innovation by actively promoting and progressing it in the organization (Burgelman & Sayles, 1986; Chakrabarti, 1974). Gatekeepers are individuals who direct information and who decide whether or not to share information with others. Boundary spanners are the persons who operate at the boundaries of the organization and take care of the link between the organization and outside world (Tushman, 1977). In addition to these traditional roles, other roles like expert, power, process and relationship promoter roles have been identified as having an effect on innovation management success (Gemunden et al., 2007). Another aspect presented by Hippel (Hippel, 1988, 2006) takes users inside innovation and emphasizes their role in decision making in addition to companies' internal contributors.

When individuals are grouped as a team, the performance of the team is dependent on the team structure and on communication networks (Allen, 1970). Team innovation is a composition of the individuals whose qualities have an effect on the total performance of the team (Day & Schoemaker, 2004; Rothwell & Gardiner, 1985; Thamhain & Wilemon, 1987). The best composition is typically a team with diverse individuals (Park et al., 2009). The same applies in decision making performance (Schmidt et al., 2001).

Team decision making can also be analyzed from the viewpoint of inconsistent multiple actors (typically teams are somehow consistent). When the conflicts between the team members cannot be resolved, decision making becomes conflictual or political (Eisenhardt & Bourgeois, 1988). People need to co-operate voluntarily to get the best results out of group decision making (Kim & Maubourgne, 1998).

The individual side of innovation management research has clearly found that individual persons have a key role in innovation decision making. They are the decision makers who make the final decision in the process. Even though the different roles have been thoroughly described, the current research lacks in its explication of how the individual decision maker ends up with decisions and if there any differences between the different roles in their decision making approach.

The organizational perspective relates to factors coming from organization structure but also from organizational level processes. Small companies require more linkages and networks in order to successfully manage innovations (Rothwell, 1983, 1993). The complexity of decision making thus gets different view in networked and rapidly growing firms (Oakey, 1991). Decision making and interactive structures in internal networks have also been found to be one of the key factors (Birkinshaw et al., 2007) in radical innovation and more empirical research on that agenda has been suggested (Smiths, 2002).

One factor affecting innovation management and decision making is the type of innovation to be managed (incremental, radical/discontinuous) (Drejer, 2002; Garcia & Calantone, 2002; Reid & de Brentani, 2004). Incremental innovations are managed with established processes and organizations whereas radical types of innovations require different skills, management, decision making and problem solving approaches. In incremental innovations the innovation case is typically known in advance - it could for example be an incremental enhancement to a product that is already on the market. Existing innovation management processes exist to handle this kind of decision making and management issues. Problems are structured and organizational structures as well as individual competences are available. The decision making approach in those cases is mainly rational and logical.

New product development research touches all aspects of previously described categories (environmental, individual, and organizational) but it is mostly related to the organizational level processes and issues and product

development innovations (Cooper, 1999; Cooper & Kleinschmidt, 1987; Zirger & Maidique, 1990). Decision making has been raised as a key area of NPD process gate decisions (Cooper & Kleinschmidt, 1995; Hart et al., 2003; Tzokas et al., 2004). NPD studies usually touch on the decisions of later phase innovation processes when the product is already in the development funnel or where the products or services are ready to be launched to the markets (Hart et al., 2003; Krishnan & Ulrich, 2001; Ozer, 1999). In these studies the early phases of innovation process have had less attention than launch decisions. Typically the studies deal with what influences certain decisions can have in the success or failure of the product development (e.g. resource allocation, product concept, product mix, etc.), not on how the decisions were made. They usually present a framework for the decisions that have to be made during different phases of the project (concept development, supply chain design, product design and production ramp up and launch).

Previous research as presented in earlier chapters describes decision making as originating from a central place in innovation management and NPD research. The importance of decision making has been highlighted by previous innovation management research from the environmental side e.g. (Christensen & Raynor, 2003; Eisenhardt, 1989c; Leifer et al., 2000; Tushman & Anderson, 1987; Utterback, 1994), from the individual side e.g. (Chakrabarti, 1974; Gemunden et al., 2007; Rothwell & Gardiner, 1985; Schmidt et al., 2001; Thamhain & Wilemon, 1987; Tushman, 1977) and from the organizational side e.g. (Adams et al., 2006; Cooper & Kleinschmidt, 1995; Khurana & Rosenthal, 1998; Krishnan & Ulrich, 2001; Reid & de Brentani, 2004).

Innovation management is thus much about decision making in each innovation process gate or between the interfaces of different phases in the process. However the essence of the decision making at the innovation front end still seeks more clarification and more empirical research is needed (Smits, 2002). The focus in the vast majority of previous studies is only on the rational decision making models side, with some exceptions e.g. (Reid & de Brentani, 2004; Yahaya & Abu-Bakar, 2007). Previous research, especially in the NPD area, has also concentrated on presenting decision making to the later – more rational - parts of the innovation process (Bacon et al., 1994; Cooper, 1990; Cooper & Kleinschmidt, 1995; Hultink et al., 2000; Krishnan & Ulrich, 2001; Ozer, 1999; Tzokas et al., 2004) and not on the early innovation process phases (front end) where the most important decision are made (Adams et al., 2006; Calantone et al., 1999; Cooper & Kleinschmidt, 1995; Khurana & Rosenthal, 1998) and where the uncertainty is at its highest (Brentani & Reid, 2012; Kim & Wilemon, 2002b).

2.3 The front end of innovation

2.3.1 Front end activities

To make the innovation process and NPD process more manageable, development projects and processes are typically divided into various stages separated by go - no-go gates (Cooper, 1990; Cooper & Kleinschmidt, 1995). The early phase of the product development process has been called the front end or fuzzy front end since the mid 1980's (Reinertsen, 1985). It has then received wider attention from 1990s onwards (see e.g. Khurana & Rosenthal, 1998; Kim & Wilemon, 2002a; Nobelius & Trygg, 2002; Poskela & Martinsuo, 2009; Reid & de Brentani, 2004; Reinertsen, 1999).

Research has found that by enhancing the decision making process in the early – front end – phases of innovation could have a substantial effect on the success of innovation (Brentani, 1986; Calantone et al., 1999; Cooper & Kleinschmidt, 1987; Rosenau Jr. et al., 1996; Stevens & Burley, 2003; Verganti, 1997). It has also been shown in the previous research that once a development project is launched it is difficult for managers to end it (Balachandra, 1984; Boulding et al., 1997; Cooper, 1994; Schmidt & Calantone, 1998; Schmidt et al., 2001). This accentuates the importance of decision making at the front end of the development projects.

At the front end, the screening phase is where the decisions about whether or not to invest in a new idea for further development are made. The screening phase is challenging in many ways. The results of the development can only be evaluated when they are ready, therefore uncertainty exists during the decision phase. Much of the information concerning the market, competition etc. do not exist at the time of the screening decision and that creates uncertainty and "fuzziness" (Brentani & Reid, 2012; Davila, 2000; Kim & Wilemon, 2002a, b; Nobelius & Trygg, 2002; Zhang & Doll, 2001).

Even though the importance of decision making is highlighted in the previous research of the innovation front end, the terminology and descriptions of the front end phase has not stabilized. At least five different descriptions of the front end can be found in the literature (Table 2.6): The stage-gate model, the new concept development model, the funnel model, the holistic approach and the tailored model (Poskela, 2009).

Table 2.6: Front end process models

Front end pro-	Key features	Authors
cess model		
Stage gate model	Three phases (ideation, preliminary investiga- tion, business case) and three decision gates (initial qualitative screening, second quantitative screening and selecting on business case).	(Cooper, 1990, 1998, 2008) (Khurana & Rosenthal, 1998; Kim & Wilemon, 2002a)
	Many applications based on the basic stage gate model (number and naming of the phases)	
New concept development model	Five front end elements (opportunity identifica- tion, opportunity analysis, idea genesis, idea selection, concept and technology develop- ment), the engine and external influencing fac- tors. Focus on the element instead of the pro- cesses. The Iterative nature of front end high- lighted.	(Koen et al., 2001)
Funnel model	A series of funnels consisting of three phases (identify, understand and conceptualize). Deci- sion making gates between the funnels.	(Cagan & Vogel, 2002)
Holistic approach	Strategic level foundations (product strategy, portfolio management, organization) form the foundation for project level activities (product concept, product requirements, market require- ments, plans, schedules and resource esti- mates) to succeed. Decisions between the gates, management has an integrating role leading to the holistic approach.	(Khurana & Rosenthal, 1997, 1998)
Tailored models	The model by Nobelius and Trygg includes the elements: mission statement, concept genera- tion, concept screening, concept definition, busi- ness analysis, and project planning. The se- quence and durations of these elements are tailored based on the front end case at hand.	(Nobelius & Trygg, 2002) (Reid & de Brentani, 2004)
	The model by Reid and de Brentani has three process interfaces (boundary, gatekeeping, and project). The decision maker and the information to and from the interfaces differ based on the decision that are to be made.	

One of the first descriptions of pre project phase activities is from Cooper (Cooper, 1990, 1998, 2008). He identified three activities in the front end phase stage-gate model: Idea generation, preliminary assessment and concept testing. The purpose of idea generation is to produce ideas and conceptualize them to be assessed in the assessment phase. If the investigation gate in the assessment is passed the concept testing phase concentrates on building a solid business case and an action and launch plan for the product. Several applications from this basic stage gate model with different numbers or names are found in the research (see e.g. Khurana & Rosenthal, 1998; Kim & Wilemon, 2002a).

The new concept development model by Koen et al. (2001) is more iterative in nature than the sequential stage-gate process model. It consists of five phases (opportunity identification, opportunity analysis, idea genesis, idea selection and concept and technology development) (Koen et al., 2001). The model has even more emphasis on the pre-idea phase (opportunity identification and analysis) and it tries to enable the natural back and forth iteration which is typical during the front end phase. The latter phases (idea selection and concept development) are more similar than the previously described stage-gate model.

The front end phase can also be organized as a set of funnels each having a distinct phases (identify, understand and conceptualize) (Cagan & Vogel, 2002). The basic idea of the funnel model is that the ideas that are first filtered through each funnel are then expanded as a source for the following funnel. Decision gates exist between each funnel where the most promising items are forwarded for the next phase.

A holistic approach of the front end process model creates the difference between strategic level elements and tactical level elements (project specific elements) (Khurana & Rosenthal, 1997, 1998). Strategic level foundations (product strategy, portfolio management, organization) form the foundation for project level activities (product concept, product requirements, market requirements, plans, schedules and resource estimates) to succeed. Decisions are made between the phases and management has the integrating role in building a holistic picture between the strategic and tactical elements.

Tailored models include a set of elements which are tailored based on each of the front end cases. Tailoring is done from the foundation of the previously mentioned front end models (typically stage gate model). There are several examples of tailored front end models. Two of them are described below. The model by Nobelius and Trygg (Nobelius & Trygg, 2002) includes the following elements: Mission statement, concept generation, concept screening, concept definition, business analysis, and project planning. The sequence and duration of these elements are then tailored based on the front end case at hand. Management has the critical role in making and tailoring decisions at each decision making gate. The tailored model by Reid and de Brentani describes the front end model by concentrating on the interfaces between the phases and decision making occurring over the interfaces (Reid & de Brentani, 2004). They focus on decision making and on front end process and the focus is on how the decision making and the decision maker (individual or organization) differs between phases.

2.3.2 Decision making at the front end

As stated in earlier chapters, the front end of innovation has in nature the level of uncertainty and "fuzziness" (Brentani & Reid, 2012; Kim & Wilemon, 2002a; Nobelius & Trygg, 2002). There are for example technological and business uncertainties but the decisions about taking an idea further still have to be made. Decision making concerning innovation management in general can be regarded as decision making under uncertainty, especially during the front end phase (Khurana & Rosenthal, 1998; Kim & Wilemon, 2002b).

Different methods supporting decision making at the NPD front end have been developed to overcome uncertainty related problems. Decision making models found in the literature are summarized in the Table 2.7.

Front end de- cision making model	Features and use of the model	Authors
Probabilistic models	Probabilistic models uses probabilities and ex- pected outcomes (e.g. revenue) to calculate possible solutions to a problem. Monte Carlo is a process simulation tool.	(Chapman & Ward, 1997; Souder & Mandakovic, 1986)
	Decision trees to understand and evaluate the path from inception to completion of a project.	(Doctor et al., 2001)
Formal scoring methods	In scoring models projects are rated and scored using qualitative questions and criteria.	(Cooper, 1988; Hart et al., 2003; Montoya-Weiss & Calantone, 1994; Tzokas et al., 2004)
Behavioral approaches	Behavioral methods like the DELPHI obtain the most reliable consensus of opinion of a group of experts. Reflective learning and sense-making models rely on individuals' ability to learn and make sense of the information. Trial and error learning methods.	(Dalkey & Helmer, 1963; McGrath, 2001; O'Connor & Veryzer, 2001; Ortt & Smits, 2006; Perminova et al., 2008; Souder & Mandakovic, 1986; Thomke & Reinertsen, 1998; Van de Ven et al., 1999; Weick, 1995)
Analytical hier- archy process (AHP)	Customized decision tree based tool which uses pairing of criteria and ideas as a basis for scor- ing.	(Calantone et al., 1999; Saaty, 1977)
Fuzzy logic	Fuzzy logic provides the framework to evaluate the possibility of events rather than probability	(Büyüközkan & Feyzioglu, 2004)

The use of intuition has received only minor attention in the prevailing research as was explained earlier in this chapter. Typically, uncertainty in decision making has been handled using traditional risk management methods by assessing the occurrences, identifying probabilities and designing mitigation strategies (see e.g. Chapman & Ward, 1997). A Monte Carlo simulation and decision trees are examples of probabilistic decision models (Souder & Mandakovic, 1986). Monte Carlo uses simulation algorithms to find a solution for a decision problem. Decision trees provide a structured approach for decision making using expected revenue and uncertainty factors (Doctor et al., 2001; Magee, 1964).

Formal scoring methods have been popular in order to manage the decision making in the screening phase (Cooper, 1988; Hart et al., 2003; Montoya-Weiss & Calantone, 1994; Tzokas et al., 2004). In scoring, the proposals are evaluated against a set of criteria such as: Fit to the strategy, available resources, available technology and financial resources. These studies have concentrated on enhancing the rationality of information collection and analysis.

Behavioral approaches such as the Deplhi method finds a consensus within a team about which projects or decisions to move forward with (Dalkey & Helmer, 1963; Souder & Mandakovic, 1986). It is particularly useful in the early phases where the qualitative information is the main information source. Methods using a trial and error learning and selectionism have also been researched as a management method when dealing with uncertain conditions (McGrath, 2001; O'Connor & Veryzer, 2001; Thomke & Reinertsen, 1998; Van de Ven et al., 1999). Uncertainty can be also managed by means of reflective learning and sense making (Ortt & Smits, 2006; Perminova et al., 2008; Weick, 1995). The way in which decision makers make sense of a situation greatly affects the alternatives that are selected. Continuously reflective learning reduces uncertainty by illuminating learning opportunities (Ortt & Smits, 2006; Perminova et al., 2008).

Analytical Hierarchy Process (AHP) method and tools uses the pairing of criteria and the product idea for selecting the most promising ideas for further processing (Calantone et al., 1999; Saaty, 1977). AHP relies on subjective managerial input on multiple criteria and scorings. Many commercial decision making software tools use AHP models. As the name already states the method is based on analytical evaluation of the criteria.

Fuzzy logic deals with the problems where a source of vagueness is involved (Büyüközkan & Feyzioglu, 2004; Zadeh, 1965). In general, the often used probability concept is related to the frequency of the occurrence of events, while fuzzy sets provide the appropriate framework to evaluate the possibility of events rather than their probability. Fuzzy logic based control has been widely used in different production systems in the industry. The same has been applied to NPD decision support software (Büyüközkan & Feyzioglu, 2004).

Rationality based methods as described above can lead to good and effective decisions. But what if the uncertainty and risk cannot be mitigated or reduced by using the rationality based decision making methods which prevalent in the research? The basic deficiency of all of the previously described models is that they assume more information and better information for analyzing and solving uncertainty related problems at front end decision making. In some cases that approach certainly has its place but it does not explain the decision making and all of its components. However, when outcomes are difficult to predict through rational analysis and the uncertainties have to be acknowledged and one must be able to face the unknown in a fast way, then the potential of intuition must be recognized.

2.4 Intuition in new product development front end

Even though there are some studies which report that intuitive judgment is used in NPD context (Dayan & Di Benedetto, 2011; Hart et al., 2003; Stevens & Burley, 2003; Tzokas et al., 2004; Yahaya & Abu-Bakar, 2007) the results vary significantly between the studies. Some use quantitative survey based research methods (Dayan & Di Benedetto, 2011; Hart et al., 2003; Tzokas et al., 2004) others use qualitative methods (Yahaya & Abu-Bakar, 2007) and others use some existing instruments (Stevens & Burley, 2003). As was noted in the basic definition of intuition, the studies differ in their definitions of intuition and intuitive decision making which makes it even more difficult to compare the results. Table 2.8 summarizes the recent empirical NPD studies where intuition has also a research target. Two studies from the New Business Development (NBD) and Technology Based Services (TBS) areas where also included in the sample, since the way they handled the development process is similar to the NPD descriptions and the findings can be in part taken as a reference for this dissertation.

Authors	Empirical data	Research method	Findings
(Hart et al., 2003; Tzokas et al., 2004)	228+438 com- panies in Neth- erlands and UK. 166 responses.	Survey study. The whole NPD process.	Intuition plays a major role in screening gate decisions (after technical feasibility, product uniqueness and market potential).
(Yahaya & Abu- Bakar, 2007)	6 companies, 16 managers	Interview data, the grounded theory method.	Intuition is used in NPD deci- sion making in strategic man- agement issues and in pro- cess issues.
(Stevens & Burley, 2003)	267 projects in chemical com- panies. (NDB)	Interview data and MBTI in- dexes.	High N(intuition) and T(thinking) persons deliver better business results for new business development (NBD). They also work more effective- ly in FFE.
(Van Riel et al., 2011)	251 innovation projects. (TBS)	Survey ques- tionnaire.	The importance of team com- position, information usage and decision perspective.
(Dayan & Di Benedetto, 2011)	395 firms in Turkey,155 responses.	Survey study and interviews.	The more turbulent environ- ment the more teams use intuition. Stress has a moder- ating effect on the use of intui- tion.

Table 2.8: Recent intuition studies in NPD (NBD, TBS) context

Intuition has been found to be one criteria used especially during the early screening gates of the NPD process (Hart et al., 2003; Tzokas et al., 2004). Hart et al. (2003) conducted a survey questionnaire based study of 166 managers in the UK and Netherlands. The companies all developed or manufactured industrial goods. The results (for British and Dutch companies) show that intuition plays a major role in the idea screening gate decisions (technical feasibility, product uniqueness and market potential still played a more significant role than intuition). Intuition also has a role in subsequent gate decisions, but other criteria were ranked much higher. The responses about the use of criteria varied between the respondent companies, but in general, intuition had a remarkable position in screening gate decisions.

The study by Hart et al. (2003) is based on the survey questionnaire responses from the managers. As a self-report it has the potential for bias either towards good or bad results. They only evaluate the use of intuition, not how the managers used intuition in decision making (what approaches decision makers had). The sample in their study was randomly selected in company level. They did not report how the respondents of the survey were selected. The demographics show respondents with many years of experience and also those with less experience. However, no difference in the use of intuition was observed based on the experience. They also encourage more research on the importance of the different criteria. It could for example be that even though intuition has not received many scores in other than screening-gate decisions its importance could be high in other gate decisions too.

Similar findings are presented by Yahaya and Abu-Bakar (2007). In their study of six Malaysian technology based companies they found that senior managers use intuition in some situations during the NPD process, especially in dealing with strategic management issues and NPD process issues. Their study is based on interviews of 16 managers from 6 technology based companies. The majority of the respondents have a long (16+ years) work history, so they can be regarded as experienced. They used Grounded Theory as data analysis method in the study.

The use of intuition was revealed in Yahaya and Abu-Bakar's (2007) study from the descriptions the managers gave during the interviews. However, no further description or analysis of the phenomenon was made in the study. The results lack in their ability to explain how managers use intuition in decision making and if there are any individual differences in the use of intuition. They also propose more research to be done in the area of senior decision makers approach to different decision making situations where intuition could be used. Junior decision maker perspectives are also left out in their data and analysis.

A study by Stevens and Burley (2003) evaluated the personalities in new a business development (NBD) context. Even though the context differs from NPD the results are at least partially usable for NPD. A Myers Briggs Type Indicator (MBTI) (Briggs & Myers, 1962) was used as a tool to analyze the personalities of the persons in the NBD projects. They developed a Rainmaker Index (RI) to describe the personality type that is able to provide higher profits than others involved in NBD projects. RI is a direct application from the MBTI NT (iNtuition/Thinking) score modified to take profit generating factors into account. MBTI personality measurement has inbuilt scores for intuitive (N) preferences and that can be used to reveal the deci-

sion makers use of intuition. Stevens and Burley used an interview data from one company over a ten year period. MBTI measurement was done for 69 analysts working on NBD projects during the research period. It is assumed that all the measured analysts were experienced decision makers.

Their findings conclude that intuition (based on the RI index) is the most important success factor for the early stages of the NBD process and that the people are at least as important as the NBD process itself (Stevens & Burley, 2003). Creative – intuitive – individuals as decision makers have been found to be superior at NPD front end decisions. They also point out that higher RI index produces higher profits in NBD projects. However, they did not specify intuitive persons decision making capabilities in more detail.

A recent study from the service development field (technology based services TBS) by van Riel et al. (2011) touches directly the screening phase of innovation projects and decision making effectiveness. A survey questionnaire was sent to 1500 companies and 251 usable answers from senior executives or managers were received. The data in their study shows that operative managers participate more in the screening phase decisions than do CEO or executive level persons.

Their study did not directly discuss the use of intuition in screening phase decision making. The results show that acquiring information plays a major role in screening phase decisions. Gathering information about the external environment was found to have a substantial effect on decision making effectiveness. A strategic entrepreneurial attitude and experience was also found to have positive influence on effectiveness. The authors highlight that the results of their study are limited to some extent by the internal information of the decision maker, which means the intuition and experience of the decision maker. That's why it was not possible to study using quantitative survey based methods and thus the need for qualitative research is addressed in their report.

Another recent study by Dayan and Di Benedetto (2011), which investigates team intuition in NPD project teams, also puts intuition as a main component of decision making. They used a sample of 310 product developers and 155 managers in their quantitative study. They developed their own measures to evaluate the team level use of intuition. The results show that the more turbulent the environment the more intuitive judgment is used by the decision making teams. Stress has a moderating role in use of intuition and creativity based on their results. Their study was based on quantitative survey data analysis and was supported by interviews.

Dayan and Di Benedetto's (2011) definition of intuitive-rational decision making as a continuum with two ends conflicts with the view that intuitive and rational components can work simultaneously, which is the definition used in this dissertation. Their view is also limited to the team context and does not analyze decision maker's personal approach to the use of intuition. The measures what they used, even though well developed, have not been used anywhere else. Thus the comparability of the results still needs more research.

The essence of the use of intuition at the front end of innovation remains unexplained in the aforementioned studies. Intuition is seen as a decision making component, but the detailed investigation of the phenomena remains weak. Even though some of the studies have also used qualitative data and analysis, the argument continues about its use at higher levels (i.e. is intuition used or not). How intuition is used by decision makers (what approaches decision makers have) has not been the research focus in any of the empirical research what I have been able to locate. The need for future qualitative research is clearly stated in the conclusion of several previous studies. The argument is that the use of intuition has a major role in innovation decision making, especially on the NPD front end phase and that needs more detailed study and explanation.

2.5 Summary and conclusions of literature review

Decision making at the front end of innovation has been regarded as one of the most important factors in innovation management (Abernathy & Utterback, 1978; Brown & Eisenhardt, 1995; Cooper & Kleinschmidt, 1987, 1995; Hart et al., 2003; Khurana & Rosenthal, 1998; Reid & de Brentani, 2004; Schmidt et al., 2001; Smits, 2002; Tzokas et al., 2004). When looking at the literature on innovation management and new product develop-

ment the majority of the research deals with innovation process descriptions and rational decision making methods. My argument is that the challenge of the innovation front end decision making is more complex because it is not just selecting the best from the available options and decision making cannot rely solely on rational and analytical methods which prevail in innovation management and NPD front end literature.

Innovation is about the unknown, about possibilities associated with doing something new and thus the process involves dealing with uncertainty. There has to be a means of dealing with the uncertainty. Intuition has been found to be a component of decision making when conditions are uncertain or ambiguous (see e.g. Kahneman & Klein, 2009; Kahneman & Tversky, 1982; Tversky & Kahneman, 1974) and Table 2.4). Especially in radical innovations (Christensen, 2003; Leifer et al., 2000), in high velocity environments (Eisenhardt, 1989c) and at the front end of the innovation process (Khurana & Rosenthal, 1998; Kim & Wilemon, 2002a) the uncertainty in decision making is high.

Table 2.9 summarizes the present research in connection with the research questions of this study. It should be noted, that even though there are number of studies who have reported how intuition is used and what approaches managers have to using intuition, there are very few – if any that have had the context of innovation front end in their research as a main focus area.

Research guestion.	Selected references	Main arguments	Current research gap
1. How does intuition reveal itself in innovation front end decision making?	(Myers, 2004; Scott & Bruce, 1995; Shapiro & Spence, 1997)	Indirect evidence is available from body cues, info filtering, etc.	No clear ways of describ- ing the use of intuition. Empirical evidence scarce.
	(Adams et al., 2006; Brown & Eisenhardt, 1995; Cooper, 1999; Hart et al., 2003; Krishnan & Ulrich, 2001; Liberatone & Stylianou, 1995; Reid & de Brentani, 2004; Sinclair & Ashkanasy, 2005; Tzokas et al., 2004)	Front end is the critical process phase. Decision making process needs more focus, different processes are needed. Fast decision making leads to better per- formance.	More empirical evidence is needed in process side. Development has been mainly in analytical decision making pro- cesses. More and better instru- ments needed in order to study the use of intuition.
2. What ap- proaches do decision makers have when using intuition at the innovation front end?	(Gemunden et al., 2007; Isenberg, 1984; Kahneman & Klein, 2009; Kim & Wilemon, 2002a; Pretz & Totz, 2007; Tushman & Katz, 1980; Van Riel et al., 2011; Zirger & Maidique, 1990)	Use of intuition is case dependent. Use of intuition is a personal quality and related to individual roles.	Empirical evidence is mixed and mainly based on quantitative studies. How individuals end up with a decision is not studied. How intuition is related to different roles at the front end has not been re- searched.
3. How do experi- enced decision makers differ from inexperienced decision makers in their use of intuition?	(Dane & Pratt, 2007; Hart et al., 2003; Sadler-Smith & Shefy, 2004; Stevens & Burley, 2003; Yahaya & Abu-Bakar, 2007)	Experienced deci- sion makers use intuition.	Does not study inexperi- enced decision makers use of intuition or empiri- cally compared the dif- ferences based on the experience. Self-report studies are prone to bias.

Table 2.9: Summary of present research in connection with the research questions

The innovation front end has been regarded as the critical process phase concerning decision making and the development of front end decision making has been requested by the existing research. Development of those decision making practices have mainly concentrated on building better rational and analytical tools to support decision making research has also been raised as an innovation front end research arena. The use of intuition at the innovation front end remains a relatively untouched topic (with the exception of: Reid & de Brentani, 2004; Stevens & Burley, 2003; Tzokas et al., 2004; Yahaya & Abu-Bakar, 2007). This study adds empirics in the form of a qualitative study and analyses especially innovation front end context in high velocity NPD environments.

Use of intuition is related to decision process structure where the role of intuition can be studied. However, the reliability of these surveys has been recently questioned (Hodgkinson et al., 2008; Sinclair et al., 2009). Use of intuition in the decision process has also been studied and found to be remarkable, however the description of how intuition presents itself in the process and what affects its usage still needs more study.

A study of intuition can be targeted to study individual decision maker characteristics where one part is the level of intuitiveness of the decision maker. Current empirical evidence of the subject is mixed and mainly based on quantitative studies. This study complements and challenges current innovation front end research by describing how the use of intuition reveals itself in an NPD context and describes what approaches the decision maker has in the use of intuition in the innovation front end. This is to shed more light on what innovation and NPD management is like – the actual management work.

The experience of the decision maker has been found out to have an effect on the use of intuition, so that experienced decision makers use more intuition or come to better results when using intuition than their less experienced colleagues. However, these results have mainly been accomplished by studying only experienced decision makers (Sadler-Smith & Shefy, 2004; Stevens & Burley, 2003; Yahaya & Abu-Bakar, 2007). This study also takes into account a picture of less experienced decision makers in comparison with the experienced ones.

This study uses two different instruments (qualitative and quantitative) to study intuition in the process and in individuals with intuitive abilities. This is to shed more light in the inquiry of instrumental validity highlighted by Sinclair (Sinclair et al., 2009). This also has an effect on the methodological side of this research which is described later in methodology section.

This research concentrates on discussing decision making processes on a conscious or unconscious level (rational/analytical, intuitive). The supraconscious level, sixth sense etc. higher level of consciousness is outside the scope of this study since there are no scientific tools available to study its existence (for further details see Sinclair & Ashkanasy, 2005).

3 Research design and methods

"To achieve something you have to have <u>dreams</u> and goals and <u>courage</u> to start the journey without knowing how much it requires effort and how to achieve the goals and <u>persistence</u> to continue despite of the difficulties and obstacles on the way. " Olli Hyppänen

This chapter explains in detail how the empirical research for this dissertation was done. It also presents the research strategy, approach and methods used in this dissertation and how the methods were used during the course of the research. Finally the evaluation of the validity and reliability of the research as well as the researcher's role is stated.

3.1 Objective and research questions

The aim of this research has been to gain deeper understanding of decision makers' use of intuition and to offer new theoretical end empirical insights into the use of intuition in decision making at the front end of innovation.

The research questions are:

- 1. How does intuition reveal itself in innovation front end decision making?
- 2. What approaches do decision makers have when using intuition at the innovation front end?

3. How do experienced decision makers differ from inexperienced decision makers in their use of intuition?

The literature review in the previous chapter clarified that decision making is one of the key areas in innovation front end and new product development. Literature review also points out that decision making has been previously researched mainly from rational analysis point of view and the non-rational (use of intuition) part is largely missing. Also empirical evidence for the use of intuition in decision making is quite limited, especially concerning NPD front end (Reid & de Brentani, 2004; Stevens & Burley, 2003; Tzokas et al., 2004; Yahaya & Abu-Bakar, 2007).

The purpose of this study is to explore the front end of innovation, to study how decisions concerning innovation projects are made, who make the decisions and how the decisions are arrived at by the decision makers. Since few previous studies have approached these issues from a nonrational decision making in NPD stand point, the main part of this study was done using qualitative – inductive – methods.

Mixed interview and survey questionnaire is used as a research method in this study. The researcher – myself – as the main research instrument, observed what happened in the innovation management and decision making process by conducting interviews and thoroughly analyzing the interview data. I also conducted the survey questionnaire with a separate control group. However the results of the survey questionnaire represent only a minor share of the results of this study. The main argument is built from the interview data. The research methods provided a large amount of qualitative interview data and quantitative survey questionnaire data. A detailed description of the data is found in the following chapters.

3.2 Research approach

The research approach using mainly qualitative interview data was selected because the research interest was in studying a complex phenomenon with a tight connection to the field and actual work. Qualitative approach makes it possible to study a complex phenomenon where the available previous research data is nonexistent or limited (Charmaz, 2006; Denzin & Lincoln, 1994; Miles & Huberman, 1994). It also allows the researcher to stay in close connection with the field and the data and to interpret and construct the meaning from the data. Data analysis is done by using a grounded theory method and statistical analysis (factor analysis and t-test) for the survey questionnaire data. The use of data analysis methods in this study is explained in more detail in following chapters.

Grounded theory as a method of inquiry was selected because the field of decision making and the use of intuition in decision making is a relatively unstructured and unknown field. Thus the grounded theory approach with its iterative data analysis and inductive method is well suited to this study. A unit of analysis in this dissertation is the use of intuition in product development project front end decision making. Emphasis is given to the use of intuition in engineering related innovation management decisions – decisions which are made by individuals. A single project and individual level approach enables rich data and thus more truthful results, since the interview data was able to handle the front end activities in a detailed and focused way (Charmaz, 2006).

The grounded theory was developed by Barney G. Glaser and Anselm L. Strauss during their studies of dying in hospitals (Glaser & Strauss, 1965; Glaser & Strauss, 1967). When they analyzed the process of dying they developed a systematic method of studying many other topics. The method was published as a book "The Discovery of Grounded Theory"(Glaser & Strauss, 1967). The research method favored developing theories from research grounded in data rather than deducing hypotheses from existing theories and testing those. The components of the grounded theory practice include:(Charmaz, 2006; Glaser & Strauss, 1967)

- Simultaneous involvement in data collection and analysis
- Constructing analytic codes and categories from data (not from deduced hypotheses)
- Using the constant comparative method, which involves making comparisons during each stage of the analysis
- Advancing theory development during each step of data collection and analysis
- Memo writing to elaborate upon categories, properties, relationships and gaps

- Sampling aimed toward theory construction (not for population representativeness)
- Conducting literature review after independent analysis

A grounded theory approach sheds light on problems or areas of study through the processes and humans are seen as active agents in their worlds. Social meanings are created by the use of language and emerge through action. Structures are created by processes where human beings are active. These ideas reflect the pragmatism tradition favored by Strauss. This approach also reflects my own personal philosophical stance. I see people as constructing their realities based on how they participate in it. In the same way phenomena in the innovation management is a construct itself. Due to the nature of my data and the method I use, the study has an interpretative and subjectivist nature.

In this study, I will interpret how meanings and actions are constructed in the decision making of NPD front end. The decision making context itself appears to be such a complex area of research that I could not see any way of studying it using a positivist - objectivist approach. My own participation and insight in the innovation field also limits the possibility of treating the data as purely objective facts about the knowable world in the way that the objectivist approach would require. On the other hand, my previous experience in innovation management and tight participation to the innovation management process gives this study an ethnographic stance also. I was one of the main research instruments of the study.

The grounded theory method used in this research follows closely the original grounded theory by Glaser and Strauss (Glaser & Strauss, 1967). My grounded theory approach is not as purely objective as Glaser suggested (Glaser, 1978). Instead I see the use of grounded theory in more of a constructivist - interpretive way. The researcher is within the process not outside or above it. It is to be noted that in their original statement of method Glaser and Strauss left some room for the researcher to use the method flexibly (Glaser & Strauss, 1967). I have used some of the developments during the past four decades and use as a guide the Charmaz's latest method description of the grounded theory approach (Charmaz, 2006). I examine the processes and have a study of action as central during the analysis.

There are several possible research strategies within the qualitative approach to conduct the research and analyze the data. I used a mixed interview and survey research method with a grounded theory analysis. This approach was selected because it fits well with the researchers' experience and background and fits with the phenomenon under study. It also requires close long term involvement with the phenomenon. The empirical data is approached through several stages, called in this study as encounters, which all focus on separate aspect of the studied phenomena.

3.3 Research procedure

This research journey began in early 2006 with an interest in the relationship between research & development (R&D) and the customer management processes. Customers are in close contact with R&D and product development in the company where I worked at that time which raised the question of studying the relationship in more detail.

Customer – R&D interaction soon revealed the concept of user oriented design, user based innovation and lead users as the paths to follow. However, when I began the data collection I decided to stay at a more general level in order to avoid jumping to conclusion and to ensure detailed data analysis and collection. The concrete research began in late 2006 with the first set of data collection. The first data collection concentrated on innovation management in general.

Interviews were lightly structured and allowed the discussion to flow into any and all innovation management subjects. I was already confident of the need for close relationships with the customer in product development so the first path to follow was innovation and customer relations. This is seen in the first set of interviews. The preliminary coding and analysis of the interviews was done using handwritten notes and transcribed interview memos. Categories at the top of the papers included "customer involvement" and "customer orientation". During the first coding and analysis the unit of analysis was the innovation project. I was trying to understand what was happening at the process level. The second data gathering was done in mid 2007 with a sample from the second company. The data analysis on the first set of interviews was complete. The same categories were found but also the concept of intuition with the form of "decision by feeling" was raised for the first time. My interest started to shift towards decision making - "How do these decisions actually happen in innovation management?". These first two interview data sets are from the same industry and from the same kind of companies. I felt that I needed to get a broader view and understanding on the subject of innovation management, especially concerning the decisions made at the front end.

The third and the fourth company and data sets are from the telecom sector. At the same time the unit of analysis shifted from the process level to include the individual decision makers in the innovation front end. The third data collection was done in late 2007 with a sample from a third company.

During 2007 the shift in interest and research was moving towards decision making. This is visible in second and third data gathering sessions. I gathered also some more theoretical knowledge about decision making and returned to the field in late 2007.

The fourth data collection was done in early 2008. I concentrated on innovation front end decision making in the interviews and dug deeper in the use of intuition. How the use of intuition reveals itself at the front end, how it is used, what approaches decision makers have in the use of intuition, when it is used etc. However the basic structure or actually the lack of structure is what remained in the interview agendas in order to avoid influencing the responses. The interviews and transcripts for the fourth interview round were the most extensive. Also support of previous findings was encouraging.

During the research I wrote initial memos at all times in addition to interview notes and interview transcripts. Memos started to have more data and form when the data was re-analyzed in each data collection round. During 2008 the first theoretical concepts started to surface and the first encounter with the data from an innovation management perspective was developed and the first drafts on the research results were completed. The first encounter with the full data revealed the importance of decision making from a fresh viewpoint and sparked a second encounter with a new perspective based on the use of intuition. During the second encounter the same procedures used in writing memos and drafts continued. Late 2008 the first integration of the research results and written drafts were completed.

Since the concretizing of the use of intuition is challenging I felt that more supportive data was needed. I was also interested in studying how experienced and inexperienced decision makers differ in their use of intuition in decision making. I conducted an additional survey questionnaire to verify the use of intuition at an individual – decision maker – level. The survey supported my previous findings based on the interview data and revealed some new viewpoints on the phenomenon under study concerning the experience of the decision maker. At the same time I had a third encounter with the interview data from the perspective of a decision maker using intuition.

The year 2009 was spent in understanding the research results, writing the concepts and drafts and integrating and elaborating the results into a coherent form. The theoretical side to the dissertation was started at this time and the first drafts of the full dissertation manuscripts were done. From 2010 onwards the manuscript with full argumentation was developed further and the theoretical argumentation based on the existing literature was created. From then on phases concentrated on developing the manuscript further with an enhanced discussion, argumentation and conclusion.

The research procedure is visualized in the Figures 3.1 and 3.2.



Figure 3.1: Research procedure. Interview data.



Figure 3.2: Research procedure. Survey data.

3.4 Interview data collection and analysis

The primary data collection and analysis was done based on interview material from 19 experienced decision makers. The decision makers were from four separate ICT companies. The companies were selected as representing high velocity companies with high innovation activities so the data would represent innovative companies operating in rapidly developing environments. It is assumed that the data from these companies provides a good representation of innovative operations since the ICT industry has been judged as having highly innovative abilities over the past few years (Christensen & Raynor, 2003).

The software and telecom industries are typically rated as high velocity industries. This study further enhances the data available about decision making in high velocity companies (Eisenhardt & Zbaracki, 1992). The high innovativeness was rated by looking at the proportion of R&D expenditure related to revenue. Three of the selected companies were spending more than 10% of revenue to R&D activities, the last of the companies has one of the largest R&D operations in the world and thus can be considered also innovative.

All of the four companies are product development companies. The companies operate in high velocity markets and environments. One of the companies is a Finnish software company, one is globally operating software company and the rest two are globally operating telecom companies. The second of the two companies were selected based on theoretical sampling. In theoretical sampling the cases are selected so that they are particularly suitable for illuminating and extending the studied phenomenon (Eisenhardt, 1989b). The selected companies had to operate in the same industry and operate in new product development. They also should have some similarities in their innovation processes. This poses some limitations on the representativeness and generalization of the research findings but that will remain one of the characteristics of this study.

3.4.1 Company demographics

The first company that was analyzed had a long history as a Finnish software company. The company was established in mid 80s. Currently the company is a medium size international software company (~750 employees in 2008). The company has experienced extensive growth over the past several years. The pseudonym used for this company is Alfa. Alfa develops financial software for purchase to pay processes. The product portfolio is large covering the large part of the process area. Alfa has been a pioneer in electronic invoice handling and delivery in the world.

The second company, named Beta, has also a long history in financial sector software development. The company has mainly developed solutions for handling payment transactions between companies and banks. Later the product portfolio was developed to also include other parts of the financial processes. Beta has a solid track record of high quality software development for the financial sector in Finland. It has basically no revenue outside of Finland. It is a small-medium size Finnish software company (~100 employees). Beta had a very large R&D department accounting for approxi-
mately 50% of the personnel. The company was also tightly process oriented and controlled by an extensive quality system.

The second set of the companies consists of two telecom sector companies. The first one, named Gamma, develops telecom equipment and related software. Gamma is a global company with several thousand employees. It has operations in several continents and countries, including Finland. Gamma has a decades long history in the telecom sector and it has pioneered several data communication innovations globally. Gamma is known from its solid engineering innovation background – technological inventions and innovations. Gamma has successfully transformed itself with the telecom sector transformation during the last two decades.

The fourth company, Delta, is a very large multinational ICT company. It has tens of thousands of employees globally. Company is one of the largest R&D investors in the world. A high level of research and study activities is thus evident. Lots of effort is put into studying the environment, consumers, competitors, technologies and the trends within the industry. Delta has undergone several transformations during its lifetime. The boom of the mobile communications was one of the largest. Delta managed to invent and innovate successfully during the takeoff of the mobile industry and exceeded the innovation abilities of all of its competitors at that time. The company and interview demographics is shown in Table 3.1.

Table 3.1:	Company	interview	demographics
------------	---------	-----------	--------------

Company	Software	Telco	Empl. Size	# interviews (persons)
Alfa	х		Hundreds	6
Beta	Х		One hundred	5
Gamma		х	Thousands	5
Delta		Х	Tens of thousands	3
				19

The companies are a good representation of the industry. Even though the software companies are small on a global perspective their operations can be extended to include larger global software companies. The selected telecom companies are fair representatives of global telecom companies. The representativeness is limited to the ICT industry but can also be extended to other high velocity environments and industries.

3.4.2 Interviewees

The interviewees were selected so that the interviewed persons are well connected to the decision making of product development and innovations. The selection of the interviewed persons was done by asking the company representatives to nominate key persons who could provide insight into innovation management in practice of the case company. The requirement was that the persons had to have been involved in innovation decision making for several years. The company representative was typically in the position as head of product development or equivalent. All of the interviewed persons have a long history with the company they are involved with and they have been in key roles concerning innovation management and decision making.

The demographics of the interviewed persons are as follows (Table 3.2).

Table 3.2: Demographics of interviewed persons





Education discipline

	Answer	Count
1.	Engineering	7
2.	Science/technology	2
3.	Business/management	8
1.	Other	2
	Total	19

Current function

	Answer	Count
1.	Engineering	4
2.	Science/technology	1
3.	Business/management	12
4.	Other	2
	Total	19

3.4.3 Interview data collection

The interview was selected as a data collection method because the research approach benefited from having direct contact with people intimately involved in the innovation management process. This is a requirement highlighted also by Charmaz (2006). The use of interviews also made possible to steer the data collection in the direction that was of most interest to the researcher. Also the studied phenomenon – intuition or the use of intuition – is vaguely structured and lacks commonly accepted methods and theories and thus the interview data provides the possibility of building theory during the research process. However, the phenomenon under study (the use of intuition) was not known when the data collection method was decided upon.

Interviews were semi-structured – minimal structures were used to guide the discussion inside the innovation management of the case company. No question sets were given to the interviewee nor did the interviewee have to prepare him/herself for the interview beforehand. The interviewees were only aware that the research was on the topic of innovation management. The lack of structure was because the research method calls for early and open involvement in the field to the data. My purpose was also to study how the interviewees see innovation management in practice and the issues related to it without any preconceptions about the purpose of the study.

The first interview in a case company was typically longer than the others. This is because the discussion often followed many different paths concerning innovations and touched on several projects and involved large time scales. I prepared some possible discussion ideas for subsequent interviews based on the first one. Because of this latter interviews were a bit more structured. I took this approach because it helped in taking the right path from the beginning and avoided side steps during the discussion.

In order to guide the interviews in the same direction some basic questions and terms were used. Firstly the term innovation was defined so that all interviewees had a common understanding of the term. Innovation is defined here in comparison to the idea or invention so that innovation is a commercialized invention as described in the literature review. Commercialization has not to be successful but that had been the target.

Also the difference between incremental, radical and disruptive innovations was discussed. Incremental innovation is in question when something that exists in small steps is developed further. Radical innovation refers in this study to a case where something that had not previously existed is found and developed. Disruptive innovation refers to an occasion where the foundations for innovation are disturbed and changed with something that makes innovations in that area obsolete.

The innovation context of a product, process or business innovation is the third categorization used in the discussions. The interviews were not limited to any previously mentioned categories or definitions, or vice versa. The scope was to discuss freely all kinds of innovations where the interviewee had firsthand experience. However, all of the interviews were about product related innovations.

Secondly the interviewee was asked to provide a list of innovations what they have witnessed in the organization. The purpose was to list the names of the innovations for further discussion. Interviewees were also asked to separate successful innovations from unsuccessful ones. Unfortunately the number of unsuccessful innovations discussed was limited, probably people were not as willing to reveal unsuccessful innovations as they were the successful ones. This is a bias towards successful incidents. People likely do not want their own failure is discussed and criticized publically.

After the previously mentioned basic questions the interviews were continued with discussions about the innovations, the reasons behind their success/failure and the process involved in their development (e.g. how the innovations were discussed, handled, decided on and put forward in the organization). All of the interviews focused mainly on the front end phase of innovations, since the discussions always started with descriptions of how the innovation in question started in the company. During the interviews the critical incident technique was used (Flanagan, 1954; Miles & Huberman, 1994). The use of the technique helped to explore the subject more deeply during the initial interview and in the later interviews. Critical incidents were incidents that were considered to have had significant importance in the course of innovation under discussion. When critical incidents surfaced the interviews were guided so that those incidents were more specifically discussed.

The basic structure of the interview remained the same in all cases. The discussion topics were as follows:

- Definitions (term innovation, innovation types)
- Examples of innovations: successes, failures
- Reasons behind the success/failure
- The innovation process: Idea generation, decision making, timescales, structures, person's involved, process description
- Innovativeness in general in the company
- Other possible issues

At the end of the interview the background information of the persons were reviewed (work experience, occupation, position, education). I also asked who else would be willing and able to discuss the innovations.

Even though the basic structure of the discussion topics remained the same, the focus in later interviews was more on the decision making context since the data analysis guided data collection in that direction. I did not bring up intuition as a discussion topic in any interviews on purpose. If the interviewed person mentioned the use of intuition then I guided the discussion in that direction.

Since the use of intuition came up during data analysis I wanted to be careful not to force the data towards the use of intuition. It would have been very easy to start to discuss the use of intuition in decision making, but that would have – in my opinion – guided the data too much. By remaining open to the innovation management process discussion I was able to be sure that the data is as clean as possible. The only change between the structure of the first interview and the structure of the later ones was a focus change towards a decision making context.

The interview data consists of 19 interviews with the length from 1 hour to over 2 hours, with the average being 1.5 hours. All the interviews were recorded and transcribed. Transcriptions were done by a second person (i.e. not the researcher). Transcriptions were 7 to 19 pages in length, the average being 12 pages. Altogether the transcribed and analyzed interview material resulted over 200 pages of written data.

3.4.4 Interview data analysis using grounded theory approach

The interviews were recorded and transcribed as described earlier. Transcripts were pure non-edited text. Before the coding I corrected the transcripts for punctuation mistakes, terminology misunderstandings and coded the paragraphs by the interviewer prefix (#H) and the interviewee prefix (#Yxz, where xz represents the number of the company =x and the number of the interviewed person =z). The transcriptions were also converted to rich text format for use in the data analysis tool.

The transcribed interviews were treated as accounts describing the innovation project where the individual person has been involved. The analysis process tried to identify different elements and their relations inside innovation management and the decision making process based on how interviewed persons had seen it happening.

I started to code the data based on the "what is happening" – action. I did only very limited initial memo writing during the first round of data analysis. In practice I started to write memos during the second round of coding with the help of a new tool. These memos could also be called as narratives describing different cases within the innovation process. This helped to categorize the data and analyze it throughout the memo writing.

During the first encounter and analysis phase I treated the original transcripts of the interviews as narrative descriptions of the "life in a company's' innovation project". I coded the texts based on activities in the process (open coding). An activity based coding was selected because I wanted to explore the process of what is really happening in innovation management and that can be probed by looking at the activities. After each text analysis I linked the new codes to existing ones if possible (focused coding) and wrote initial memos describing the possible relationships and findings of that interview as they related to the previous data and findings. The initial coding was done on a sentence level, not a line-by-line or paragraph level. I selected the sentence level since it was appropriate for getting ahold of the real action. Line-by-line would have been too detailed and paragraph level would not have found the separate actions inside one paragraph. I coded based on "action" to get a grasp of the processes and what is happening in those processes.

At that time I realized that the tools I was using to analyze the data were not sufficient to keep the data in order and to make further analysis based on coding and categories. It was at this time that I started to use the AT-LAS.ti (ATLAS.ti, 2006) tool. This meant that I had to recode all of the material using ATLAS. I tried to use the same basic coding I used during the first coding and analysis round, because I wanted the evolution of my research and thought process to be visible. Naturally some changes were made to the coding, because of easier and more flexible ways of working provided by the ATLAS tool. Going through the same material revealed some new insights into the two first data sets, which were not visible before. These are described in later sections.

During the coding I cross-checked the transcribed text with my own notes which I had taken during the interviews and added some notes to the initial memos which I wrote simultaneously. Usually I coded two interviews at the same time, one after another, and compared the observations and codes between the two interviews. The purpose was to follow a procedure called the constant comparative method as described by Glaser & Strauss (1967). Naturally the coding started with open coding, but after the first interviews I started also to use codes from the list when I found similar to those discussed before. Also some *in vivo* codes were used. When applying an *in vivo* code, the selected text itself is registered as a quotation and used as a code name.

Since I had to code the first two sets of interviews once again after changing the analyze tool, my second coding round which I did for the same data with the new tool, already resembled focused coding. The thought process which was used during the manual - pen and paper - initial coding was not visible the second time. I had however taken these ideas into the initial memos which I wrote during the second coding round. I started to write memos and narratives immediately with the second coding round. I also wrote some notes directly into code explanations in order to grasp the extended meaning of the code and the possible connections on the spot. After the analysis of the first company I started to build networks of codes and memos. The networks were first built up very rapidly just to establish the possible connections between the different ideas, codes and concepts on the paper. From the very beginning I used a flip board extensively to clarify the ideas on the paper and have them visible to keep my thinking focused. The walls of my study sometimes had so full of flip board pages taped to them that nothing else was visible. Continuous flow back and forth with the transcripts and codes was done regularly to keep the code base viable.

After the fourth interview round I did the first full encounter with the data. That was the most extensive analysis during the whole research period. As a result the three main categories emerged. Out of the three core categories, one of the core categories - decision making - was selected as the core category for further analysis.

The second encounter with the data focused on exploring the decision making context in innovation management. The second encounter was done with "intuition glasses" on. The grouping of the codes and the categorization was done by looking at the data from an intuition point of view. It resulted in four core categories to describe how the use of intuition reveals itself in front end decision making.

The third encounter with the data was done based on the roles and properties of the roles that use intuition in decision making. The matrix where codes and interviews were presented was used to categorize the roles. From the matrix the most significant codes were highlighted and an approach for describing which decision makers use intuition (role descriptions) was formed. These were used to describe the properties of different roles.

In order to make sense out of the data I followed several qualitative data analysis tools to support the research process (Miles & Huberman, 1994). From the beginning of the research I started to draw diagrams showing the relationships and connections between the codes and categories. This helped to further the analysis of the new interview data. During the data analysis process lots of cross-tabulation tables were used to uncover patterns or "code concentrations" as a seed for core categories. Extracts from the interview data were used to support the data analysis process and to make the results of the analysis process more transparent to the reader. The use of these methods is made visible in the results chapter of this study.

3.5 Questionnaire data collection and analysis

3.5.1 Questionnaire data description and collection

Secondary data collection and analysis was done in the form of a survey questionnaire in order to get more support for the use of intuition, to verify the findings from the interview data analysis and to answer the third research question. I also judged this to be important since I needed more evidence for the use of intuition in decision making.

The first sample in survey questionnaire consisted of 19 interviewed persons – the interviewees (experienced). 18 responses were received. The second sample consisted of managers with little or no experience and experienced specialists. The purpose was to get a sample of decision makers with little experience in innovation decision making. All the persons in the second sample were selected from two of the interviewed companies, companies Alfa and Beta. This was because access to the other two companies was limited.

The second sample consisted of 93 persons. The response rate with the second sample was 73% (68 persons) (Table 3.3). The number of total respondents to the questionnaire survey was 86 persons (Table 3.4).

The survey was conducted using the web-based tool. Participants were invited to the survey by email where the link to the web form was posted. The survey tool allowed persons freely to answer the posted questions, however respondents were required to answer all questions. Response time varied between 211 and 8331 seconds.

The survey of the interview group was posted on the 5th of November 2008 and closed on the 16th of April 2009. Despite of the several reminders one person did not respond to the survey. The survey of the second group

was posted on the 13^{th} of February 2009 and closed on the 18^{th} of April 2009. Two reminders were sent to the participants, which only slightly increased the response rate.

W	Work experience Years in current organization				
	Answer	Count		Answer	Count
1.	1-5 years	4	1.	1-5 years	42
2.	6-10 years	18	2.	6-10 years	23
3.	11-15 years	17	3.	11-15 years	3
4.	16-20 years	14	4.	16-20 years	0
5.	above 20 years	15	5.	above 20 years	0
	Total	68		Total	68

Table 3.3: Demographics of the control group (inexperienced decision makers)

Hi	ghest educa	ation		Ec	ducation discipline	Current function				
	Answer	Count			Answer	Count			Answer	Count
1.	Bachelor	27		1.	Engineering	18		1.	Engineering	16
2.	Master	34		2.	Science/technology	19		2.	Science/technology	12
3.	Doctorate	0		3.	Business/management	20		3.	Business/management	36
4.	Other	7		4.	Other	11		4.	Other	4
	Total	68			Total	68			Total	68
	Hi 1. 2. 3. 4.	Highest educa Answer 1. Bachelor 2. Master 3. Doctorate 4. Other Total	Highest educationAnswerCount1.Bachelor272.Master343.Doctorate04.Other7Total68	Highest educationAnswerCount1.Bachelor272.Master343.Doctorate04.Other7Total68	Highest educationEducationAnswerCount1.Bachelor272.Master343.Doctorate04.Other7Total68	Highest educationEducation disciplineAnswerCountAnswer1.Bachelor271.Engineering2.Master342.Science/technology3.Doctorate03.Business/management4.Other74.Other	Highest educationEducation disciplineAnswerCountAnswerCount1.Bachelor271.Engineering182.Master342.Science/technology193.Doctorate03.Business/management204.Other74.Other11	Highest educationEducation disciplineAnswerCountAnswerCount1.Bachelor271.Engineering182.Master342.Science/technology193.Doctorate03.Business/management204.Other74.Other11Total68	Highest educationEducation disciplineCAnswerCountAnswerCount1.1.Bachelor271.Engineering181.2.Master342.Science/technology192.3.Doctorate03.Business/management203.4.Other74.Other114.	Highest educationEducation disciplineCurrent functionAnswerCountAnswerCountAnswer1.Bachelor271.Engineering181.Engineering2.Master342.Science/technology192.Science/technology3.Doctorate03.Business/management203.Business/management4.Other74.Other114.OtherTotal68Total68Total68Total

Table 3.4: Demographics of the whole survey group

Work experience				
	Answer	Count		
1.	1-5 years	4		
2.	6-10 years	18		
3.	11-15 years	20		
4.	16-20 years	18		
5.	above 20 years	26		
	Total	86		

1	Years in current organization				
	Answer	Count			
1.	1-5 years	47			
2.	6-10 years	28			
3.	11-15 years	4			
4.	16-20 years	5			
5.	above 20 years	2			
	Total	86			

. .



Education discipline					
	Answer	Count			
1.	Engineering	25			
2.	Science/technology	21			
3.	Business/management	27			
4.	Other	13			
	Total	86			

Current function						
	Answer	Count				
1.	Engineering	20				
2.	Science/technology	13				
3.	Business/management	47				
4.	Other	6				
	Total	86				

Decision makers were well represented in the selected sample. All of the inexperienced decision makers were selected as part of the target control group. They had all have recently received a managerial or chief engineering related position which requires decision making. Even though they were not all directly involved in innovation management, they all worked in the engineering organization.

The interviewed persons were regarded as experienced decision makers based on their long experience in the decision making concerning innovation management. They have all been working for years in positions close to innovations that require constant decision making. The response rate (73%) is also high which further supports the representativeness of the selected sample.

3.5.2 Questionnaire and variables

This study (the analysis of survey questionnaire data) applies the existing research instrument and measurement constructs to study intuitiveness. Research design and methods

The validity of the constructs is evaluated and reported in the results chapter of this study.

Intuition and the use of intuition was further studied in this study using Cognitive Experiential Self Theory (CEST) developed by Epstein (Epstein, 1990, 1994). CEST proposes that there are two information processing systems which people use – experiential and rational. Those systems are independent and operate by different rules. CEST includes intuition as a part of an experiential system that is contrasted with an analytical-rational system. However, CEST indicates that both systems contribute jointly to behavior (Epstein et al., 1996).

The measure instrument for intuition developed by Epstein et al. which is based on CEST is the Rational Experiential Inventory (REI) (Epstein et al., 1996). The REI consists of rational and experiential subscales which are each divided into ability and favorability subscales. The ability scale estimates a person's belief in their own ability to use experiential or rational thinking. The favorability scale reflects a person's preference for engaging in experiential or rational processing. I used the long version of REI (Pacini & Epstein, 1999) for this study. The long version consists of 40 Likert-scale questions, 10 for each subscale (rational ability, rational engagement, experiential ability, and experiential engagement). The survey questions are available in Appendix B.

There are several other instruments that are also available which can be used to measure the use of intuition. Allinson and Hayes (1996) developed another self-report inventory, the Cognitive Style Index (CSI) to locate people based on a uni-dimensional, bi-polar information processing concept. The validity of CSI has however been found to be deficient in recent studies (Hodgkinson & Sadler-Smith, 2003a, 2003b). There are also instruments like the Affect Infusion Model (AIM) by Forgas (Forgas, 1995) or the Agor Intuitive Management (AIM) survey by Weston Agor (Agor, 1989b) which have a different way of studying the use of intuition. The availability of these different instruments shows that there is still a mixture of instruments available which all produce a different view of intuition and its use in decision making. I selected CEST and REI because REI has been found to be one of the available instruments which sees information processing as based on two separate but parallel systems (which is in accordance with the definition of intuition in this study). REI is also widely used and readily available. REI has been criticized by Hodgkinson et al. (Hodgkinson et al., 2008; Hodgkinson et al., 2006) but I found it unnecessary to develop a totally new instrument for this study. REI will sufficiently support the purposes of my argumentation. For a detailed explanation of the CEST and REI refer to the Epstein, Pacini et al. (1996).

3.6 Questionnaire data analysis

The statistical tools used to analyze the construct validity and survey results were factor analysis and t-test. The 40 items from REI were entered into a principal component factor analysis with varimax rotation. A scree plot was used to verify the explanative power of the factor solution selected. The t-test was used to verify that there were no statistically significant differences between the interview group and control group. The results are elaborated upon in the results chapter of this thesis.

3.7 Validity, reliability and the researcher's role

3.7.1 Validity and reliability

I used triangulation by data type and by research method to improve the reliability and validity of this study and to support my findings in the interview data. In addition to the interview data a separate survey was conducted. For the survey instrument, one of the most well-known and widely used decision making self-evaluation tools - REI (Rational Experiential Inventory) was used (Epstein et al., 1996).

How to assess the validity and reliability in qualitative or quantitative survey research differs significantly. The appropriateness of the measurement instrument used in a quantitative survey can be assessed by first evaluating if the measurement instrument is measuring what was intended to measure (validity). Secondly by evaluating to what degree a re-measurement using the same instrument would lead to the same results (reliability)

(Bohrnstedt, 1983; Nummenmaa, 2004). The reliability and validity of the survey was evaluated by a factor analysis and by calculating Cronbach's coefficient alphas for the measurement items.

There is a long standing belief in psychological research that human judgments are less accurate than statistical ones (see e.g. Goldberg, 1970; Meehl, 1954, 1965). Human judgment and related biases have also been researched a great deal by e.g. Kahnemann and Tversky (Kahneman et al., 1982; Tversky & Kahneman, 1974). The qualitative research was evaluated using the criteria presented by Charmaz: Credibility, originality, resonance and usefulness (Charmaz, 2006). These criteria are in line with general evaluation criteria for the qualitative studies presented by Miles and Huberman (Miles & Huberman, 1994).

The qualitative research that I have done in this study is about making judgments based on the interview and survey data. The analysis and judgments made by a single researcher are always subjective and only questionably reliable. This can lead to false interpretations or at least a different analysis than another researcher would have produced. I have tried to avoid this by being transparent in how I ended up with the conclusions what I did. My methods are explained in the method and data analysis section and in the results section. This should help the reader to follow my thought process, and leaves my logic open to scrutiny.

I selected the first companies where the interviews were held based on their software engineering background. This selection limits the representativeness of the data. In order to improve this, two other interview companies were selected that were not purely software engineering companies, but were on the telecom engineering side. Even though this slightly improves the representativeness of the data, it is still limited to engineering type of companies. All the companies and their projects in the data were also limited to new product development.

The interviewed individuals were selected partially by myself, partially by the company contact person and partially based on recommendations from the first interviewed persons. I asked the first person I interviewed to recommend one or two other people who had thorough knowledge about innovations and innovation management in the company and who could discuss those topics. The purpose was to find people who had enough experience in innovation management. However, there is a risk that the recommended persons might share the same opinions and status as the person who recommended them. This is referred to as an *elite bias* or *representativeness heuristics* (Kahneman et al., 1982). I believe this to be a relatively rare case.

When the interviews started, the interviewees were assured anonymity. This was to get people to speak more freely about their experiences and about confidential issues. This anonymity is preserved in the study. Only the researcher can access the raw data and the outsider cannot connect the data to single employee of the company in question. This is accentuated since the interviewees were selected during the research process, not by the research setting. The companies from which the interviewed persons were selected are also treated anonymously. The anonymity of the companies is also preserved as much as possible. They can however quite easily be deduced based on the professional background of the researcher.

The interview structure and the questions remained basically the same during the all interviews. The interview questions were very open ended and allowed the discussion to explore all areas of innovation management. During the latter interviews an effort was made to steer the discussion towards innovation decision making as is typical with this kind of research method. Based on the first interviews, it became clear that decision making was the path to follow.

It has been noted that intuitive judgments are best studied if the research design minimizes the possibility of self-critical reflection and deliberation (Kahneman, 2003b). This research design with its openly structured interviews and no pre-arranged intuition focus follows the advice offered by Kahneman. The role of intuition only became apparent to the interviewed persons during the survey questionnaire phase. At that time all of the interviews had already been conducted.

It has to also be noted that the focus on intuitive judgment was not known to me when I started the research journey and selected the research approach and method. Only during later phases of the data analysis phase did I realize the appropriateness of the method.

3.7.2 Researcher's role

I started with a data first approach as described in chapter 3.2. Data collection started in the middle of my obligatory post graduate studies. That is to say that I already had some theoretical grounding at the start of the data collection but for the most part stayed as close as possible to the original idea of the grounded theory research. Since I had spent so many years in the industry, I did not have any existing academic practice or background theory available.

I started gathering the data openly by just trying to understand what happens in innovation management, what are the basic processes and how do the processes and actors operate. This revealed to me new avenues in innovation management even in a company where I had worked for several years. The early start had some disadvantages too, since I had to learn about data collection and analysis methods at the same time. This resulted in a re-coding and implementation of a new tool for the data analysis after the first two data collections periods. This may have had a negative effect to the validity of this study. However, I was aware of this potential problem in advance and I tried to avoid it during the analysis period.

The research has also some ethnographic elements because my work related to innovation management as a head of software production at the time of interviews. I was closely involved in product development. I had been involved in every decision concerning product development innovations at the company from late 2003 to late 2008. Later on I was also involved in innovation strategy level decisions. This of course has some effect on the way how I interpreted the interviews and how there were a possible source of bias in my analysis. I have tried to be as open minded as possible, but this is one of the limitations of this study.

On the other hand, I have a much deeper understanding of the innovation process in practice than does the normal researcher. It can also been seen as an advantage since I was familiar with the context and the phenomena to some extent from the beginning of the research. That provided me with the possibility of finding and interpreting the kinds of issues that may have otherwise been left unfound in the process. In this way the research data is more profound and richer.

4 Results

"It is not the strongest of the species that survive nor the most intelligent, but the one most responsive to change"

Charles Darwin

In this chapter the results of this research are presented. First the empirical data used for the research is presented. The selection of the case companies and the interviewees are described in the research method section of this dissertation. After the presentation of the empirical data, a detailed description of the grounded theory development of this research is presented in order to give the reader the possibility to evaluate the research process and development of the results. Next the quantitative research part of the dissertation is presented and finally a synthesis of the research findings is developed.

4.1 Company descriptions

4.1.1 Company Alfa

When the company Alfa was established, the business focus of Alfa was not in the software business, instead Alfa's main business was consulting with 3rd party software. When the technology shift happened from main-frame/mini computers to personal computers (PC) it greatly affected the whole software industry. The financial software sector (where the company operated) started to shift towards PC based financial software.

Results

One of the main points of Alfa's business was to concentrate on building packaged software and productizing it with the customer.

> "When we managed to get some revenue from the product licenses, we believed that it pays back to start the productization of the software in addition to pure customer tailored software development. It is more sensible to sell the same stuff many times than to do tailored project deliveries. There is more profit potential."

The basic differences and rules between consulting based tailoring and packaged software based licensing was a key item in the company strategy. This was a clear business model selection. As a small, owner financed company, all of the product development work had to be done as customer paid development, because no outside financing was in place to afford long term product development projects without cash flow into the company. This also meant that all of the products developed with the customer were not productized. Only the ones which seemed to have larger user potential were productized.

> ...we did a customer tailored application and when there was more demand we decided to productize it as an official product. The decision was made because of demand from several customers. We already had good relations with key customers from the 80s and many new customers also. We had good contacts in the financial management of customer companies.

What was the reason to concentrate on the software business instead of the consulting business? It could have been the educational background of the company management. All of the key people in the company had a degree in economics. At that time there was already some software business education in place in universities. However, since there was little experience or research on the software business, much of the decisions were based on feeling.

I think that the business decisions were done on the spot – not to analyze the background information too much.

Case by case decisions. The market demand was certainly the primary reason why to productize and also my own thoughts that this could be a good idea ... it was a feeling based decision.

The importance of complementary knowledge and common understanding was of vital importance in company Alfa. There was no-one in the team that was clearly more powerful who could always make the final call, rather the paths forward and decisions were made together. Certainly the most technically competent persons had a louder voice for technical decisions and also vice versa in other areas of the business.

The first clear change in the course happened when the market for one of Alfa's main products declined in Finland. There was suddenly a need to develop business into new areas. The first version of financial management software was built. Necessity makes you try hard and innovate since you need to survive.

> ...There was competition in the market...Basic reason could be the economic situation and the down turn of the markets...Consultation revenues dived. That was the compelling force.

The innovations that Alfa created (which are discussed in this study) happened mainly in late 1990s when the company had already moved into the software business.

4.1.2 Company Beta

Alfa and Beta have many similarities in their operations – they are approximately in same age, they have grown with same speed up to early 2000s and they were both software companies developing software into financial processes. However, they have some differences which are highlighted below.

Productization of the software was even more of a focus in Beta than in Alfa:

They (the customer) were given an offer and they paid for the work done. Applications were ready and fulfilled their requirements. We didn't do any tailoring, instead we had all the time clear that this has to be productized and suitable for other customers too. I had this in my agenda personally but it was in other people's minds too. Maintenance revenue was the key here.

Individual decision making was clearly emphasized in this company. Even though the discussion of the management team was important the final decision was made by the CEO. Management control was tight. This is seen in many comments:

> Together we discussed and started the development project, but [person] surely made the final decisions.

People involved in decision making – or discussions - changed from time to time. This was probably done on purpose by the CEO in order to get fresh ideas on the table. The decision process itself did not have a clear structure. Instead it seems that the decisions were made when the timing was right for making a certain decision. This almost resembles garbage can decision making (Cohen et al., 1972). However, it was effective way of making decisions in the company.

Technology played a key role in product development decisions. Also, a clear process structure and simple processes were on the focus. Products were developed starting with a small feature set and then enhanced gradually after the first part was stabilized.

> (the products) are based on that very simple basic idea...That we were planning it from the customer's point of view. What the simplest possible process to solve this problem is and how can it be implemented by the software. This is the thought process behind the majority of our development projects.

4.1.3 Company Gamma

Gamma operated in the telecom sector. The development of telecom equipment required a tight interconnection between hardware, integrated circuit, embedded software and management application software development. Projects were typically large and also long in duration.

The development was connected to inventions and company Gamma was also active in the patenting arena. However, most of the innovations came from customer ideas and Gamma further processed those ideas into innovations. The search for innovativeness was almost institutionalized in the company. New ideas were discovered based on competitor analysis, keeping up to date with market trends, from customer input and also by internal analysis.

> Then there are those cases when we have started to think about what does not work well. Things that could potentially have a lot of value for the customer but that do not work well at the moment. We have systematically started to analyze the situation.

They had noticed that innovation required the right personal attitude and support from the organization otherwise it was very difficult to nurture innovativeness in the company.

> Inventing and innovation is an attitude question. You need to have a supporting infrastructure for that in place...You need to have an inbuilt attitude towards innovativeness. It is dependent anyhow on individuals, the pride and desire to invent new things, so that in the middle of all that daily hassle one still tries to put forward these things.

> Innovation can have two directions. First, we have a problem and it needs to resolved, that resulted innovation. Secondly, when have a ready solution but we don't yet know the problem. These can both be good!

Results

Clearly an "engineer attitude" was driving innovation in the company "we have a problem, lets' solve it". Technology driven innovation was visible "we have a good technology available; let's find a problem that it solves". In the interviews the problem solving attitude was brought up by many of the interviewed persons.

> It is the problem itself...Every engineer knows that when you do things it goes from solving one problem to another. It is really problem solving – the real engineering work...If there is not this attitude in individual minds we are not going to survive...

Competition surveillance was also one area of inventions or ideas for the innovation. When you see what others have done to solve a problem, you might get a better idea of how to solve the same problem or a new idea to solve a similar problem elsewhere.

In totally new development and in a search for radical innovations the risks of unsuccessful projects are high. This company mitigated the risk by having clear back-up plans. This required the possibility of putting extra resources into R&D and innovation projects.

> In some cases it could be beneficial to try and fail and try again to get a good solution finally. You need to have that space for trial and error, so that if something really fails...Yes we have back-up plans. Sometimes in very comprehensive ones too [like competing R&D projects]...

4.1.4 Company Delta

Delta produced telecom products for mass consumer markets. In innovation and development co-creation with the community was highlighted. Delta had mobilized a widespread innovation community both internally and externally. This is how we mobilize the whole society and ecosystem where we operate to co-operate transparently with and for us.

The large size of the company and its R&D department provided it with huge muscles which could not be compared to any of the previously interviewed companies. Its large size also brought some challenges in the form of organizational and communication problems which build some overhead in the organizations efficiency. Alfa and Beta operated in a totally different world when compared with Delta and their innovation operations. However, all three companies had been successful in innovation year after year.

Decision making was not based on pure rational facts when the decision touched on unknown, uncertain cases.

> When we set-up a project we have to think about how the business model affects the case. Do we have a clear model, do we have an end user, what he does he want,...It is more or less based on gut feelings at that time.

Still clear criteria existed which limited the decisions made based on "gut feeling". Clear themes guided the whole operation.

It was the [theme]. It is based on the experience of how people lead others. That was what they realized during the years...Some clear theme needs to be established, that helps you manage the whole thing. Clear theme guides and manages almost by itself.

The importance of a common vision and a commonly understood theme was regularly highlighted by the interviewed persons. Delta was such a big company that it needed to had some common elements which tied the company together and built a feeling of togetherness, because otherwise organization and operations could have torn the company apart.

4.1.5 Summary: Innovative companies

All four companies can be regarded as innovative if you look at them from a R&D expenditure, product portfolio, industry requirements or simply a history track record perspective. Some of the companies are more on the technological innovation side, like Gamma, and some are more on the technology and user/consumer goods side, like Delta. All of the companies have example cases from incremental and radical innovations; some of the innovations can be described as at least partly to be disruptive. This is the case when totally new business models or application areas were formed. Most of the innovation activities in these case companies are related to new product development or process innovations although some of those can be categorized as also having paradigm level effects.

None of the companies are young start-up type companies – they all have reached mature age in their businesses. However, Alfa and Beta are small companies if compared to Gamma and Delta.

Innovation management practices in the companies vary. This is probably because of the size, industry area and age of the companies. The telecom industry is environmentally strictly regulated (standards) and this certainly has an effect on innovation management practices. Processes, including innovation management processes, need to ensure that the necessary standards are met and that the results can be clearly reported to the authorities. The operating environment of all of the companies can be described as a high-velocity environment. Technology changes and customer expectations are in constant change and the companies have to respond to those needs and changes in a rapid way.

The organizational structures of large companies are typically deeper than in smaller ones. This is also seen in these case companies. Heavy organizational structure added to a strict process culture can stiffen the organization and reduce innovation activities. This was not seen in this group of companies. It could be said that even the largest company has very vivid innovation practices, but how that is done is different than in the smaller companies. Gamma and Delta have established clear processes for taking into account all of the relevant parties as a means to foster innovativeness. In Alfa and Beta the close internal relations and proximity with the key customers naturally ensures innovativeness.

Notable is that the process cultures in Alfa and Beta are very different. Beta has developed and follows certified quality management procedures, whereas Alfa has a lightweight process approach. This does not affect on Betas performance or innovativeness in a negative way.

The roles of individuals in all companies seem to be equal. There are visionaries and workhorses in all companies. The specialization of work roles is taken further in Gamma and Delta. Alfa and Beta have more "all-round" persons, people with several roles and responsibilities. This is quite natural because of the size of the companies – specialization comes into place when the company is larger.

The decision making practices are related to the process and organization structures of the case companies. Despite the clear structures and responsibilities for example in Beta and Gamma, the decision making was sometimes done by by-passing the official path. A sign of lobbying and political decision making was noticed in many descriptions of the interviewed persons.

4.2 First encounter with the data: Decision makers' experiences of innovation management

The first analysis of the data was done by coding the data by activities and actions regarding innovation management. All 19 interviews were included in the data. The most grounded codes were connected to customer orientation (customer orientation, productizing with customer, commercialization importance, productization of the knowledge) the second group included codes like decision by feeling and group decision making.

After the first encounter with the data, the <u>core categories that</u> <u>emerged were: Innovation enablers, compelling forces and decision</u> <u>making</u>. Innovation enablers and compelling forces could further be associated with innovation. For innovation to happen you need to have

Results

innovation enablers and compelling forces. Both are needed in order to have a basis for innovation to succeed. When innovation is possible it has to be decided what to do. The decision process has several different properties which are described in the network figure in Appendix B and in the following chapters.

4.2.1 Innovation enablers

Innovation enablers are the basic prerequisites that are needed in order for innovation to be possible in the organization. Based on the analysis in this study enablers are not enough to make innovation successful, but without them innovation is impossible. Enablers are like the foundation of a house. Innovation enabler categories reported in the following chapters are: *Customer orientation, productization with the customer, a clear way of working, complementary knowledge, enabling technologies and keeping up to date with trends.*

Customer orientation

The most frequently mentioned grounded code recognized by the interviewees with regard to innovations was *customer orientation*. Customer orientation leads most of the development activities. A customer typically has a problem which needs a solution. That leads a customer oriented company to innovate new solutions. Clear customer orientation and productization with the customer also has had a direct relationship with the successful and unsuccessful innovations.

> We have been close to customers and listened to their issues and thought together, that has been the way to solve the cases. From and with the customer have we started... We delivered the solutions to the customer and learned how they operate. We found out where the biggest problems were, the development needs.

> ...was developed first without the customer. There was a strong growth orientation. We didn't handle the development process correctly, the customer was not close enough...Customer had the different view on the subject

than what we had. That's why the ... path has been so difficult.

Productizing knowledge is different than productizing with the customer. The former means that the company may have the knowledge but it is not necessarily tested with the customer before the productization is done. In this case – as noted earlier - there is a higher risk for unsuccessful innovation.

> We tried to recognize the situation and create a new application, productize and sell it. The commercial side dived. We did the product which did not have demand on the market.

There are also negative effects on customer orientation. Since commercialization is so tightly connected with development it has a tendency to reduce research and thus hurt the innovativeness of the company.

> Everything needs to have a commercial justification. That is the direction nowadays...how can you then innovate something new...

Productization with the customer

A company needs to be technology oriented by nature in order to be able to use or apply new technologies in innovation. This is a prerequisite and one of the innovation enablers. In order to productize with the customer you need to have something concrete to be productized. That something can be a technology or product idea which will bring value to the customer. Company's own core development makes it possible.

> we have a customer who pays it ... then it was just a question as to what we can do with it so that it can be productized to fit with other customers too...

> ... we start in the customer's world and think about what could be the best solution what we can offer it to serve the customer's needs...

Productization with the customer is probably the only way a company like Alfa or Beta can move forward. These companies must first sell one idea and one product to the first customer, get the financing from that deal and then if possible commercialize it and also deliver it to other companies. It is remarkable that the business model of the software business was all the time dominant in discussions with Alfa and Beta even though the business started with consulting and tailoring the software. The bigger the companies grew the more the emphasis shifts to the software business (i.e. license revenue).

Clear way of working

One of the innovation enablers is a *clear way of working*. A good simple process structure clarifies the basic procedures in the company and in innovation management. This provides the opportunity to concentrate some effort into innovating activities.

The process with which we in [company] did these things was well developed. We had a good structures e.g. for prioritization...We think that things are in surprisingly good shape in our operations. This helps discussions and synchronization. This can be reason why everything seems so self-evident

Clear way of working relates also to personal interfaces inside the company. Everyone can work together productively and everyone knows how to work together. Team commitment to innovations is highlighted in many interviews.

We were shaking because of the energy it gave to us

Clear way of working also relates to internal product structure so that you have clear ways of working with the products produced. This means for example interfaces and other product architectural things between different units of the products.

Complementary knowledge

There are innovation enablers such as *complementary knowledge*. They are the fuel to spark the new innovations. When there are different viewpoints on the discussion over the same problem novel solutions or proposals might arise. These could not have been raised if only one solution had been put forward. If the group is composed of different kinds of people with different kinds of professional backgrounds and a knowledge that fosters innovative thinking and innovations you can create an environment with good complementary knowledge.

> ...the core group what we first had we all had some kind of knowledge of financial management but also of individual strengths either in management, technologies, organization design or business management.

> We created a cross organizational taskforces where technology experts together with experts from the marketing and customer side worked together in workshops

Enabling technologies

The *enabling technologies* category consists of codes: technological innovations, flexible technologies, technology shifts what are happening, applying new technologies and compelling technologies. It is a category which is a combination of all technology related codes which influence whether innovation is possible.

> We saw that when Windows had a breakthrough in technology side it gave us new possibilities

> At that time Internet came fast and it could be said that it was the reason for the whole development project we launched

> We wanted that [technology] to be open, so that it could be sold to anyone

> ...different solutions which were not optimal for the customers at that time, but could have been solved much better with this new technology

There was a change in legislation which made the electronic archiving of invoices possible.

Usually it is understood that the technological advancement is the most important factor in new innovations. Based on what interviewees said this is not the case. This is addressed in the other categories presented in this study.

Keeping up to date with trends

Keeping up to date with trends includes activities like influencing in the standardization bodies, seminars, professional article browsing etc.

You get the inspiration sometimes just by following what happens in the standardization bodies, sometimes inspiration comes directly from customer needs, sometimes when competitor is revealing their solutions... It is based on your experience or the way you see the trends in the market.

You need to understand the information and be able to apply the knowledge to what you get from those sources in order to keep yourself up to date with trends and other developments in the industry.

4.2.2 Compelling forces

People and companies need compelling forces to do the necessary decisions and actions. Necessity is often the mother of invention and decision makers threatened with failure often discover ways to cut costs, produce better products and market them more effectively (March, 1994). If there is not yet enough pull available from the market then the necessary pressure needs to be created intentionally. The compelling forces categories which are reported in the following chapters are: *Changing business environment, competing in the market place and commercialization*.

Changing business environment

One of the most powerful forces is change in the business environment. If you are to survive in a changing business environment you need to invent and develop new businesses. This is even more compelling in companies which are relying only on their own cash flow without outside financing.

> When that business ended we started to do build the same functionality by ourselves with other tools...In the 90s the supply did not meet the demand in the market. We were figuring out what to do now.

> It was from the sales and marketing where we got the first indications that the market might be developing positively in that area.

The *changing business environment* category is closely connected with the *keeping up to date with trends* category which was described in the "innovation enablers" core category. The *changing business environment* is grouped under "compelling forces" because it is by nature more compelling than *keeping up to date with trends*, still they both have the same kinds of attributes.

Competing in the market place

Competition in the market place is the second powerful force motivating companies to innovate.

> The product was developed because of the competition in the market. We had to do it, otherwise the competitor could have said that we don't have the same functionality as they do.

Companies need to be aware what happens in the market place at all times. It is a continuous fight against time pressures and required functionalities. Entry into the market is usually very time critical. If you are late in launching new innovations, the market may be taken by the rival companies. We had a unit which was up to date all the time about what was happening in the market.

To be late means that we came out approximately at the same time as the big players in the market. The game is lost at that point.

Markets are not at the same level regarding the competitive situation. The ICT market is a high velocity market where the competition is tough and changing rapidly.

Commercialization

When a small company grows, commercialization always plays a key role. Everything that you do you have to be able to sell. There is no additional financing available for these companies. The importance of the commercialization is especially high in Alfa and Beta since all the operations have to be financed by incoming cash flow.

> We are not going to develop the product for the shelf but for the customer. Sales have the best contact with the customer... You have to keep in mind all the time the commercialization aspects.

Duplicating software relates to *commercialization*. It is one of the fundamentals in the packaged software business. You develop the core software once, package it and sell it to many customers as a readymade packaged solution.

In the organization there was an understanding that by productizing and commercializing good ideas we can grow fast...Duplicability was what we wanted to achieve.

Keeping it simple helps in developing packaged software. It is also a process issue. When you keep your processes simple and clear you work efficiently without additional overhead costs in your operations. That is again one of the fundamentals in successful business.

Eventually it originated from the thought that we want to build such an application that it could be delivered by one click.

If you are to get business from the software market, you need to get the commercialization done as easy and as fast as possible. The easier the software is to install, the better the revenue potential from the product, because of the increased predictability of the implementation projects and the higher the volume in the license revenue base.

4.2.3 Decision making elements

The third core category after "innovation enablers" and "compelling forces" is "decision making" related codes and categories (Figure 4.1). In innovation management, after the groundwork for innovation has been laid (innovation enablers and compelling forces) decisions need to be made. Categories under the "decision making" core category reported in the following chapters are: *Decision by feeling, decision by analysis and individual and group decision making. Decision by feeling and group decision making* were the most grounded codes under "decision making" core category. Results



Figure 4.1: Network of codes for decision making

Decision by feeling

Decision by feeling is reflected in many comments from all of the companies. Sometimes the decisions were made during unofficial occasions like coffee table discussions and sometimes the decision was made on the spot, very quickly.

> This decision was made because it just felt like that...There is a good process in use which produces things what you don't even recognize.

> ...we were a small company and the decision making was very simple. We didn't need any analysis because it looked so clear and straight forward.

> It just came up and we decided to start to implement it like that...In one way or another we have always had that insight that the feeling has to be strong that this is needed.

In order to be able to make the decisions by feeling you need to have experience in the market, technologies and especially from the customers. Decision by feeling is like how a nurse cares for a patient. You just know the subject and area so well that you know by heart what to do. There are many guiding ideas which help you to go forward and you don't even think about those. They just seem so obvious. When you are so tightly connected to that it is almost same as your values.

Decision by analysis

Decision by analysis is related to a *clear way of working*. If you have systematic ways of doing things (processes in place), you have the possibility of making decisions by analysis and you are making the decision constantly using the same framework. Market and competition analysis is just one example of the tools used in decision by analysis approach. In company Beta the emphasis to clear and strict process culture was especially evident in the interviewees' descriptions about innovation practices in the company.

> It is then again the decision making process. We have studies and gates, it is the accepted product creation process. It goes on quite analytically...I don't know if it is that special, but we try to listen to the sales and customers and then in co-operation make the decisions.

> You can go on with facts when you have that kind of environment where you know the basic rules ... it is based on the criteria that we have.

Individual and group decision making

It is noted that decision processes where steered by one individual in one of the companies. This is because of personal properties but also because of the size of the company. It is natural that in a smaller company the decision making was very simple and follows the individual decision process. When the company grows in size, decision making becomes more complex and multiple actors are involved in the process.

There wasn't too many others whom to discuss with. I made the decisions.

He was very often the final decision maker.

Group decision making was highlighted in rest of the companies. It was visible in un-official discussions as well as in official decision making bodies. The bigger the companies are the more important the cross functional planning and decision making becomes.

> ...we decided that we will start to develop application to this. We had kind of subconscious vision...It was a common mind.

> It was these development meetings where we had participants from sales, r&d, implementation...the list of issues that we prioritized and selected the best for further planning. Then the management group decided what to develop.

> ...more people always to decide those things which were somehow binding. But the discussion was really very democratic.

4.2.4 Summary after the first encounter with the data: Innovation decision making

The emerged core categories after first encounter with the data analysis were "innovation enablers", "compelling forces" and "decision making". "Innovation enablers" came up as a natural category because the interviews were about innovation management. The discussions easily moved into paths describing what made the innovations possible in these companies. All of the companies built an offering into the marketplace, so it is natural that *customer orientation* and *working with the customers* was raised as a key code in the data analysis.

"Compelling forces" are needed in addition to enablers to push the initiative further. Usually the needed push came as a result of the competition, changes in the market or some internal necessity to act. I felt that the interesting topics concerning "innovation enablers" or "compelling force"s had been fully explored. Instead "decision making" – how decisions are really made – had more interesting topics to explore further.
The core category "decision making" was analyzed further in more detail. The "decision by feeling" code/category and the sub codes like "spontaneous decision making" were noted in the majority of the decision making descriptions. That guided how the analysis looked at what is happening in those situations.

Decision by feeling was highlighted in the vast majority of decision making occurrences. Eighty six (86) of the quotes were coded as decision by feeling whereas only twenty two (22) quotes were coded as decision by analysis. Decisions usually happened by accident or spontaneously without deliberate reasoning. This indicates that there are a large portion of non-analytical decision makers in these studied cases.

Another clear observation was that most of the quotes were related to group decision making and the minority to individual decision making (59/29). This further indicates that decisions concerning innovations are more often done in group decision making situations than by individual decision makers. It does not explain how an individual on a team ended up making a decision which was then brought up in the group decision making situation.

After reading and analyzing again the quotes related to the "decision by feeling" code/category the use of non-rational decision making elements raised my interest, because of the high number of occurrences of the codes. One of the non-rational decision making elements is the use of intuition in decision making (see chapter 2.1.3). That was the target for the second encounter with the full data. A matrix of the most appeared codes is available in Appendix C.

4.3 Second encounter with the data: Facets on using intuition at the front end of innovation

The second encounter with the data was done by reading and analyzing again the transcribed data and coding it from a different perspective. This time the coding was done by looking for occurrences describing or linked to the use of non-rational – intuition related - decision making elements purely, rather than for activities or actions as was done in the first encounter. By selecting the new angle for the analysis of data I

Results

aimed at getting deeper into the selected core category and the phenomena and looking at the data from a fresh perspective. This is very important since there is always a danger of getting stuck on the first perspective especially if it relates directly to your profession.

The first categorization during the second encounter with the data resulted to four core categories: "Symptoms", "preceding enablers", "simultaneous enablers" and "safeguards". The interviewees expressed that the decisions must "just get done" - spontaneously and based on the feelings. These expressions where categorized under the core category "symptoms of the use of intuition". The use of intuition in the decision making process requires that basic preceding enablers supporting intuition are in place. Decision makers explained that there has to be some order in the decision making process and that you have to value intuition as a decision making tool. This resulted in the category "preceding enablers". Many interviewees were found to be very enthusiastic about their profession and their duties. Their approach was like an explorer finding new ways to navigate across the seas. These qualities include lowering the barrier that prevents you from using your intuitive abilities - the category was named "simultaneous enablers". Some of the interviewed persons described that they were able to immediately test their decisions in the form of a pilot or other direct feedback. Expressions related to testing the intuition were categorized under "safeguards" for the use of an intuition category.

These four emerged core categories are explained in more detail in the following sections. The coding of the quotes (CxPy: Company <u>x</u> Person <u>y</u>), as seen in later chapters, is used to preserve the anonymity of the interviewed persons, but also giving the reader the ability to follow the researchers thought process and to evaluate the process.

4.3.1 Symptoms of the use of intuition

Symptoms are the expressions and codes which indicate that intuition could be affecting the decision making event in the case. The interviewed persons on many occasions described their decision making as being made partly based on feeling. The categories reported are: *Decision by* feeling, spontaneous decision making, unconscious vision/seeing, individual thought process, decision by accident and non-decisiveness.

Decision by feeling

Decision by feeling category as described in section 4.2.3 belongs to this category. Decision by feeling surfaces in cases where there may not be any special reasoning behind the decision that was made. The interviewed person had a feeling at the time of decision and the decision got made. They are also not saying that intuition was used to guide the decision; instead some other expressions are used. Typically a person could not articulate what lead to the decision or why it was made. Decision by feeling does not mean that the person had a good feeling at the time of decision but that the decision was made simply by relying on what felt right. Interviewees also described that in some cases the feeling did not appear immediately; instead it took some time to mature.

> I have an opinion that business decisions have just been made so that there is not too much thought process – there is just the situation at hand and we had to figure out what to do. We didn't go very far to analyze the backgrounds etc. the decision points just came. Decision was certainly one driver as to why we started to develop and certainly also our own idea.[C4P6]

Spontaneous decision making

Spontaneous decision making describes the event were a decision gets made rapidly without deliberate, at least not recognizable, thought process. It did not require long discussions or descriptions. Unofficial undocumented discussions or meetings could have preceded the event. Interviewees described this way of acting as a natural way of working in the company, a simple method for making decisions.

> The decision was made by me and [person] – I think. Probably the final say came from [person]. It [decision] just appeared and we decided to start doing it like that. It

was the decision what we saw as necessary at that time even though we had no clue at all as to whom we would sell it to later on [C3P2].

Spontaneous differs from *decision by feeling* in the speed of the decision making. There might have been some unconscious thought process or maturing phase in the background but in spontaneous decision making these forces are not recognizable at all. A sudden spark or a light bulb going on describes a spontaneous decision making incident.

Unconscious vision

Unconscious vision relates to a common understanding code. Unconscious vision was reflected by many of the interviewees in one of the companies. Seeing was another description of the same phenomena.

> When we were dealing with [subject and subject], we had the main elements in place even though that was new way of doing those things. Suddenly we saw that there could be a seed for a [product]. No one else had it at that time. We knew that it is worth trying. [C4P2]

Seeing or unconscious vision highlights the appearance or existence of some vision or view. The decision making incident itself could have been fast or slow but some view or vision was found to be present at the time. Vision was described by interviewees having group characteristic – not a single person vision but a common group vision.

> We just had a common vision of the case [name]. Because of that we ended up doing what we did. [C4P1]

Individual thought process

Individual thought processes related events were described in such a way that the interviewee had a thought which guided the decision process. Thought is not clear or firm as a vision, but more of an unstructured vague idea. Often it was also described as individual thought, not as a common view or vision. This category is similar to the *decision by*

feeling category. However, this is not as comprehensive as it was with *decision by feeling*.

I just had that thought...that fitted to my thought process...it comes as a leading thought [C3P1].

Decision by accident

When the decisions happen by accident and the feeling is that the process is just drifting forward was described by many when was asked to describe how the process actually went forward. The interviewee described no clear structure for how the decisions were made, instead that was a gradual process of maturing and formalization.

> Well – at first it was just an idea. An internal idea. In a way we could not concretize it at that time as well as we probably should have. It was just an idea that we should somehow be able to do [solution description]... Then at some stage it just started to concretize into that form that we started to sketch the solution on the paper...the decision matured their time [C3P4].

When the decisions happened by accident the interviewees realized afterwards that the decision was really made at that time. During the incident it was not recognizable. Accidental decisions became visible during the later phases of the process and at this point it became apparent that the actual decision had been already made.

Indecisiveness

One kind of indecisiveness can be understood as a sign of intuition. Indecisiveness is closely related to decision by accident but indecisiveness has a clear active component of indecision.

> In my opinion there was not really much decision making in those cases...[C4P1]

Results

It is deliberately noticed that there was no willingness to make the decision however it was later realized that the decision had already been made earlier. *Indecisiveness* is also related to *drifting /decision by accident* in a kind of flow which just takes things forward. *Indecisiveness* just has the active – indecisive - part with it.

4.3.2 Preceding enablers

When the interviewees described their actions they highlighted the need for some basic structures and shared values – regardless of the use of intuition in the final decision. These elements are categorized as "preceding enablers". Preceding enablers are needed as a basis for the intuition to be used during the decision process. The categories in the preceding enablers are: *Values, clear way of working/processes, knowledge and discussion*. Without these preceding enablers it is not possible for intuition to surface – there is no support for it. Preceding enablers are elements of people's behavior, the existing conditions or processes.

Values

One of the necessary prerequisites – preceding enablers - for the use of intuition is that the person or the group values trust or support towards the use of intuition. If they do not believe that intuition is a valid construct in the decision making process then intuition is not trusted and is not used. When the importance of the values is discussed, the interviewee stressed the importance of common goals.

It is from experience, it is my values. It is a combination of all these facts, values and feelings. It is something so deeply grounded that you cannot change it... We had such a strong belief on it...everything is just based on it that you and the team believes in it...very strong commitment to it. [C1P1]

Values and trust take long time to develop and institutionalize in an organization. It is the same with intuition. New organizations or groups are less capable of using intuition in their decision making than older –

more mature – ones. Typically the incidents where values and trust were mentioned came from the groups who had worked together for years.

Clear way of working/processes

Intuition needs a solid grounding as a basis. If everything is a mess there is no support for the use of intuition. In chaotic environments decisions get made using irrational basis (Aaltonen, 2007). Several interviewees from the same company described the importance of clear structures and processes as a basis for their innovation work.

> First the basics has to be in order...very strong process culture...We had a good templates for prioritization of activities...at least some systematic in those activities...[C3P1]

Organizational routines and processes build the fertile soil for intuition to be used. If there is no order at all and the decision processes are in chaos then there is nothing that the use of intuition can be attached to or linked to.

Knowledge

Knowledge works in the background of the decision making process where intuition is used. Without it, intuition cannot function properly, however this is not simple heuristic. Heuristics and intuition are closely related, but heuristics can typically be applied to predefined patterns of actions whereas intuitions cannot. The interviewees often described the existence of knowledge as the *importance of experience*. *Complementary knowledge* relates to the same code in that if you have complementary knowledge within a team or group of people you do not personally have to have all the knowledge in your possession.

> It was the knowledge which made the spark...it would not have worked if there had not been so many experienced persons...[C1P2]

> We had the knowledge inside the organization...there was experience, we had been doing before the same kind

of exercises...we brought the cross-functional experience to the process...[C1P3]

Knowledge or experience of course requires time to develop. That is why it is only available in mature groups or organizations with more acquired knowledge. The interviewed persons all had a substantial background as decision makers in innovations and thus the experience they had gathered was comprehensive.

Discussion

A rich discussion culture is needed to get people to share ideas and knowledge and use intuition more easily. In informal discussions, many ideas were put forward without the need for more formal meetings or decisions. Even though some interviewed persons mentioned the role of individual decision makers they all recognized the importance of discussions - official and unofficial.

> We had many common discussions – and the decisions just happened...the group actively discusses...it was a kind of brainstorming what we had...the group just discussed about it....[C4P3]

> It was kind of brainstorming all the time. People just threw the ideas. Then we tuned it further. How did we then select the best one? Well - let's say that those were the best ideas which somehow wake you up. [C2P2]

Discussion culture nurtures the change of ideas and knowledge which in turn builds appropriate grounds for the intuition to be used. A discussion culture also builds trust and anchors values into the organization, which were some of the already mentioned preceding enablers.

4.3.3 Simultaneous enablers

The interviewees who clearly used intuition (because the indications were highly visible) in their decision making had additional qualities that made it easier for them to use intuition. These qualities are more personal in nature than the preceding enablers. They also worked more closely on the decision making incident itself. I named this core category "simultaneous enablers". Categories of simultaneous enablers are: *Courage and enthusiasm, imaging and discovering, and final push and creativity.*

"Simultaneous enablers" are the qualities which are required even though the "preceding enablers" are in place. If the threshold for the use of intuition is too high intuition is not used even though the preceding enablers are there. Simultaneous enablers work in favor of preceding enablers and lower the threshold for the use of intuition in decision making. It is difficult to distinguish between the" preceding enablers" and "simultaneous enablers". I judge the preceding enablers to be more important for the use of intuition than the simultaneous enablers. This is because without the preceding enablers the use of intuition lacks a basis to work from. Without the simultaneous enablers it is just unlikely that intuition will be used.

Courage and enthusiasm

You need to have enough courage to trust your intuition. In some cases people are afraid to reveal that the basis of their decision was intuition. Instead they try to formulate the decision based on logical reasoning. As self-confidence develops, trust in intuition also grows.

> It was the common way of working – not that much more analyzing. We had knowledge of those things but the tools were at that time deficient...courage was the trump there. We made a lot of compromises afterwards. Meetings were everybody involved was participating. [C4P3]

Enthusiasm for the case at hand lowers your intuition threshold. Strong enthusiasm is seen in cases where there are very enthusiastic – vigorous – persons who put all their heart and soul into the project.

> Typically they are exceptional enthusiastic...you have the pride and willingness to innovate new things. With all the other rush duties you try to put forward new innova

tions...There has to be someone who personally drives the case forward. If there is enough value to you personally you will put it forward. I believe it is much connected to your strong will, and you nurture it with discussions and open culture.[C2P1]

When you are very committed to innovation and putting all your efforts and energy to it, also develops your self-confidence and that way lowers the threshold for the use of intuition.

Imaging and discovering

Imaging is a way of looking forward in the innovation process. When you are able to imagine possible outcomes and results, the decision process is easier and lowers your barriers for the use of intuition. The clear vision of the future also guided some of the interviewee's decision making. The interviewed persons figured out how to proceed by thinking about possible future paths.

> We started to think about productizing. We had knowledge about to make good use of. Knowledge was widespread, not only in [area] but also in other disciplines. We did some small market studies and also customers gave us ideas. We thought about technology developments and had brainstorming sessions about the future.[C4P5]

> The first discussion was around Spring...We started to figure out what could be the solution for the future. At that time internet was hype and that was the basis for the imaging...all the possible solutions were considered. [C3P3]

Vivid imagination combined with the aforementioned "preceding enablers" is very strongly correlated with the lowering of intuition. When coupled with *enthusiasm* it creates a positive flow to the whole process of innovation.

The nature of the explorer or discoverer - "to go beyond the last frontier" – is the quality which is found in many interview instances. When you are discovering something new there is no previous knowledge or information available and you rely on intuition. The stories told by the interviewed persons provides a feeling of a exploring the nature of innovations.

> We were discovering it and then we thought it over and understood that what these changes really were. In that way it started to take form. I think I wrote the first specifications and then we started to implement it. [C3P4]

> It is the problem itself. Every engineer knows that when you innovate you go from one problem to another. That is the real engineering work – the problem solving work. Sometimes it then grows to something really new and big – and then we have innovations. When there is something that is not in line with the existing knowledge then it that is worth of exploring further. Those could be very valuable. Those are no self-evident nor fast things to discover – it takes time to mature. [C2P1]

Discovering is also at the very core of innovativeness, which is what innovation really is; the discovery of new areas and cultivating the findings into something new and valuable.

Final push and creativity

Some of the innovation enablers work closely with the process where intuition is used and some are more distant. Sometimes a final push is needed to go forward and use the intuition in the process. Creativity and sensitivity to intuition are closely related to the intuitive process.

> Together we found out that...then we started to understand...we sought and thought together...ideas were thrown in the air...

> the persons who can seek and then take those up to discussion...when we then figured out that this is it what it is all about...[C2P2]

It is the ultimate trust of your own ideas and feelings that gives the final push – called a "child-like belief" by one interviewee.

> It was crystal clear for us that if it had worked once it will work again in other circumstances too – it was just plain child belief. None of us had experience and knowledge to say that it could work, but we just stated let's do it. [C4P2]

4.3.4 Safeguards

All of the interviewed persons brought up in one form or another the importance of testing their decisions before going too far along the selected path. Those interviewed persons who were more connected to technical product development mentioned the use of prototypes in early phases of the innovation process. Those individuals who were closer to the customer interface highlighted the immediate customer feedback in a form of a pilot customer or clear customer orientation in the development process. I marked this core category of codes as "safeguards". Safeguards included the categories: *Testing, prototyping and inside sponsor*.

Testing and prototyping

Safeguards make the use of intuition easier since it provides a fast way of probing and testing the intuitive decision leading to the correct direction. When there are safeguards available, the feedback is fast and accurate and people are more willing to trust their initial intuition.

> I saw it so that there was no time to think too much. We started from the scratch...lots of different things to take into the design. Very fast decisions had to be made. Of course we then made prototypes and commented those, with close colleagues and customers...[C2P4]

> We innovated and combined things in innovative way. Why we were so successful was that we built up an extremely short path from the customer. The path from cus-

tomer to development. We were able to respond market needs with extreme flexibility. [C1P3]

Testing and prototyping (sometimes called piloting) are the common tools in the innovative processes. They are typically used to get immediate feedback from customers or users. In that way they also act as safeguards for the use of intuition. Confidence in intuition is stronger if you know that whatever happens you are not going to be off track because the earlier the feedback – the smaller the mistake, the easier it is to recover from.

Inside Sponsor

Some of the interviewees told that a sponsor inside the company eases the innovative process. A sponsor can be also categorized as a "safeguard". When there is a person sponsoring the project or the idea that smooth's the way and removes the need for extensive justification for the decisions taken. This is important also because when the decision is made with an intuitive basis, there is no bullet proof justification available.

Safeguards work in favor of "simultaneous enablers", since they lower the threshold for a person to use the intuition. It is the timing difference in the process which makes them their own group or domain. Safeguards are there to be used when a decision is made and a path is selected. "Simultaneous enablers" work before the decision to help you to use intuition in the decision making process.

4.3.5 Summary after the second encounter with the data: Four facets of intuition frame

Use of intuition in innovation management shows up from different facets as was described in previous chapters. This can be considered as a new knowledge in the intuition and innovation management research. The interviewed persons all pointed out different aspects that made the building of the frame possible and how intuition reveals itself in innovation front end decision making. Some of the interviewed persons talked directly about the use of intuition but others did not mention the word

Results

"intuition" itself. However, in all of the interviews there were clues that helped to categorize and formalize the frame of intuition use in decision making.

Four core categories (called as facets) were raised based on the codes from transcribed interviews as described in the previous chapters ("symptoms", "preceding enablers", "simultaneous enablers" and "safeguards"). Symptoms are the signs which signal that intuition may have been in place during the process. The use of intuition is not necessarily visible in the first place. That is why you have to create markers to make the use of intuition visible. Symptoms are to mark whether intuition could have been used in the decision making process. By tracking symptoms the use of intuition is operationalized in the process.

There are preceding enablers that are required for intuitive decision making. Preceding enablers are organizational-level structures or processes which are needed in order to have an "intuitive support" set-up in place. Without preceding enablers the use of intuition is not impossible, but clearly less likely. Simultaneous enablers are closer to the use of intuition in the decision making process. They are more environmentally connected. Preceding enablers and simultaneous enablers both work in favor of the use of intuition before it is used.

Safeguards are there to test the results of an intuitive decision afterwards and provide immediate feedback to the process. Safeguards are also, as preceding enablers, organizational level items. Safeguards work as a single element used to embrace learning and reflect on decision making.

Intuition as a phenomenon is a complex issue. It shows up in very different ways and the interpretation of the use of intuition is always subject to criticism. The following Figure 4.2 paints the frame how the intuition was portrayed in the interviews and how it is used in the empirical data of this study. The connecting lines roughly show the relationships between the different domains inside the phenomenon. The figure is not to be interpreted as a cause-effect diagram with arrows showing causal relationships or event-state diagrams as such, rather it is an illustrative description of the codes and categories from the interview data with the connections between them. It is a combination of a cognitive map and an event-state diagram (see e.g. Miles & Huberman, 1994).

What the interviewees did not discuss during the interviews was the cases of false intuitive decisions. This is a clear deficiency. The people and cases studied were originally targeted to also include material from unsuccessful decisions. However, during the interviews the interviewees talked briefly about the unsuccessful cases and the interview data material remained limited. That prevented a comparative analysis between the successful and the unsuccessful cases and decisions.



Figure 4.2: Illustrative figure of the use of intuition - "Four facets of intuition frame".

When the frame for the use of intuition in innovation front end was built, the next question was what role did the decision makers play in that process – what was their approach in using intuition. That question guided the third encounter with the data which is reported in the following section.

4.4 Third encounter with the data: Alternative approaches to decision makers' use of intuition

The third encounter and categorization was done based on the approaches and roles of the persons using intuition. Intuition emerged as a guiding core category during the first encounter and was revealed more in the second encounter with the data. During the third encounter I decided to take a more detailed look at how the use of intuition appears in the interviews of experienced decision makers. The different descriptions the interviewed persons provided about the decision making occasions made me curious about how intuition is visible in different roles. Even though the decisions were not solely done by the interviewed persons (group decision making), the persons who explained how they made the decisions or ended up to their conclusions.

The written role descriptions do not have any common structure since the highlighted characteristics varied between the different role descriptions. The descriptions have been written to describe what approaches the interviewed persons took in relation to their use of intuition in decision making. Some direct quotes have been taken in with the text to better explain the characteristics and my reasoning behind the design of the role descriptions.

The descriptions are not exact copies of the persons interviewed. Instead they try to emphasize the dominating qualities of a role when that person uses intuition in decision making. They are more like narratives about experienced decision makers and their use of intuition. There is always an interpretative element in writing these narrative role descriptions – as is always the case in qualitative research. Some individuals have also been placed in two roles, because they have qualities relevant to both.

The names used in the role descriptions are fictitious and not related to actual persons interviewed. All the fictitious names are male, even though there were women in the group. This is to secure the anonymity of those interviewed.

The basis for the role descriptions was done by selecting the most frequently appearing intuitive related codes into a matrix together with the interviewed persons (persons are coded as C4P1, C4P2,... to preserve the anonymity). Eleven of the most frequently appearing codes and the nine-teen interviewed persons are shown in a matrix in Table 4.1. The role descriptions were done based on the Table 4.1 and by taking into account the original transcriptions which were again analyzed and recoded.

The emerged roles were: <u>Drifter, thinker, negotiator, tester, discoverer,</u> <u>believer and seer</u>. I categorized the roles based on the number of occurrences of the codes appearing in the interview as well as my thought process for the most descriptive naming of that category (role). The order of the role descriptions as presented here follows no priority or other logic. I wrote them based on the order of appearance during the analysis process. These roles or categories are explained in more detail in the following sections.

	Values	Imaging	Discussing	Testing	Enthusiasm	Drifting	Knowledge	Visioning	Believing	Seeing	Discovering
C4P1										х	х
C4P2							х				
C4P3						х					
C4P4											
C4P5			х								
C4P6		х									
C3P1					х						х
C3P2										х	
C3P3		х									
C3P4						х		х			х
C3P5											
C2P1					х						х
C2P2		х					х			х	
C2P3											
C2P4			х						х		
C2P5											
C1P1	х			х					х		
C1P2											
C1P3				х							

Table 4.1: Most appeared code	s per interviewed	person (CxP	y)
-------------------------------	-------------------	-------------	----

4.4.1 Drifter

Mike is a person who goes with the flow – he is a drifter. He is silent in group decision making and does not have much of an opinion over the decision process as a whole - "the decisions just happened". However, Mike feels that the decisions were correct and he would have ended up with the same decisions by himself. Two drifters were identified in the interviews.

The use of intuition in Mike's case is supportive. He relies on the intuition as he drifts with the process. Probably a drifter does not even recognize the use of intuition.

> That way it started. What is the innovativeness here is that there were many of those things which just fitted together. We floated away with the process flow. [C4P3]

The interviews revealed two types of drifters – active and passive drifters. A passive drifter is part of the team and goes with it. He emphasizes group decision making importance and common understanding – he is a consensus seeking person. As in most of the cases customer orientation and productization of the knowledge is highlighted in his way of decision making. Decisions just happen spontaneously.

Mike has a lot of experience in what he is doing. He has stayed for years in a similar kind of position and a similar role. Mike really knows the substance. As a passive drifter Mike does not raise his opinions for discussion, he rather keeps quiet even though he has solid experience and strong subject knowledge.

> It was the commonly accepted way of doing things, there was not that much thought behind it. We had knowledge and that kind of application was not available. I Think [person] had the lead in it. I felt I was not that excited about it.[C4P3]

So why is Mike a drifter? One reason is that he is so used to the way the company operates that he does not see any point in making changes. In this way, Mike is a profoundly committed company citizen. It could also be that Mike has had enough and he does not really care how things go. This is however not the case with the two identified drifters.

Sam has more active role in drifting – he is an active drifter. Sam is part of the team but emphasizes the importance of individual decision making. His own thoughts play an important role when he discovers the path forward. Active drifting is highlighted by his way of proceeding – learning by doing.

> At some stage of the process just started to concretize...We started to seriously put something down on the paper as a form of a specification. I wrote the first specifications and then we circulated them within the team.[C3P4]

Sam has an active role in the process but he still drifts with the flow – he does not take active role in guiding or steering the process forward. An active drifter is much more talkative than the passive drifter.

Sam takes the lead in bringing issues to the table for discussion. He collects all of the ideas, actively keeps up relationships with colleagues and gets his voice heard in meetings. But that is where he stops. He does not guide the process forward, instead he relies on the process itself to move things forward. In the decision making he notices more clearly the presence of intuition than does the passive drifter.

We just get the feel of doing [product] in that way... it was more like a gut feeling – with the discussion.[C3P4]

Why is Sam an active drifter? Mike and Sam share some qualities long work history, solid competence. They are drifters because they feel comfortable as such. They do not want to take the lead and step up to a new role. Their use of intuition is based on their high degree of competence resulting from their education and experience. They are not willing to step out from their comfort zone and take an active part in guiding the process forward. Sam is an active drifter probably because he is more extrovert than Mike.

4.4.2 Thinker

Thinkers are knowledge based and on-the-job learners. Knowledge based individuals gather knowledge and emphasize the importance of experience in the decision making process. Three thinkers were identified amongst the interviewed persons.

Keith has a lot of experience he uses in his decision making to guide his thought process – he is a thinker. Keith seeks complementary knowledge to support his knowledge and is eager to productize knowledge with the customer. He gets complementary knowledge by engaging himself actively in co-operation with internal and external parties.

> You get the inspiration from standardization forums, from customer requirements, form competitor offering etc. Then you think it forward up to a conclusion and find out that – wait a minute – is this really now processed enough to get the best solution. [C2P5]

Keith has also experienced failures and has learned lessons from them. He has his "eyes open" all the time and he actively promotes new ways of seeing things. Keith has lots of ideas and uses intuition as he goes forward with the customer.

> All the time we discussed and weighed the different options... we were professionals in that area and at the same time very close to the customer. We have been able to sketch fast what is needed. Everybody tells a bit differently what is needed, but still they want the same thing. We massaged the ideas, together with the customer in the workshop meetings. Gut feeling takes us forward. [C3P3]

What makes Keith a thinker is that he highlights the importance of ideas, thoughts and the thought process connected to cultivate those –

imagination. A high number of discussion oriented meetings with common thinking – brainstorming – guides his actions. The group where Keith operates shares the same vision – even though there are sometimes very conflicting opinions about how to realize that vision. All the people are experts and have strong opinions about their own area but at the same time recognize and respect the other experts' views.

> There were many persons with very strong will in the meetings...the competence areas however were not overlapping – that was the salvation of the whole process. [C2P2]

Keith proceeds cautiously and likes to have some back-up plans to cover his back. He uses his imagination – intuition – more often when he has other routes already figured out. He uses assumptions and tests those assumptions gradually.

> Experience is the best teacher here. In some cases it is even good to explore a bit and then fail, then you might end up with the very good end result finally. Some room for maneuvering needs to be there. Of course it helps if you have some back-up plans - if this goes wrong the whole game is still not over. [C2P2]

Keith proceeds step-by-step, not taking too much of a leap at one time which also gives him confidence about being on the right track. Each step produces some tangible, understandable results. He tends to go towards simple solutions. Keith is able to quickly sketch out his initial ideas based on the customer need or on internal discussions and feedback. Experience with failure keeps Keith on track, that experience prevents catastrophic failures. He is "on his toes" all the time.

Keith and Mike (and Sam) are alike in a way they use and value experience and knowledge. Keith's imaginative nature guides him in a more creative direction than Mike. Keith realizes the danger in being too creative. That is why he uses back-up plans and safeguards to cover his back.

4.4.3 Negotiator

George is open and he always has vivid discussions with the team – he is a negotiator. He relies on good relationships within the team. Two conversational type negotiators were identified in the interviewee group.

The team where George works has worked together for a long time. Even though George is open in nature, it takes some time before George trusts a newcomer. During discussions, George raises past experiences, both successful and unsuccessful, in order to introduce a broader perspective.

George has an analytical approach to decision making when he is analyzing the data and proposes the decision to the team. The analytical stance that he has favors in having an official market, technology and competition analyses. George emphasizes the importance of direct feedback which can be obtained from prototyping the solutions with the customer.

> The functionality of the product must be verified already at the very beginning of the development process with the customer. [C4P5]

The conversational nature that George has is more controlled than with an imaginative person such as Keith. George does not let "thoughts fly" as Keith does in brainstorming sessions. This might be somewhat limiting to innovation but it brings more structure to the process itself.

Does George use intuition in decision making? Maybe, but that is not seen in the data of this study directly. Sometimes people camouflage the use of intuition in the form of well-articulated logical reasoning- either purposely or not – since having a rational explanation for the decision helps in justifying that decision to others. Rationality is appreciated in group decision making and when documenting the decisions.

George does not limit the intuition of the other people in the group, but he remains on the rational side of the decision making. This division of roles or responsibilities into intuitive and rational can be positive to the whole team since George is the person who ensures that the targets are met, the process is documented and that heads are not too high in the clouds. Complementary knowledge of the team members is highlighted in the same way as in Keith's group.

> We had these kinds of conversations pretty often. Sometimes those were very tempered but with the good friends you can say things directly without too much conflict others being offended. [C2P4]

The importance of good and confidential customer relations is important for George. As in internal groups, he relies on having long term relations and personal relationships with the key stakeholders. "Lobbying" the stakeholders is one of the methods that George uses.

There were those contacts...we had a good name and we were appreciated within customers. [C4P5]

Having confidential relationships with the customer gives you the ability to involve innovative solutions in the discussion. Those are not directly rejected because the earlier projects have built the trust between the parties. George appears to be a rational person. He is a technically oriented person. The only intuitive parts of his decision making are in limiting how he involves radical new ideas in the discussion – without thoroughly analyzing them.

4.4.4 Tester

James makes fast decisions and checks the results immediately by using a test – he is a tester. James proceeds cautiously in one way, but at the same time he is very fast. He takes a stepwise approach, making fast decisions with the ability to test the intuition or innovation immediately. Three testers were identified in the interviewee group.

James has a clear vision of the future and he has clearly outlined his path. He has a clear process which he follows and he regularly compares results against the targets and his vision. The importance of commonly shared and understood values and vision is high. James and everybody

Results

working with him understand this and apply it to every decision making event. The process that James follows is simple and clear with defined key criteria of how to test the decisions and results in practice.

> We have a vision and strategy and the targets what we want to achieve... then you have a process where you screen the ideas – harvest them forward... we test them continuously to find out how they fulfill our targets. [C1P1]

Is James using intuition in the decision making process? The process and how it is followed is a very analytical approach to decision making. There is a vision which is commonly shared, the plans are derived from the vision and the plans are followed step-by-step with clear decision making points. However the ideas come from feeling. James relies on his gut feelings and puts that to the test. He dares to proceed since he has the immediate possibility of testing the decisions and come back to them if it is needed.

> It is based on your gut feeling...then we have those test cases and we follow if it goes well in those cases. [C1P2]

James never strays down the wrong path for too long. That is the reason why James is so quick in his decision making. He knows that he will get immediate feedback and can steer the process in the right direction based on that feedback. James uses gut feelings with a direct feedback loop. If he was not a fast decision maker, the whole process would slow down and eventually collapse.

> We had different streams which operated almost all the time in decision making situations... in every meeting we were able to make decisions...the basis for the decisions where found as deep as your own values... [C1P1]

Typically the interaction with the customer was tight in testing the decisions. James has always been close to customers and he needs to keep that contact alive and sharp in order to get real time feedback into the process. He is also active in forming marketing messages based on innovations. The lead user, beta testing or piloting groups are the typical tool James uses to operate as well as cross-organizational task forces.

The best way to get insight of what is needed is to keep real close to the user – all the time. [C1P3]

The "pipeline" between the customer and development has to be kept to a minimum length so that the information remains as "raw" as possible. This builds the responsiveness and flexibility to the whole process. It also encourages all of the participants when they see that their response is taken immediately into account and it has the effect on the results.

4.4.5 Discoverer

Chris likes to discover new things and he is really enthusiastic towards innovation. Chris is discoverer. There were four interviewees in the group who can be categorized as discoverers.

Chris seeks new paths, new breakthroughs and sometimes selects the wrong way – as all discoverers do. Still he accepts that and tries anew with a different approach or path. Chris is very enthusiastic and personally dedicated to innovation. He has the desire to innovate new things. Chris seeks solutions to problems and he is proud of what he has been able to discover.

It was crystal clear already in the beginning that this [product] has to be done. We innovated and developed that mostly by ourselves... in the background it was still however the customer's problem or need. [C3P1]

Discoverers are deeply involved in problem solving and have a deep understanding of the substance either in a technical context or in a business context. Chris is also active in protecting the intellectual property rights of his inventions - patenting the solutions. Discoverers actively seek room for innovation and search for innovativeness in their customers or their own organization's operations. Chris is usually part of the unofficial team who actively discusses and processes the ideas. ...it is the people in the field who are good at pointing out these flaws and problems, they are always the best innovators. They are the ones who really can dig out the new innovative things. [C2P1]

His decision making has an analytic approach with clear ways of working and clear processes. Mathematical models are used to evaluate different solutions. Sometimes Chris is a stereotypic engineer with problem solving models and finely graded scoring tables. Decision making is labeled depending on the problem solving approach used to solve commercial problems from customers. How does Chris use intuition in his work?

Even though Chris is keen on using clear models to evaluate solutions, decision by feeling is the main reason for agreeing to decisions. The importance of experience and knowledge is the basis for which Chris builds on his intuition. A discoverer goes forward step-by-step like a tester but he is not so dependent on being able to test the solution or decision immediately. Discoverers tolerate uncertainty longer than testers.

4.4.6 Believer

John knows how to proceed, he is very self-confident in his actions. John believes he has the right solution – he is a believer. Believers have a high self-confidence and a trust that they know what they are doing. Two believers were identified in the interviewee group.

Even though John has high degree of self-confidence he works well with the team, he is not going on alone. However, there must be a high degree of trust between the team members. John does not want to waste any effort on solving trust related conflicts inside the team, everyone must be committed to working towards a common goal.

> It needs high commitment from the whole team...Well, it's a trust issue, and if we have different ways to handle the issues inside the team it takes too much effort... [C2P4]

John believes he has correctly assessed the needs and values of the market. He makes assumptions and acts based upon them. Assumptions are highly based on John's opinions or feelings about the case – his intuition. Those feelings are even stronger and more grounded in John's way of acting as with James's, who is a tester.

...we believe that end users value it much...we have an assumption, from which we build up the case... [C1P1]

The strong beliefs are based on strong self-confidence and knowledge. Sometimes John appears to be arrogant in his actions. A believer has the biggest risk of failure, since he blindly trusts his own opinions and intuitions as a self-evident, without questioning them.

4.4.7 Seer

Tom is able to quickly formulate the trends that what he sees happening - he is a seer. Three seers were found in the interview group.

Tom follows the technological debate and can see the best next steps in order to take advantage of those technological advancements. Tom is able to look several steps forward and see commercialization facts about what needs to be taken into account at the very beginning.

> We saw that [technology] will advance... we have seen that we have some strengths... [C4P1]

Tom does not make much noise, but he does very well with both technology and business. However, he is not like Mike – a drifter – because Tom works actively in the process. Tom is sometimes seen as "besserwisser" with his opinions. In that sense he has the same arrogant qualities as John. Tom is able to argue for his opinions so that they gain acceptance.

Intuition appears in Tom's role in the way he "sees" things fitting together. We saw that such technology fits in to this picture...we saw that it fits to our strategy...then we finally saw that it is possible and viable to implement [C3P2].

Tom lobbies internal and external key stakeholders to get his opinions accepted and implemented. With a forward looking ability Tom is able to plan and to take the necessary actions in the right order with the right people. His political approach to the process is not that visible as with George's role. Tom is a senior doer and knows the backgrounds, individuals and previous decisions.

Tom is the most senior technical person on the team and he relies on his competence when it comes to arguing the pros and cons of the solutions. He sees self-evident possibilities with which to combine technological possibilities with commercial needs.

4.4.8 Summary after the third encounter with the data: Decision makers' approaches in using intuition

The interview data revealed seven different role types and approaches that decision makers have in using intuition. These descriptions are adding the knowledge in intuitive decision making research concerning individual decision maker. These types share some common characteristics but it is possible to differentiate them from each other. Some of the roles are closer to each other than others as was described in previous chapters. Figure 4.3 visualizes the connections between role descriptions. The visualization is based on the commonalities between the roles. The commonalities are derived from the role descriptions and the previously presented code-person matrix table (see Table 4.1). There is a connection between the role descriptions if they share some of the codes.



Figure 4.3 Illustration of the role descriptions and their connections (based on commonalities).

A drifter has clear non-active component which is not visible in any other roles. All of the other roles are more active actors in the decision making process. A drifter shares some commonalities with discoverers because they both have a lot of experience in their background. They can also tolerate uncertainty better than other roles. Drifter and thinker values high the importance of experience and knowledge.

A tester has a strong tendency to rely on possibility, to test the results immediately. Testers also share the qualities with believers who believe in their own feelings and have high self-confidence. They both proceed and then check. They act decisively based on gut feelings. A tester has also shared commonalities with thinkers since they both prepare back up plans. Tester and negotiator are close to each other since they have a common characteristic in working close with the customer and prototyping the solutions with the customers.

Results

Believers and negotiators complement each other and thus make a good pair. Negotiators are more management oriented whereas believers have a strong engineering focus. Believers and testers share a common vision, a tendency to guide their actions.

Discoverers are at the extreme in their enthusiastic exploration of new possibilities. No other role type has such a drive towards innovation. They are explorers by nature. A discoverer also follows his own path and does not want too many counterparts to slow down the innovation advancement. A discoverer and seer are located next to each other because of their commonalities and enthusiastic innovation focus.

Thinkers and seers are close to each other. A seer is the most knowledgeable and technically oriented in the whole group and he sees new technological possibilities whereas a thinker is able to see possibilities because of his other non-technical qualities. Seers and thinkers use intuition mainly in the sense-making process phase. Thinkers and testers share the same tendency towards back-up plans. Thinkers proceed more cautiously than testers but they both rely on the existence of a secondary route if the first one fails.

A negotiator is an easy going person who easily moves the group forward along the decision process. He takes the time to develop paths forward together with the team. Negotiators do not like to take an impulsive role on the team – like discoverers. Instead seers, thinkers and testers are the better candidates to negotiators team.

The summary of the role descriptions is shown in role-description matrix in Table 4.2. Descriptions are linked with the description for the use of intuition in decision making, the type of intuition (affective, experience based (heuristic), holistic) and domain in which certain types are typically working (engineering, management) also the way of working is described based on the interview data.

Table 4.2: Role	e descriptions	matrix
-----------------	----------------	--------

	Use of intuition	Intuition type	Domain	Work habit	Note
Drifter,	Usually does not	Holistic. Experience	Engineering	More on his own. Solid	Active and passive
Mike and	recognize the use of	based		company man.	drifters
Sam	intuition. Supportive			Substance expertise	
	use				
<u>Thinker</u> ,	Uses intuition in	Heuristic. Experience	Engineering	Stepwise forward.	
Keith	sensemaking process.	based		Second route prepared.	
	Seeks complementary			Team worker	
	knowledge.				
Negotiator,	Not visible	Holistic	Management	Trust seeking. Analytic	
George				approach. Prototyper.	
				Active teamworker	
Tester,	Uses intuition fast in as	Heuristic	Management	Common visio. Process	
James	early as possible during			oriented. Pushes	
	decision making			forward	
Discoverer,	First intuition then	Holistic	Management	Exploring forward. High	
Chris	analysis			innovation push.	
				Tolerates uncertainty	
Believer,	Assumptions are	Holistic	Engineering	Self confident. Still team	
John	intuition based			player. Strong	
				experience. Risk of	
				mistakes	
Seer, Tom	Intuition guides the	Holistic	Engineering	Technological. Seniority	
	way forward - seeing.				

4.5 Survey questionnaire results

As described in the methods section the Rational Experiential Inventory (REI) self-test was used to study intuitiveness and the use of intuition of decision makers. The study seeks to understand the actual use of intuition in experienced decision makers. The study also seeks to understand the differences between experienced and inexperienced decision makers by using a control group consisting of inexperienced decision makers.

The sample included 18 interviewed experienced decision makers and 86 other decision makers. The demographics are presented in the methods section. Participants completed the long version of REI (Pacini & Epstein, 1999). The long version of REI consists of 40 likert-scale items (1=completely false – 5=completely true), 10 for each subscale (experiential engagement, rational engagement, rational ability, experiential ability). Engagement describes preference for that mode of thinking and ability describes belief in one's own ability to successfully use that mode.

4.5.1 Variable structure

To validate the REI measurement construct structure, a factor analysis was done for measurement constructs. The REI questionnaire items were entered into factor analysis with the Varimax rotation using SPSS Statistics software tool (SPSS). An orthogonally rotated solution was used. Based on the examination of the interpretability factors, four factors were extracted as in original REI.

Based on the analysis, a REI factor model was cleaned by omitting 13 questions/measures because of their unsatisfactory and contradictory factor loadings, so the final factor model included 27 items. The final factor model with the factor loadings is presented in Table 4.3 (The survey questionnaire with full questions is available in Appendix B). This factor model is acceptable but it is to be noted that there was a high number of cross-factor loadings.

	Compon	ent		
	Exp.eng.	Rat.eng	Rat.ab.	Exp.ab.
I often go by my instincts when deciding on a course of action	0,87			
Using my gut feelings usually works well for me in figuring out p	0,74	-0,22	-0,23	
I trust my initial feelings about people	0,63			
I believe in trusting my hunches	0,56	-0,34		-0,24
I tend to use my heart as a guide for my actions	0,55	-0,35	-0,28	
Knowing the answer without having to understand the reasonin	0,54			0,31
I try to avoid situations that require thinking in depth about som	0,52	0,24		0,48
I like to rely on my intuitive impressions	0,50	-0,43		0,24
Intuition can be a very useful way to solve problems		-0,72		0,27
If I were to rely on my gut feelings, I would often make mistake	1	0,71		
I would not want to depend on anyone who described himself o)	0,65		
I suspect my hunches are inaccurate as often as they are accu		0,64		
I don't like situations in which I have to rely on intuition		0,64		
I don't think it is a good idea to rely on oneA's intuition for impo]	0,58		
I can usually feel when a person is right or wrong, even if I can	0,22	-0,40	0,22	
I enjoy solving problems that require hard thinking			0,77	-0,21
I have a logical mind			0,76	
I usually have clear, explainable reasons for my decisions			0,69	
I am much better at figuring things out logically than most peop			0,67	-0,34
I enjoy intellectual challenges			0,67	-0,40
Using logic usually works well for me in figuring out problems ir	-0,20		0,60	
l don't reason well under pressure				0,70
Reasoning things out carefully is not one of my strong points				0,58
I'm not that good at figuring out complicated problems			-0,23	0,56
I don't have a very good sense of intuition		0,31		0,52
I have no problem thinking things through carefully	-0,21			-0,37
I am not a very analytical thinker	0,33		-0,23	0,33

The variable structure between the original REI and the resulting model of the factor analysis is significantly different. However, four extracted factors were characterized by the original REI (experiential engagement, rational engagement, rational ability, experiential ability) variables. The naming of variables is the same as in original REI. Experiential refers to the intuitive terminology in this study. Ability refers to a person's belief in their own ability to use the mode, and engagement refers to one's preference for the mode, Table 4.4.

Table 4.4: REI variable structure

REI variable name	Variable description
Experiential engagement	Persons preference to use experiential methods e.g. intuition in decision making
Rational engagement	Persons preference to use rational methods in deci- sion making
Rational ability	Person's belief in their own ability to successfully use rational methods in decision making.
Experiential ability	Persons belief in their own ability to successfully use experiential methods in decision making

A four factor (cleaned) solution explained 48.44% of the total variance (19.65%, 14.62%, 8.54%, 5.63%). The scree plot is presented in Figure 4.4. The explanatory power of the model would have been better if more factors (six factor model) could have been taken into account in the model, but since the original REI included 4 variables the same number of variables were used here.

Scree Plot



Figure 4.4: Scree plot of cleaned factor model.

Originally, the 10 items for experiential ability was reduced to six items. Cronbach's inter-item coefficient alpha for the experiential ability is 0.60. The rational ability items were also reduced to six. Cronbach's inter-item coefficient alpha for the rational ability was 0.81. Rational engagement constructs were reduced to seven items. Cronbach's interitem coefficient alpha for the rational engagement is 0.76. The remaining eight items were used to operationalize experiential engagement. Cronbach's inter-item coefficient alpha for the experiential engagement is 0.81.

4.5.2 Descriptive statistics and correlations

Descriptive statistics and correlations of the used (cleaned) measurement constructs are presented in Table 4.5. Engagement and the ability to have strong positive correlations were expected. Negative correlation in experiential ability and experiential engagement could reflect intuitive abilities a person have that he is not – at least intentionally – using. The more abilities he has the more cautious he is in using those abilities.



Table 4.5: Descriptive statistics

4.5.3 Results

The difference between intuitive and rational engagement and the abilities of experienced and inexperienced decision makers was compared in

Results

this study. Mean values and standard deviations were calculated for the groups based on their REI test results. The results of the comparison between the experienced and inexperienced managers showed that there are no significant differences in the engagement or ability on either the rational side or experiential/intuitive side. This result is clearly adding the knowledge in intuitive decision making research, since results from the previous research has shown that experienced decision makers use intuition more than inexperienced decision makers.

A T-test was used to verify the statistical validity of the results. The results of this test are presented in Table 4.6. No statistically significant difference was found between the results from the experienced and the inexperienced decision makers. Also there was no significant difference in the distribution between the experienced and the inexperienced decision makers (x^2).

Table 4.6: Results, Chi-Square (x²) and t-test

Expe	Experienced, n=1		Other, n=68				
Target group: experienced or not	mean	s.d.	mean	s.d.	x ²	t	p.
What is your accumulated working experience ?	4,50	0,79	3,25	1,23	16,44		n.s.
How many years have you been in your current org.	2,67	1,46	1,43	0,58	29,65		n.s.
What is your highest education?	2,22	0,94	1,81	0,89	6,78		n.s.
Experiential engagement	2,89	0,69	2,81	0,54		-0,54	n.s.
Rational engagement	3,42	0,60	3,34	0,56		-0,51	n.s.
Rational ability	4,16	0,44	4,01	0,54		-1,05	n.s.
Experiential ability	4,02	0,35	3,95	0,52		-0,63	n.s.

The experienced decision makers analyzed in this study had high scores in both rational and experiential abilities and were close to the median in rational and experiential engagements. The control group, consisting of inexperienced decision makers had almost the same results. This challenges the previous research, which has found noticeable differences between the experienced and inexperienced decision makers (Dane & Pratt, 2007; Harper, 1988; Isenberg, 1984; Sadler-Smith & Shefy, 2004).

The matrix of role descriptions which was presented in chapter 4.4 is added to the REI scores and is presented in Table 4.7. The results did not provide any new insights into the use of intuition between different
roles. Intuition type (affective, heuristic, and holistic) or domain (engineering, management) did not show any noticeable difference in the REI results. This may also be because of the relatively weak reliability of the REI model and its ability to explain in detail the use of intuition.

Table 4.7:	Role	descriptions	matrix
------------	------	--------------	--------

	Use of intuition	Intuition type	Domain	Work habit	Note	REI (RatExp.)
Drifter,	Usually does not	Holistic. Experience	Engineering	More on his own. Solid	Active and passive	High-High
Mike and	recognize the use of	based		company man.	drifters	
Sam	intuition. Supportive			Substance expertise.		
	use of intuition.					
Thinker,	Uses intuition in	Heuristic. Experience	Engineering	Proceeds stepwise		High-High
Keith	sensemaking process.	based		forward. Second route		
	Seeks complementary			prepared. Team worker.		
	knowledge.					
Negotiator,	Not visible.	Holistic	Management	Trust seeking. Analytic		High-Med
George				approach. Prototyper.		
				Active teamworker		
Tester,	Uses intuition fast as	Heuristic	Management	Common vision. Process		High-High(med)
James	early as possible during			oriented. Pushes		
	decision making.			forward.		
Discoverer	First intuition than	Holistia	Managament	Evaloring High		Lligh Lligh
Discoverer,	First intuition then	HOIISTIC	ivianagement	Exploring. High		Hign-Hign
Chris	anaiysis.			Telerates upgertainty		
Poliovor	Accumptions are	Holistia	Engineering	Colf confident, still toom		High Mod
believer,	Assumptions are	HOIISTIC	Engineering	sen connuent, sun team		nigh-ivieu
John	intuition based.			player. Strong		
				experience. Risk of		
Coor Torr		11-1:-+:-	Farris e e sin e	mistakes.		= = / =)
seer, Iom	intuition guides the	HOUSTIC	Engineering	rechnological seniority.		High-High(IOW)
	way forward - seeing.	l	1	L		

4.6 Synthesis

Based on the interview analysis, decision making plays a significant role in innovation management. This was reported as one of the core categories after the first encounter with the interview data. Decision making core category was further analyzed and non-rational decision making elements came up as one of the key elements during the data analysis. These non-rational elements were grouped under a common term – intuition.

The second encounter with the data concentrated on studying and analyzing how intuition reveals itself in the decision making and innovation management front end. It resulted in a four faceted intuition frame description with core categories: *Symptoms, preceding and simultaneous*

Results

enablers and safeguards. Intuition appears in different ways in decision making and the symptoms for of the use of intuition vary. There are prerequisite elements (enablers) which are needed for a fruitful environment for the use of intuition in decision making. Temporally close to intuition use are intuition enablers and safeguard components. Enablers ease the use of intuition and safeguard are there to verify that the decision was acceptable and that there are ways to reflect and learn from the decision making process. Preceding enablers and safeguards are organizational-level items which need to be in place to make the use of intuition possible – either to foster it or to secure it. Preceding enablers work further away from the use of intuition, but are needed to build grounds for the use of intuition.

The third encounter with the data revealed what approaches and individual roles decision makers have when using intuition. Seven different role descriptions and approaches in using intuition emerged from the data (*drifter, thinker, negotiator, tester, discoverer, believer and seer*). These approaches that decision makers have to using intuition clearly differ between roles. There are different roles and identities amongst the persons using intuition. Roles have different qualities which can be described and evaluated. This provides the basis that there is a possibility for changing and developing the use of intuition in decision making. The role descriptions share some characteristics but it is possible to differentiate them from each other. Some roles work better together in a team than others as was described in chapter 4.4.

Figure 4.5 describes the relationships in the intuition framework and the role description components categorized by individual, environmental and organizational perspectives of innovation management. The categorization of those three perspectives used in this study is the theoretical lens used to organize the innovation literature (Balachandra & Friar, 1997; Reid & de Brentani, 2004; Van de Ven, 1986). In Figure 4.5 the theoretical framework of individual, environmental and organizational perspectives of innovation management (chapter 2) are linked to the developed four faceted intuition frame based on interview data analysis done in chapter 4.3 of this study, including the role descriptions developed in chapter 4.4. It is acknowledged that this linkage is not as straightforward as it now seems. The synthesis is to be taken within the





Figure 4.5: Synthesis and relations of intuition frame and three perspectives to innovation management

The survey conducted on experienced and inexperienced decision makers also showed that decision makers use intuition in their actions. Generally speaking no significant difference was found between experienced and inexperienced decision makers in the use of intuition, both use intuition equally. The mean in ability to use intuition is on high level and the mean in engagement is on average level as shown in Table 4.6. The result is interesting since it challenges the prevailing view that experienced decision makers differ from inexperienced decision makers in their use of intuition. This is elaborated more on discussion chapter of this dissertation. If the use of intuition is looked at in more detail on experienced decision makers' role level, some differences were found between the different roles. Heuristic based intuition type role, tester, is making decisions more based on rational analysis than on intuition. Tester relies on having the possibility to test the results of the decision and he makes the decision fast, but he keeps the rational analysis as a foundation where to base the intuitive decision.

Also holistic based but engineering domain role, believer, base the decision making more on rational analysis than on intuition. Believer's fist assumptions come more from intuition but the final analysis is rational. The role descriptions were formulated to describe the different approaches what decision makers' have in using intuition and not to link those directly to the interviewed persons. That's why the results of the survey questionnaire cannot be interpreted straightforward vis-à-vis to the roles.

Cognitive frameworks and mental models are used to describe intuitive decision making. Cognitive frameworks are developed based on work, life experience and education. Both cognitively based intuition (experience, seeing the bigger picture, sense making) and affectively based intuition (gut feelings) are needed for effective innovation decision making. The combination of both, the holistic perspective combines the two original descriptions. This study defines the approaches that decision makers have in their use of intuition and linked them to the three intuition definitions (affective, experience based, holistic).

Holistic intuition was the most frequent type of intuition. This could be because all of the interviewed and surveyed respondents were relatively experienced. Even though the control group in the survey consisted of inexperienced decision makers, they were not newcomers to the business nor to engineering. The importance of experience and cumulated tacit knowledge is noticeable. Intuitive decision making is based on past experience and knowledge.

"The really valuable thing is intuition" Albert Einstein

The results of this dissertation are discussed in the following chapters in relation to the existing literature in order to present the contribution made by this this research. In order to further elaborate the research findings, the results of this dissertation are discussed more broadly than in the other literature that was presented in the literature review section. This is done in order to present possible new avenues for future research and to extend the scope of the findings to other relevant literature. The literature review section was purposely as compact as possible. I wanted to present other literature here also, which was brought up while writing of results and discussion sections.

This chapter is organized as follows. Firstly, research questions are used as a framework to analyze the results of this dissertation. Secondly, the theoretical contribution is made visible based on innovation and NPD front end literature as well as decision making and intuition literature. The contribution to intuition research instruments is also discussed. Thirdly, practical implications of this dissertation are highlighted. Fourthly, the study is evaluated using criteria for qualitative and quantitative (survey questionnaire) research. The final chapter of this dissertation proposes the avenues for future research.

5.1 Responses to research questions

The purpose of this research has been to explore the front end of innovation, to study how innovation projects gets decided upon, who makes the decisions and how those people make decisions. The aim of this research has been to gain a deeper understanding of decision makers' use of intuition and to offer new theoretical end empirical insights in the use of intuition in decision making at the front end of innovation. This aim was further divided into three research questions which are answered based on the empirical findings of this study and compared to the existing literature.

Question 1: How does intuition reveal itself in innovation front end decision making?

First encounter: Decision makers' experiences of innovation management

In the first encounter with the research data, the evidence from the interviews with experienced decision makers, revealed three categories affecting innovation management: <u>decision making</u> elements of innovation management, <u>innovation enablers</u> and <u>compelling forces</u> that push the innovation process forward. The decision making category was studied further in this dissertation. The two other categories also play a major role in innovation management and some discussion of this occurs in later parts of this chapter, but the detailed analysis for those categories is left to forthcoming studies to research further.

Decision making

The interviewees in this study all discussed the early phases of the innovation process where the selection and decisions concerning new innovation were in the focus. The findings of this study support previous research concerning the early phases (front end) and the importance of decision making at the front end of innovation (Adams et al., 2006; Calantone et al., 1999; Cormican & O'Sullivan, 2004; Khurana & Rosenthal, 1998; Krishnan & Ulrich, 2001). One of the shared codes between the decision making elements and innovation enablers was a "clear way of working", which is especially connected to the innovation process. The importance of decision making in innovation management in general and NPD in particular was also remarkable in this study. These results complement the previous research in innovation management from the environmental side e.g. (Christensen & Raynor, 2003; Eisenhardt, 1989c; Leifer et al., 2000; Tushman & Anderson, 1987; Utterback, 1994), from the individual side e.g. (Chakrabarti, 1974; Gemunden et al., 2007; Rothwell & Gardiner, 1985; Schmidt et al., 2001; Thamhain & Wilemon, 1987; Tushman, 1977) and from the organizational side e.g. (Adams et al., 2006; Cooper & Kleinschmidt, 1995; Khurana & Rosenthal, 1998; Krishnan & Ulrich, 2001; Reid & de Brentani, 2004) where the importance of decision making has also been highlighted.

The interviewed persons in this study were all senior and experienced decision makers. Their roles in the companies are closely related to gate keeper, champion and promoter roles. This study has made visible the individual roles in innovation decision making with the role-specific characteristics and their appearance and behavior in decision making at the innovation front end. The results of this dissertation add new evidence to previous research, touching on champions and promoters (Chakrabarti, 1974; Gemunden et al., 2007), and boundary spanning and gatekeeper roles (Reid & de Brentani, 2004; Tushman, 1977; Tushman & Katz, 1980) by showing how decision making is present in their work. Individuals had different roles and positions in decision making incidents and that is seen from the presented role descriptions. The role of decision making at the innovation front end is reflected by the approach that each individual has to innovation front end decision making. The individual component is answered in more detail during the discussion of the second and third research question and later in this chapter.

The importance and difficulty of decision making is accentuated at the front end phase of innovation where the uncertainty is highest. The core category "decision making" was analyzed further in more detail during the first encounter and especially during the second encounter with the full interview data. The "decision by feeling" category and the sub codes like "spontaneous decision making" were noted in the majority of the front end decision making descriptions. That guided the analysis to look what is happening in those situations.

When looking more closely at the decision making at the innovation front end, as described by the interviewees, one of the decision making elements found was intuition. The use of intuition is highlighted in the discussions and answers of the experienced decision makers in the results of the interview analysis and questionnaire survey results. The intuition was not purposely highlighted in the interviews, but the use of intuition can be found in the way experienced decision makers describe their behavior and decision making practices. These results challenge the prevailing view about the superiority of rational decision making methods in innovation management (Cooper, 1990; Khurana & Rosenthal, 1997; Krishnan & Ulrich, 2001) and in new product development (Büyüközkan & Feyzioglu, 2004; Calantone et al., 1999; Montoya-Weiss & Calantone, 1994; Souder & Mandakovic, 1986), where the development of decision making models has been to concentrate on building new rational and analytical methods. However, the everincreasing amount of data and complexities and uncertainty in fast paced decision making situations will make this approach even more challenging in the future. The results of this study provides new evidence and knowledge to that discussion and develops the picture of the use of intuition in decision making in innovation management as a one path to going forward in developing innovation management practices.

This study challenges previous conceptions and emphasizes approaches other than rational decision making methods at the innovation front end. The findings support intuitive decision making literature findings concerning strategic decision making (Khatri & Ng, 2000; Miller & Ireland, 2005; Sinclair & Ashkanasy, 2005; Sinclair et al., 2009), uncertainty in decision making (Kahneman, 2003a; Tversky & Kahneman, 1974) and managerial decision making (Agor, 1986; Dane & Pratt, 2007; Isenberg, 1984; Sadler-Smith & Shefy, 2004; Shapiro & Spence, 1997; Simon, 1987). Based on the results of this study, intuition plays a large role in strategic decision making (decision making concerning innovation management can be regarded as strategic). The existence of intuition in decision making was discovered in the experienced decision makers' expressions when they discussed their innovation management practices. The findings of this study complement the results of previous studies in highlighting intuition as viable tool in NPD front end decision making (Dayan & Di Benedetto, 2011; Hart et al., 2003; Stevens & Burley, 2003; Tzokas et al., 2004; Van Riel et al., 2011; Yahaya & Abu-Bakar, 2007). This was uncovered in more detail during the following encounters with the research data.

Innovation enablers

The other two categories that emerged during the first encounter with the data were <u>innovation enablers</u> and <u>compelling forces</u>. <u>Innovation enablers</u> are organization and environment-related aspects that enable innovation to surface and continue in an organization. Innovation enablers was a natural category since the interviews were about innovation management. The discussions easily went along the paths describing what made innovations possible in companies. All of the companies are building offerings for the marketplace, so it is natural that "customer orientation" and "productization with the customer" were raised as the key codes in the data analysis.

The organizations in this study varied from mid-size to large global organizations. However, the individuals interviewed from large global organizations were discussing innovations within their team or unit. One common theme that was raised by the majority of the interviews was related to innovating with the customer – regardless of whether the customer was another company or a general consumer. This does not make any difference between different organizational structures or the size of the organization and challenges previous research where the size of the company size had been found to have an effect on innovation success (Rothwell, 1983, 1993). Whatever the size or the structure the close linkages to the customers and users are needed in order to make the right decisions. These findings support previous research that organizational set-up - customer oriented organizations – has a positive effect on innovation success (Khurana & Rosenthal, 1998). Organizations of all sizes and structures should be organized so that customer orientation is achieved.

The final decision at the front end is often dependent on an individual decision maker. Findings from this study revealed the differences in the innovativeness of the decision makers and differences in how decision makers use intuition in decision making situations. The role of individuals in innovation management has been researched from the individual competencies point of view and also from a team composition point of view. Previous research on the individual side has presented different role descriptions (champions, gate keepers, boundary spanners, promoters) that have differ-

ent qualities in innovation projects (Chakrabarti, 1974; Gemunden et al., 2007; Tushman, 1977; Tushman & Katz, 1980; Zirger & Maidique, 1990). The results of this study complement previous research by explaining how those individuals use intuition at the front end of innovation.

The interviewed persons highlight "complementary knowledge" from a team composition point of view. The team where knowledge and competencies exist in the different areas fostered innovativeness in the interviewed companies. The importance of communication networks has been highlighted in the previous research (Allen, 1970). The same aspects of team composition have been raised by Thamhain and Wilemon (Thamhain & Wilemon, 1987). They found that team performance is related to leadership, job content, personal needs and general work environment. One of the driving elements in a work environment was team communication. Previous research has found that the best performing teams had a diversity of individuals (Day & Schoemaker, 2004; Schmidt et al., 2001; Thamhain & Wilemon, 1987). Even though the results of this study explain the use of intuition at an individual level, innovation in organizations is typically a team-level activity and the importance of communication is evident. Recent studies concerning team intuition found that teams use intuition in turbulent environments (Dayan & Di Benedetto, 2011). This supports the results of this study, since the studied NPD environments are usually turbulent.

Compelling forces

The third innovation management category, <u>compelling forces</u>, is what is needed to push innovation forward in the organization. People and organizations need compelling forces to make the necessary decisions and actions. Compelling forces are related to environmental issues in innovation management. One environmental characterization is a high velocity environment.

The evidence from the interviews support previous research findings on the importance of decision making in innovation management and changing decision making approaches especially in high velocity environments – the environments researched in this study (Eisenhardt, 1989c, 1999). Compelling forces in high velocity environments, especially the "competing in the marketplace" and "changing business environment" codes, highlighted the existence of compelling forces and the importance of decision making as one component for successful innovation management. These decision making styles are faster and the room for rational and analytical analysis is limited because of compelling forces. Innovation management in the software or telecom industry can be regarded as a high velocity environment and the decision making approach has the same characteristics. The life cycle aspects already raised by Abernathy and Utterback (Abernathy & Utterback, 1978) present different decision making approaches based on the industry life cycle phase. High growth and innovative industries have different decision making approaches than do low growth in low innovation industries.

<u>Second encounter: Facets on using intuition at the front end of innova-</u> <u>tion</u>

The second encounter with the research data revealed the use of intuition in more detail and resulted in describing the "four faceted intuition frame". That is one of the significant findings of this study. These facets provide a means to study the use of intuition in more concrete way. The appearance and use of intuition can be characterized based on the <u>symptoms</u>, <u>preceding</u> <u>enablers</u>, <u>simultaneous enablers and safeguards</u> (see Figure 4.2).

<u>Symptoms</u> indicate that intuition is present in decision making despite that the lack of "visible" signals. This is how to discover the use of intuition, since the use of intuition is often unconscious and not directly noticeable. The interviewed persons did not directly speak about the use of intuition. Instead they used expressions like: decision by feeling, spontaneous decision making, unconscious vision, individual thought process, indecisiveness and decision by accident. There are thus no reliable instruments available to uncover it otherwise. That was the challenge of quantitative instruments in studying intuition and one of the reasons the main instruments and methods of this study were qualitative.

<u>Preceding enablers</u> are subjects that need to be present so that the intuition is used in the decision making process. When decision making occurs in close proximity, <u>simultaneous enablers</u> work in favor of using intuition in decision making.

When the decision is already complete, <u>safeguards</u> are helping the decision maker to verify the "correctness" of the intuitive decisions which were done. The existence of safeguards can be seen as one kind of intuition enabler for the decision maker. Safeguards can also be seen as one means of increasing the flexibility in the innovation process (Buganza & Verganti, 2006). Safeguards are also related to reflective learning research (Ortt & Smits, 2006; Perminova et al., 2008), by proposing one element which enables reflective learning in innovation management.

These findings support the intuitive decision making literature e.g. (Dane & Pratt, 2007; Hodgkinson et al., 2008; Miller & Ireland, 2005; Sinclair & Ashkanasy, 2005; Sinclair et al., 2009) by providing new evidence and more detailed descriptions in the form of "enablers" of intuition in decision making and adds empirical evidence from the innovation front end. The discovery of symptoms also adds on the research concerning how practitioners experience intuition in their decision making (Raami et al., 2010). The use of intuition in decision making must be seen as a combination of symptoms, enablers and safeguards as shown in the Figure 4.2. In that way the full picture of the use of intuition in decision making is better understood.

The following chapters continues answering research question 1 and opens up the research findings of this study to new literature that was not evaluated in the literature review chapter. This is done in order to openly evaluate the findings and give the possibility of opening new research avenues based on the research findings.

When the results of the first and second encounter with the research data are further elaborated, the findings of this study can also be related to complex evolving systems (CES) (Mitleton-Kelly, 2003) or complex adapting systems (CAS) (Brown & Eisenhardt, 1997; Buijs, 2003; Chiva-Gomez, 2004; McCarthy et al., 2006). These research streams complement the study of innovation management and decision making contexts. Complex adapting systems are in constant change and do not operate in predefined ways where fixed processes and purely rational decision making approaches can be used. The role of intuition in decision making in complex adapting systems is notable since intuitive decision making is adaptable and fast and is able to cope with the constant change that all CAS/CES environments have. McCarthy et al. (McCarthy et al., 2006) have viewed the NPD as a complex adapting system and proposed changing and adaptable decision making rules to cope with the situation. The use of intuition in decision making is one solution for that need.

The use of intuition in decision making in innovation management can also be explained as decision making in complex environments as described by Aaltonen (Aaltonen, 2007). Multi ontology decision making presentation can be complemented by adding intuitive decision making components in relation to complex ontology. If intuition and the use of intuition is seen to be the component in complex ontology, it is different than heuristics, since heuristics are more related to ordered ontology. The existence of heuristics, especially accessibility and availability, must not be underrated (Kahneman, 2003b). Heuristics are existent but the proportion of heuristics in decision making remains unclear. Intuition can still play a major role in complex environments and decision making situations.

The description of intuitive decision making in relation with multiontology sense making complements the multi-ontology sense making conception and connects the use of intuition in decision making in complex environments. Complex environments are environments where the most actively developing innovative industries exist and the tools needed to survive in these environments – like the use of intuition in decision making – are helping in surviving in those environments.

<u>Question 2: What approaches do decision makers have when</u> <u>using intuition at the innovation front end?</u>

<u>Third encounter: Alternative approaches to decision makers' use of in-</u> <u>tuition</u>

Answers to the previous research question discussed innovation management, decision making in the innovation front end and the use of intuition in decision making. Those answers guided the research towards more detailed questions concerning individual decision makers. The following chapters answer these questions in more detail.

The use of intuition is different based on the role that the decision maker has in the innovation front end. Decision makers have different approaches to the use of intuition in innovation decision making. Seven role characteristics describing the approach to the use of intuition were found from the interview data: <u>Drifter, thinker, negotiator, tester, discoverer, believer and seer</u>. Individuals in all of these roles use some proportion of intuition and rationality in how they make decisions but their approach to the use of intuition differs based on the role. These role descriptions can be regarded as a new knowledge especially in intuition literature and research. Figure 4.3 illustrates the relationships between the role descriptions based on commonalities (common codes) that they share.

Drifter, discoverer and thinker are roles in which use of intuition is based on long time experience or technological knowledge. The most senior technological person using intuition is seer. They appear differently in their organization's decision making situations, but the basis for their intuitive ability is their professional and decision making experience.

Negotiator is the role that uses intuition the least. He has a holistic approach to decision making and decisions are mainly based on an analytical approach. A negotiator has many of the same qualities as the believer. A believer is more engineering focused whereas negotiator is management-focused person. They both have a common vision to guide their actions even though intuition is not visible in their actions.

Testers are similar to believers and negotiators in their guidance by vision. However, testers are more in favor of experience based - heuristic intuition whereas believers and negotiators have a more holistic approach.

Why are the role descriptions different? Since intuition and its use has developed based on experience and holistic view on issues, so has the context of the experience, educational background, and personal interest areas affected on the characteristics of how the individual uses intuition. It is a character trait and very person dependent. However, it is possible to form a set of role descriptions with some commonalities, as has been done in other role descriptions before (Belbin, 1993; Briggs & Myers, 1962; Gemunden et al., 2007; Mintzberg, 1973; Tushman, 1977; Tushman & Katz, 1980).

As the literature review reveals, empirical evidence and the use of intuition – the approach the decision maker has – is mixed. The role descriptions described in this study – a significant finding - depict a set of roles and their approach to the use of intuition from experienced decision makers' viewpoints working in the NPD context. There are more roles or different descriptions if inexperienced decision makers are added. A different context would have also produced a different set of characteristics and roles as the decision making style and the use of intuition is context dependent (Crittenden & Woodside, 2006; Eisenhardt, 1989c; Papadakis et al., 1998; Schoemaker, 1993).

The role descriptions presented in this study can be further elaborated with classical team role and managerial role descriptions by Belbin (Belbin, 1981, 1993) or Myers and Briggs (Briggs & Myers, 1962). The use of intuition is clearly more related to Belbin's intellectual monitor-evaluator (ME) team roles or Myers Briggs Intuition-Thinking (NT) roles. The monitor-evaluator analyses the problems and the level of intuition in this analysis could be elaborated. How are these intuitive roles related in detail to Belbin's team roles or MBTI roles is left for further studies to evaluate.

The decision makers' use of intuition and the role descriptions can also be evaluated together with other decision maker managerial role descriptions from previous research (Mintzberg, 1973). The decisional roles presented by Mintzberg (1973) were: entrepreneur, resource allocator, disturbance handler and negotiator. All of these roles could have an intuitive component in their decision making approach. Especially when combined with his later research concerning unstructured decision processes (Mintzberg et al., 1976), the results of this study could have revealed more insight into the connection between unstructured decision processes and decision makers' role descriptions. A combination of these role descriptions would be a fruitful addition to manager role description research.

Isenberg's (1984) description of senior managers' five different ways of using intuition (sensing the problems existence, rapid reactions, building integrated pictures, check for rational analysis, by-pass analysis) has some similarities to the role descriptions found in this study. The findings of this study are based on an NPD context and support previous findings of the

differences between different decision makers' approach to the use of intuition.

This study and the roles presented here have commonalities with all of the roles described in previous research concerning champions, promoters, boundary-spanners and gate-keepers (Chakrabarti, 1974; Gemunden et al., 2007; Reid & de Brentani, 2004; Tushman, 1977). These descriptions are derived from innovation the management context as are the roles from this study. The roles and the new approaches described here bring more understanding on decision makers' use of intuition and how that may reflect on managerial activities. All of the previously mentioned roles have to be engaged in decision making and the use of intuition in their decision making would be interesting topic for further research. Since managerial decision making is not always rational but based on intuition, the group decision making within decision makers using intuition differently would also be a topic for coming research. The approaches found in this study could act as a research framework for coming studies.

The role descriptions of this study were made based on studying the innovation front end. Based on the descriptions that the interviewees gave of their activities in innovation decision making, it can be said that the activity of the roles is different depending on the process phase. Seers and discoverers are more active at the very beginning of the innovation process because they have a strong innovative and forward looking stance in their actions. A tester is more active after decisions have been made and negotiators, thinkers and drifters are more active throughout the process.

The different role descriptions in previous research all state the different behavior of roles in different phases of the process, at different levels or positions in the organization or in different contexts. The findings in this study portray an additional description of roles that are active at the innovation front end, having experience in decision making in an NPD related context. The results provide new insight to the approach to the use of intuition, which has not been achieved using quantitative research instruments (Van Riel et al., 2011). These results provide more understanding on managers' use of intuition and how that can be reflected on decision making and managerial activities.

<u>Question 3: How do experienced decision makers differ from</u> <u>inexperienced decision makers in their use of intuition?</u>

The use of intuition did not differ significantly between experienced and inexperienced decision makers. Both have a high ability in intuitive decision making and they also use intuition in innovation decision making. Both groups have a high level in ability and average level in engagement for rational decision making. This can be considered as a new knowledge in intuitive decision making literature and research.

Uncovering the similarities between experienced and inexperienced decision makers is interesting. Previous research has shown that decision making style and practices changes and develops when you are more experienced or you have more knowledge e.g. (Raami et al., 2010; Stevens & Burley, 2003; Wally & Baum, 1994; Yahaya & Abu-Bakar, 2007). Similar findings from the research connected to leadership style and change development has shown that behavior and leadership style changes when you switch from a junior position to a senior position e.g. in champion role (Howell & Higgins, 1990). This study did not confirm these previous empirical findings. There are several reasons why the findings of this study are different than in previous research.

It is possible that the engineering context itself had already "selected" the similar types of persons in decision making positions. Previous research has shown that the context has an effect on decision making (e.g. (Crittenden & Woodside, 2006; Eisenhardt & Martin, 2000; Papadakis et al., 1998; Schoemaker, 1993). Despite one's experience, individuals also tend to use similar practices and methods in an engineering context. This is especially related to personal ability to use intuitive or rational methods.

Innovation decision making, especially at the front end of innovation is uncertain by nature and decision making practices can be quite similar because of the context of the decision. It is to be noted that not all of the inexperienced decision makers in this study's control group were directly connected to innovation decision making. But certainly, in the context of engineering companies, fast decision making situation in general are directing the decision making practices. Context and fast decisions are related to intuitive/rational engagement. The role of the individual and the use of intuition can be stronger especially in radical/discontinuous innovations. This supports the findings of Reid and de Brentani (Reid & de Brentani, 2004, p. 182).

One reason for similarities in decision making style comes from the personality of the decision makers. It is possible that the personality types of these two companies are closer to each other, also because of the aforementioned engineering context. Is it the case that the engineering context as a growth path to managerial positions is relatively short compared to other industries? This could be the case, since the personalities and decision making practices of experienced and inexperienced managers are so similar.

The answer to this third research questions was achieved using the available quantitative research instrument. However, the construct validity of the Rational Experiential Inventory (REI) instrument left some problems to be solved. This criticism has also been raised in previous research (Hodgkinson et al., 2006; Pretz & Totz, 2007; Sinclair et al., 2009). The validity of REI as a research instrument needs to be evaluated in further studies. The acceptable factor model used in this study differed from the original REI factor model significantly and certainly needs further evaluation and development. The reliability of the model used in this study was not high. However, REI is the model used in many studies and the results received in this study add to the knowledge gained in previous studies that used the same model.

In this study, the original English REI question set was used. This may have had an effect on the results because almost all of the respondents were native Finnish speakers. It is always a matter of how you understand and interpret the questions, especially if the questions are not in your native language. However, all of the survey questionnaire respondents and interviewed persons have used English as their working language for several years, so language barriers should not largely affect their understanding of the questionnaire.

5.2 Theoretical contribution

The main contribution of this research is to intuitive decision making research. The research also contributes to the innovation front end and to NPD research. Contributions and new research avenues for the other research (e.g. sense making) are also briefly discussed in later chapters. This dissertation contributes also to the intuitive decision making research instrument body of knowledge by proposing a fruitful means of studying the use of intuition in decision making and challenging the validity and reliability of some of the prevailing survey based research instruments.

5.2.1 Contribution to intuitive decision making research

This study contributes to the intuitive decision making discussion by increasing our understanding of the innovation front end (Dane & Pratt, 2007; Hodgkinson et al., 2008; Matzler et al., 2007; Miller & Ireland, 2005; Sinclair et al., 2009) since the innovation front end has not been the context of previous intuitive decision making studies.

The results of this dissertation contribute to intuitive decision making research along several avenues. Firstly it brings new evidence and examples to the measurement of the use of intuition (Burke & Miller, 1999; Hodgkinson et al., 2008; Kahneman & Klein, 2009). Secondly, since decision making at the front end is judged to be strategic and contain a degree of uncertainty, the findings also support the previous strategic decision making and decision making under uncertainty research where use of intuition has received attention (Agor, 1989a; Eisenhardt, 1989c; Khatri & Ng, 2000; Miller & Ireland, 2005; Sinclair et al., 2009; Tversky & Kahneman, 1974). Thirdly, in previous research, intuition has received considerable research interests concerning managerial decision making (see e.g. Agor, 1985; Dane & Pratt, 2007; Isenberg, 1984; Sadler-Smith & Shefy, 2004; Shapiro & Spence, 1997; Simon, 1987). This dissertation challenges some of the previous findings, for example the difference between senior and junior managers' use of intuition. However, for the most part the findings in this research complement previous research descriptions of decision makers' use of intuition.

The results of this study showed no significant difference between experienced decision makers and inexperienced decision makers in how they used intuition in decision making. This result challenges the previous research where senior managers have been found to use intuition more often than do lower level managers (Agor, 1986, 1989a; Allinson & Hayes, 1996; Khatri & Ng, 2000; Parikh, 1994; Raami et al., 2010). By providing empirical evidence for the use of intuition by experienced and inexperienced decision makers, this study contributes to the discussion of decision making between these different roles in the workplace. The comparison of manager to decision maker could make some differences to the results, but the evidence suggests that in this context, senior manager is equivalent to experienced decision maker.

The study adds empirical evidence not only to the existence of intuition in decision making but also to how intuition appears in decision making cases and how different actors use intuition in decision making as explained by the presented role descriptions of decision makers (drifter, thinker, negotiator, tester, discoverer, believer, seer) (Dane & Pratt, 2007; Pretz & Totz, 2007). The description of roles and their approach to the use of intuition in particular provides new definitions to the intuitive decision making research. The use of intuition can also be seen as a component that is flexible in the innovation process. This is something which has been a requirement in the previous research (Buganza & Verganti, 2006).

Both *cognitively* based intuition (seeing the bigger picture, sense making) and *affectively* based intuition (gut feeling) is needed for effective innovation decision making. A combination of the two original descriptions is referred to as a *holistic* perspective. The holistic perspective acknowledges both aspects of intuition and does not see them as conflicting or exclusive. Instead, the holistic approach sees both sides of intuition as working simultaneously even though they are separate cognitive systems (Dane & Pratt, 2007; Hodgkinson et al., 2008; Sinclair et al., 2009). This study contributes to this discussion by defining the roles of intuition and the links that those role descriptions have to the three aforementioned intuition definitions.

5.2.2 Contribution to intuition research instruments

Intuition as a phenomenon is a challenging research object. This is because the use of intuition is not necessarily visible in the people's actions. Individuals do not always know or recognize the use of intuition. Some might even deny having used intuition in decision making because it is not regarded as a reliable tool as rational decision making models. There are several quantitative and survey questionnaire based research instruments which are constructed as self-tests (Allinson & Hayes, 1996; Epstein, 1990; Epstein et al., 1996; Forgas, 1995; Hodgkinson & Sadler-Smith, 2003a; Sinclair et al., 2009). The Rational-Experiential Inventory (REI) survey questionnaire, which is based on the Cognitive Experiential Self Theory (CEST) was used in this study (Epstein, 1990, 1994). The results of the survev questionnaire and the factor analysis support the previous findings by e.g. Sinclair et al. (Sinclair et al., 2009) that the reliability of the Rational Experiential Inventory (REI) is questionable. The modified REI factor model that was used in this study was able to show the ability to use the intuition and a preference for the use of intuition (engagement) in the decision making as was predicted. In that way it partially supports the use of REI as a survey instrument for the study of intuition.

The qualitative research approach and grounded theory method that was used in this research as the main method, was found to be explanatory and usable tool for the study of the use of intuition. This is because it allows for and even requires an interpretive analysis phase. This phase helps in making sense of what happens and is sometimes unstructured and difficult to analyze in intuitive decision making (Charmaz, 2006; Miles & Huberman, 1994).

New research instruments can be built on based on the categorization of roles in this study and thus provide a better understanding of the use of intuition in decision makers' behavior. This could open up intuitive decision making to further research and help in building new ways of fostering and developing managerial and organizational skills in this area. The "four faceted intuition frame" could be used as a framework to study the use of intuition in different contexts and research set-ups.

5.2.3 Contribution to innovation management research

The innovation management and NPD literature in particular has used three perspectives to discuss innovation management: environmental, individual and organizational (Cooper & Kleinschmidt, 1995; Kleinschmidt et al., 2007; Reid & de Brentani, 2004; Trott, 2005). The findings of this study add a new dimension to those descriptions (see also Table 2.7) by positioning the findings of the use of intuition in decision making (symptoms, preceding enablers, simultaneous enablers, safeguards and roles) to the perspectives as shown in the Figure 5.1.



Figure 5.1: The appearance of intuition (*in Italic*) in innovation management perspectives

The symptoms provide an indication that intuition is being used at front end decision making. The preceding enablers and safeguards are processlevel items which can be used in the development of organizational processes (Cooper, 2008; Khurana & Rosenthal, 1998; Reid & de Brentani, 2004). Simultaneous enablers are characteristics related to the organizational environment which could be taken into account when making an organization more intuition friendly (Christensen & Raynor, 2003; Eisenhardt, 1989c; Tushman & Anderson, 1987). The findings of this study also contribute and support the findings of Reid and de Brentani that the use of intuition is high for radical/discontinuous innovations (Reid & de Brentani, 2004). The description of the roles and how they use intuition in decision making complement a category of individual innovation management research (Chakrabarti, 1974; Gemunden et al., 2007; Tushman, 1977; Tushman & Katz, 1980). The role descriptions presented in this dissertation can be compared to the previously defined roles of gate keeper, champion, promoter and boundary spanner.

This study increases the understanding of the innovation front end by revealing the use of intuition in decision making in innovation front end. Uncertainty and complexity is highest during the front end phase (Davila, 2000; Kim & Wilemon, 2002b; Nobelius & Trygg, 2002; Zhang & Doll, 2001) and the findings of this study shows that the use of intuition is also high in that phase. Intuition was found to be an important component of innovation front end decision making based on interviews with experienced decision makers. This study contributes to innovation management research by highlighting the use of intuition as a decision making technique especially at the front end of innovation (Adams et al., 2006; Cooper, 1990; Cormican & O´Sullivan, 2004; Hart et al., 2003; Khurana & Rosenthal, 1998; Krishnan & Ulrich, 2001).

The findings support previous research in the importance of intuition in NPD screening gate decisions (Hart et al., 2003; Tzokas et al., 2004). The results of this study add a new viewpoint to previous results by explaining how the decision makers use intuition in screening gate decisions. This study also concentrates on how senior (experienced) decision makers operate. In previous studies, this was raised as an area requiring further research (Yahaya & Abu-Bakar, 2007).

This dissertation brings new evidence and support to previous NPD (NBD) research concerning the personalities of decision makers (Stevens & Burley, 2003; Van Riel et al., 2011). Stevens and Burley found out that intuitive individuals and intuitive decision making is the most important factor in early stages of the NBD process. They used a MBTI based criteria for evaluating personalities. The results of this dissertation and the role descriptions add on to previous descriptions of the intuitive personality at the front end of innovation. The importance of the experience and the use of intuition of decision makers was highlighted by Van Riel et al. (Van Riel

al., 2011). The results of this dissertation support and expand upon their work.

The NPD environment has been evaluated as a high velocity and uncertain environment. The use of intuition has been found to be one of the main components in decision making when the environment is turbulent (Dayan & Di Benedetto, 2011). Dayan and Di Benedetto have studied team intuition in particular and the results of this dissertation complement and partially challenge their findings in the sense that intuitive judgment has to be looked at from an individual perspective in order to get a fundamental and more detailed picture of the phenomenon.

By taking a grounded theory approach in this study, it was possible to build a new framework for the use of intuition in the innovation front end with the descriptions of intuitive decision making characteristics ("four faceted intuition frame") in innovation management and with the role descriptions of intuitive decision makers as shown in Figure 5.1. The results of this study show that a qualitative approach is in general one of the most suitable ways to research intuition because it is complex in nature and not explicitly defined. Therefore it is not quantifiable and measurable in the same way as some other phenomena e.g. rational problem solving and decision making.

5.2.4 Contribution to decision making and sense-making research

Managers sense-making and decision making activities are tied to their cognitive frameworks or mental models. Past experiences shape their template for the understanding of future experiences (Weick, 1995). Intuition used in complex environments (as is in innovation front end) is based on experience - cognitive frameworks and mental models (Aaltonen, 2007). The framework is developed based on work and on life experience and education. Frameworks and models are developed over time and through experience and interaction with others (Aaltonen, 2007; Kahneman, 2003b). This study contributes to this discussion by describing mental models (in this study, the approach that the decision makers have) of different actors in the innovation management process and their properties and mental models as specified in Table 4.7.

A systems intelligence approach sees decision making and decision makers in complex environments that are closely engaged to their environments in order to make the right decisions. This study adds to the discussion of how organizations can adopt systems-thinking in their operations (Ackoff, 2006; Hämäläinen & Saarinen, 2008a; Saarinen & Hämäläinen, 2004). This adoption comes from understanding and nurturing the intuitive aspects of intelligent decision making.

5.3 Practical implications

This dissertation is tightly connected with field and practice – the actual work of innovation management. The interview based qualitative and interpretative approach allowed me to get into the challenges of the innovation front end and decision making. Also, my own participation in innovation management ensured a seamless connection with real life practices. In that sense, the results of this study stem from practice and empirical data and observations.

It is clear, based on the findings of this dissertation, that intuition plays a major role in innovation management, especially at the front end phase. Organizations need to recognize the role of intuition and see how it can be best utilized in their respective environments. It is also clear that decision makers in engineering companies operating in NPD are using intuition with mainly positive outcomes. Thus the question for the organizations is how to foster intuitiveness and how to use intuition in the organization.

The use of rational decision making has dominated practices for a long time. Basically the development of all decision making, decision making models and managerial education has been based on analytical and rational models. It is time to explore non-rational decision making models and acknowledge the existence of the use of intuition in decision making in innovation management. It is still to be noted that rationality must have a clear and strong position in managerial decision making, otherwise objectivity, and traceability and likely the reliability of managerial work and decision making is endangered.

At the individual level, the practical implications support individuals to develop and trust their own intuitive abilities and enhance the use of intuition in decision making. The existence and significance of intuition in a successful decision maker was clearly shown in the findings of this study. Again, as noted earlier, intuition must not be used as the sole criteria for decision making instead a holistic aspect that uses both the experience based and affect based intuition complements for analytical decision making.

In the future, complex and fast changing decision making environments will mean that the importance of the sense-making will increase (Aaltonen, 2007; Brown & Eisenhardt, 1997; Eisenhardt, 1989c; Tidd & Bessant, 2009). Traditional, rationally justified, sense making way may not be enough to cope with the fast changing decision making environments. As Aaltonen described, you need to have a second order change in order to respond to change that is occurring in the environment (Aaltonen, 2007). Intuition can be one tool to be used in that second order change.

Despite the fact that the study does not present a straight forward theory or model to develop innovation management practices in organizations, it provides a means of developing the use of intuition in decision making (importance of experience and knowledge and openness to intuition) and it could spark some ideas about how and when intuition is suited to innovation front end practices. At least the existence of intuition is clear and must be recognized by all practitioners.

When organizations are developing their decision making practices they need to first know how intuition is used in their decision making and in what way it is used. By identifying the symptoms and decision makers' decision making methods, organizations can recognize that intuition is used in decision making. If spontaneous decision making, decision based on feelings and visional decision making is found, the use of intuition is probably being used in decision making. Even though the reliability of survey tools to evaluate the individual decision maker was not found to be very high in this research, the use of these tools can help in evaluating the individual preferences and abilities in the use of intuition. Especially as the tools develop, there are fast and easy ways of getting started in developing the use of intuition. In the following Figure 5.2, some of the means of fostering the use of intuition are described. This can be done by recognizing and developing the enablers and safeguards as was found in this research. Firstly, by increasing the knowledge and experience at a personal or organizational level the possibility for holistic intuition is raised. This is done by using normal training and on-the-job learning practices. It should also be noted that one of the enablers was "clear working practices". This means that the use of intuition is fostered when organizations already have clear routines and processes in place where they can base normal operations and the choice to use intuition in decision making is given to the individual.

Secondly, by simply recognizing that intuition plays a role – conscious or unconscious - the threshold for the use of intuition is lowered. This can be done by accepting intuition as a valid component of decision making. Common values supporting the use of intuition also play a major role. As was noted in high-velocity decision making situations, sometimes the push in the decision making situation enables decision makers to use intuition. Therefore, building on decision making situations so that time is limited can also foster the use of intuition.

Thirdly, as a safeguard for the use of intuition, possibilities for testing and piloting decisions early enough is also helpful in the use of intuition and maturing of decisions. Different piloting practices and reflective learning practices are helping in that area. The availability of a sponsor for the project or the decision was found to be one component that supports the use of intuition in decision making.



Figure 5.2: The ways to foster use of intuition.

The creation of situations and contexts in the organizations where the use of intuition is made easier could also encourage organizations and managers to use and trust intuition in decision making. One way to do this is by fostering creativity techniques and by having open discussions in organizations about intuition.

Intuition and how companies can make the best use of it in their businesses must not be underrated. This needs to be taken into account at all levels of an organization. During the last years, the use of intuition has been discussed in management literature and most effective ways to use it are constantly being evaluated (Campbell & Whitehead, 2010; Dayan & Di Benedetto, 2011; Kahneman, 2010; Kahneman & Klein, 2009; Raami et al., 2010; Sinclair et al., 2009).

5.4 Evaluation of the study

The evaluation of the study is done in three phases. This is because the study contains both qualitative and quantitative parts, each of which requires different evaluation criteria. First the overall research setting and the study in general is evaluated. Secondly the qualitative part of the study is evaluated and finally the quantitative part of the study is considered.

To increase the trustworthiness of this dissertation I used a multi-method approach. The main content and the results were derived by using qualitative grounded theory method which was then validated and strengthened using a quantitative survey questionnaire. I have also kept the entire research process and documentation as transparent as possible in order to give the reader clear view about what has been done and how I ended up with these conclusions of the results that were presented in the results and discussion chapters. The structure throughout the dissertation follows the same logic – from the research questions through literature review and results up to the discussion – to help the reader to follow and judge the contribution of this research.

The selection of the interviewed decision makers and the companies was done in the ICT industry. The ICT industry is generally regarded as a highly innovative high velocity industry and is therefore a good field in which to study innovation decision making. Because of the high velocity, lots of decisions are made all of the time. Due to the speed of change, new innovations are needed constantly. In order to broaden the study beyond the software industry, decision makers from two companies from the telecom sector were included in the interview pool. The selection of four companies with 19 in-depth interviews has provided fairly good coverage of the ICT industry in general. Even though all the interviews were done in Finland and to with Finnish decision makers, the companies operate internationally and thus the results are not limited to the Finnish context. The respondents to survey questionnaire also included non-Finnish participants.

The selection of the interviewees was done using the prime contact in the company. The person was asked to identify a group of experienced persons who have been closely connected to innovation decision making in the company (I was the prime in one of the companies and selected the persons from the company I was working with). The control group for the survey questionnaire part of the study was done by selecting all of the persons involved in decision making that had little experience with decision making. The level of experience was rated based on the job title and the time an individual had been in that position (i.e. lower level managers with less than two years in the position).

This research used the grounded theory method (Charmaz, 2006; Glaser & Strauss, 1967; Strauss & Corbin, 1998). I have used several techniques in this research as described by Miles and Huberman in order to make sense

of what is happening in the data (Miles & Huberman, 1994). The techniques used and described in the method section of this dissertation includes: Counting, clustering, noting patterns, factoring and categorizing, noting relations between variables, finding intervening variables and building a logical chain of evidence as well as making theoretical and conceptual coherence.

Qualitative research can be evaluated using the criteria presented in detail by Miles and Huberman (Miles & Huberman, 1994). Grounded theory research can be evaluated based on the techniques developed by Glaser and Straus's. Their technique evaluates of how well your constructed theory renders the data by criteria of fit, work, relevance and modifiability (Glaser & Strauss, 1967; Strauss & Corbin, 1998). Other important criteria are discipline in the research process, evidence and aesthetic issues. Criteria presented by Charmaz for evaluating grounded theory studies are (Charmaz, 2006): Credibility, originality, resonance and usefulness.

I have evaluated the qualitative part of this research by using the following criteria: <u>Credibility, usefulness and transferability, originality, confirm-</u> <u>ability (objectivity), resonance (applicability)</u> and <u>aesthetics</u>.

To increase <u>credibility</u> I have described in detail how the interviews were conducted and how the results were derived from the interview data. Cross tabulation and dependency networks are used to illustrate how the logic of the thought process leads up to the results and conclusions. The interviews were conducted in the interviewees' native language to ensure the accurate interpretations. A second person transcribed the interview recordings to improve the credibility of the interview data. All of the quotes used in the dissertation are translated and available in their original and translated forms. The path from quotes to codes to categories is made clearly visible in the tool used in the analysis (ATLAS.ti, 2006).

It is acknowledged that the interview process between interviewer and the interviewee is always subjective of nature and reflects the both persons' backgrounds. I tried to keep the structure as open as possible without forcing the interviewee along any predefined direction. This was achieved by having lightly structured interview and open ended interview question structure. <u>Usefulness or transferability</u> refers to the applicability of the findings to contexts outside the original research context. The companies in the study are all NPD oriented. This fact evidently brings one challenge to the transferability of the results. It is clear that NPD engineering differs greatly from e.g. service oriented companies. How much this is reflected in the decision making is not self-evident. In both cases when we look at the early phases of innovation management, there is uncertainty and complexity in play during decision making. If the environment is turbulent there is also a sense of urgency which requires speed in the decision making. In that sense, the decision making contexts may be similar and the findings of this study might be transferable to other industries.

The selection of experienced decision makers for the interviews leaves one handicap in the transferability. The findings, especially the role descriptions, are relevant only to experienced decision makers and cannot be applied directly to for example junior programmers. The presented intuition framework and decision making elements are independent of the role descriptions and can be used in other contexts as well. The only question would relate to transferability limitations between NPD engineering and other types of organizations.

In qualitative grounded theory research the replicability and confirmability are always discussed because of the subjectivity brought in by the selected research method. Instead <u>originality</u> is the criteria presented by Charmaz to evaluate the freshness of the findings and the significance of the work. When evaluating the originality of the results, this study has new and fresh categories and an analysis with new conceptual ideas. It also takes the theoretical presentation of the use of intuition in decision making further. Dependability refers to the replicability of the research process and findings. The same methods that were used to improve the credibility also apply to dependability. The research process and the documentation of the research process and its findings are to support the replicability of the research.

<u>Confirmability</u> or objectivity concerns the neutrality of the research. In qualitative methods, the research is often subjective. This is the case here. The researcher as the main instrument in the research means there is a subjective flavor to the findings. The methods used in interview outlining, tran-

scriptions, systematic coding, testing the initial frameworks with other people increase the objectivity of the research. The results of the study can be said to be confirmable, at least to some extent, because of the discipline and documentation used during the research process.

My own role in one of the companies studied and my previous knowledge in the field of innovation management of course highlights one of the biggest concerns about the objectivity of the findings. I have acknowledged it throughout the research process and tried to deliberately mitigate the possible subjectivity of the findings. On the other hand, I could have used some ethnographic elements in this study since I have been personally involved in the process of innovation front end decision making.

<u>Resonance</u> refers to the applicability of the results. Do the results make sense to the participants? Do the categories portray the studied experience in its entirety? Based on discussions that I had after the analysis with some of the interviewed persons, the results do make sense. The resonance is noticeable. The way the data was analyzed through three consequent encounters was to ensure that the presented categories and the analysis broadly covered in the studied experience.

When it comes to the quantitative part of the study, one limitation is that the instruments are not yet well defined and commonly accepted. This is a problem for all quantitative intuition research (Sinclair et al., 2009). There is still a journey to go when defining the reliable quantitative intuition research instruments.

This study was done in engineering companies which could have effect on the results. Probably the persons working in NPD engineering context are naturally similar to one another independent of their experience or position. This highlights domain specificity as was noted also by Pretz and Totz (Pretz & Totz, 2007). The reduced REI that was constructed in this study should be used for other than engineering related contexts in order to test the generalizability of the modified model.

As noted, the construct validity of the REI was questioned based on this analysis. Similar findings of the REI validity have also been reported in other studies (Hodgkinson et al., 2008; Hodgkinson et al., 2006; Pretz & Totz, 2007).

The quantitative part of this dissertation plays a smaller, more like confirmatory, role in this study. The evaluation of that part is relatively straight forward and is presented in the relevant chapter of the dissertation. As a summary, the evaluation showed that there are some challenges to the factor model as presented by the REI model. Still it was possible to construct the factor model which had the acceptable level of reliability that is typical when using survey questionnaires.

The criterion discussed above says little about the aesthetics of the dissertation – how the researcher writes and reports the findings and arguments. When evaluating these kinds of issues in particular the subjectivity of the writer can dominate, which is why I am not evaluating it here. Intuition as a concept is a challenging research topic and provides the researcher with flexibility in how he or she reports the findings. This has been found in the previous research in how intuition is used in decision making. Thus the assessment of aesthetic issues is left for the reader to evaluate.

5.5 Ideas for further research

This study has concentrated on a narrow part of the innovation management problematic, the front end of new product development. That is why the study leaves many questions available for further research. The focus on the front end phase of the innovation management process opens up the role of intuition in other parts of the process. Is the launch phase so much different than the front end concerning the use of intuition? Naturally the phases are different, but they share some common characteristics like uncertainty and complexity. The uncertainty and complexity are just of a different nature than at the front end.

Even though I was able to study the innovation front end over a couple of years it leaves open the possibility for some longitudinal aspects of the study. The practices and the organizations need more than a year or two to develop. So the following questions remain unresolved: How do organizational practices in the use of intuition change when company develops or matures? How does the use of intuition develop in an individual over their carrier? What is the effect of the context on the use of intuition? Each of these questions could spawn an extensive research path. The NPD context itself was also only a narrow view of one kind of engineering company. There are vast numbers of different companies, cultures etc. that still need to be researched.

The role descriptions of this study can be researched in connection with classic team roles (Belbin, 1981; Briggs & Myers, 1962), managerial roles (Mintzberg, 1973) or champion, gate-keeper, boundary-spanner and promoter role descriptions (Chakrabarti, 1974; Gemunden et al., 2007; Reid & de Brentani, 2004; Tushman, 1977) for combined and complete descriptions of the roles acting in the innovation front end. This would increase the understanding of decision makers' roles and of roles in general and help in further developing the decision making practices in innovation management. How do the roles develop over the time would also be an interesting research topic. The results of this study post a snapshot of the role descriptions, some longitudinal elements could have been inserted into the study but that would have required substantially more time in the data gathering phase. Maybe this is just a starting point for the next research journey.

On the methodological side, the instruments used in the study of intuition need more elaboration. The intuitive decision making research path itself has been quite diverse and there is some immaturity of the methods and instruments. This study has contributed significantly, but there is still much to be done. The grounded theory method typically does not produce any formal or grand theory but the results of this study and the research and results of further studies will provide a basis for future theories.

The use of intuition in general needs to be better recognized in education and in research. The courses of study at universities concerning decision making seldom touch on the use of intuition in decision making. Intuition does not easily fit into to prevailing traditional management research discussions. It is likely that intuition and the use of intuition has been, until now, treated as an unscientific area of research and has thus been neglected in the majority of training programs in universities. This is a defect that needs to be corrected.

6 References

Aaltonen, M. 2007. *The Third Lens: Multi-Ontology Sense-making and Strategic Decision-making*. Hampshire: Ashgate.

- Abernathy, W., & Utterback, J. 1978. Patterns of Industrial Innovation. *Technology Review*, 80: 40-47.
- Ackoff, R. L. 2006. Why Few Organizations Adopt Systems Thinking. Systems Research and Behavioral Science, 23: 705-708.
- Adams, R., Phelps, R., & Bessant, J. 2006. Innovation Management Measurement. a Review. *International Journal of Management Reviews*, 8(1): 21-47.
- Agor, W. H. 1985. The Logic of Intuition: How Top Executives Make Important Decisions. *Organizational Dynamics*: 5-18.
- Agor, W. H. 1986. How Top Executives Use Their Intuition to Make Important Decisions. *Business Horizons*, January-February 1986: 49-53.
- Agor, W. H. (Ed.). 1989a. *Intuition in Organizations: Leading and Managing Productively*. Newbury Park: SAGE publications.
- Agor, W. H. 1989b. Test Your Intuitive Powers: AIM survey. In W. H. Agor (Ed.), *Intuition in Organizations: Leading and Managing Productively*. Newbury Park: SAGE Publications.
- Aho, E., Brunila, A., Eriksson, J.-T., Harjunen, P., Heikinheimo, R., Karjalainen, S., Kekkonen, T., Neittaanmäki, P., Ormala, E., Peltonen, P., Pöysti, K., Strengell, M., Stenros, A., Teperi, J., & Toivanen, H. 2008. *Kansallinen innovaatiostrategia*. Helsinki, Finland: Ministry of Employment and the Economy.
- Allen, T. J. 1970. Communication Networks in R&D Laboratories. **R&D** Management, 1: 14-21.
- Allinson, C. W., & Hayes, J. 1996. "The Cognitive Style Index: A measure of intuition-analysis for organizational research". Journal of Management Studies, 33: 119-135.
- Armstrong, S. J., & Hird, A. 2009. Cognitive Style and Entrepreneurial Drive of New and Mature Business Owner-Managers. *Journal of Business and Psychology*, 24(4): 419-430.
ATLAS.ti. 2006. The knowledge workbench: www.atlasti.com.

- Bacon, G., Beckmann, S., Mowery, D., & Wilson, E. 1994. Managing Product Definition in High-Technology Industries: A Pilot Study. *California Management Review*, 36(3): 32-56.
- Balachandra, R. 1984. Critical signals for making Go/NoGo decisions in new product development. Journal of Product Innovation Management(1): 92-100.
- Balachandra, R., & Friar, J. H. 1997. Factors for Success in R&D Projects and New Product Innovation: A Contextual Framework. *IEEE Transactions on Engineering Management*, 44(3): 276-287.
- Banning, M. 2008. A review of clinical decision making: models and current research. *Journal of Clinical Nursing*, 17: 187-195.
- Bantel, K., & Jackson, S. 1989. Top Management and Innovations in Banking: Does the Composition of the Top Team Make a Difference? Strategic Management Journal, 10: 107-124.
- Barnard, C. I. 1938. *The Functions of the Executive*. Cambridge: Harvard University Press.
- Barney, J. B. 1991. Firm Resources and Sustained Competitive Advantage. Journal of Management, 17(1): 99-120.
- Belbin, R. M. 1981. *Management Teams: Why They Succeed or Fail.* London: Heinemann.
- Belbin, R. M. 1993. *Team Roles at Work*. Oxford: Butterworth-Heinemann.
- Benner, P., & Tanner, C. 1987. Clinical Judgment How Expert Nurses use Intuition. American Journal of Nursing, 87(1): 23-31.
- Birkinshaw, J., Bessant, J., & Delbridge, R. 2007. Finding, forming and performing: creating networks for discontinuous innovation. *California Management Review*, 49(3): 67-84.
- Bohrnstedt, B. W. 1983. Measurement. In P. H. Rossi, J. D. Wright, & A. B. Anderson (Eds.), *Handbook of Survey Research*: 69-121: Academic Press.
- Booz, Allen, & Hamilton. 1968. *Management of New Products*. New York: Booz, Allen & Hamilton Inc.
- Borgatti, S. P. 2002. **NetDraw: Graph Visualization Software**. Harvard, MA: Analytic Technologies.
- Borgatti, S. P., Everett, M. G., & Freeman, L. C. 2002. Ucinet for Windows: Software for Social Network Analysis. Harvard, MA: Analytic Technologies.
- Boulding, W., Morgan, R., & Staelin, R. 1997. Pulling the plug to stop the new product drain. *Journal of Marketing Research*(26): 164-176.

- Brentani, U. d. 1986. Do firms need a custom-designed new product screening model? *Journal of Product Innovation Management*, 3(2): 108-120.
- Brentani, U. d., & Reid, S. 2012. The Fuzzy Front-End of Discontinuous Innovation: Insights for Research and Management. *Journal of Product Innovation Management*, 29(1): 70-87.
- Briggs, I., & Myers, K. 1962. *The Myers-Briggs type indicator*. Palo Alto: Consulting Psychologists Press.
- Brown, S. L., & Eisenhardt, K. 1995. Product Development: Past Research, Present Findings and Future Directions. Academy of Management Review, 20(2): 343-378.
- Brown, S. L., & Eisenhardt, K. 1997. The Art of Continuous Change: Linking Complexity Theory and Time-Paced Evolution in Relentlessly Shifting Organizations. *Administrative Science Quarterly*, 42(1): 1-35.
- Buganza, T., & Verganti, R. 2006. Life-Cycle Flexibility: How to Measure and Improve the Innovative Capability in Turbulent Environments. *Journal of Product Innovation Management*, 23: 393-407.
- Buijs, J. 2003. Modelling Product Innovation Processes, from Linear Logic to Circular Chaos. *Creativity and Innovation Management*, 12(2): 76-93.
- Burgelman, R. A., & Sayles, L. R. 1986. *Inside Corporate Innovation*. New York: Macmillan.
- Burke, L. A., & Miller, M. K. 1999. Taking the Mystery Out of Intuitive Decision Making. Academy of Management Executive, 13(4): 91-99.
- BusinessWeek. 2009. S&P/Business Week Global Innovation Index. New York: Bloomberg.
- Büyüközkan, G., & Feyzioglu, O. 2004. A fuzzy-logic-based decision-making approach for new product development. *International Journal Production Economics*(90): 27-45.
- Cagan, J., & Vogel, G. M. 2002. Creating breakthrough products: innovation from product planning to program approval: Prentice Hall.
- Calantone, R., Di Benedetto, A. C., & Schmidt, J. B. 1999. Using the analytic hierarchy process in new product screening. *Journal of Product Innovation Management*, 16: 65-76.
- Campbell, A., & Whitehead, J. 2010. How to test your decision-making instincts *McKinsey Quarterly*, Vol. May 2010: 1-4.
- Chakrabarti, A. K. 1974. The Role of Champion in Product Innovation. *California Management Review*, 17: 58-62.

- Chapman, C., & Ward, S. 1997. *Project Risk Management: Processes, Techniques and Insights*. Chichester, UK: Wiley.
- Charmaz, K. 2006. Constructing Grounded Theory A Practical Guide Through Qualitative Analysis. London: SAGE Publications Ltd.
- Chiva-Gomez, R. 2004. Repercussions of Complex Adaptive Systems on Product Design Management. *Technovation*, 24(9): 707-711.
- Christensen, C., & Raynor, M. 2003. *The Innovator's Solution: Creating and Sustaining Successful Growth*. Boston: Harvard Business School Press.
- Christensen, C. M. 2003. *The Innovator's Dilemma: When New Technologies Cause Great Firms to Fail* (3 ed.). Cambridge: HBS Press.
- Cohen, M. D., March, J. G., & Olsen, J. P. 1972. A Garbage Can Model of Organizational Choice. Administrative Science Quarterly, 17: 1-25.
- Cohen, W., & Levinthal, D. 1990. Absorptive Capacity A New Perspective on Learning and Innovation. *Administrative Science Quarterly*, 35: 128-152.
- Cooper, R. G. 1988. The new product process: a decision guide for management. Journal of Marketing Management, 3(3): 238-255.
- Cooper, R. G. 1990. Stage-gate systems: a new tool for managing new products. *Business Horizons*, 33(3): 44-53.
- Cooper, R. G. 1994. Perspective: Third generation new product process. *Journal of Product Innovation Management*, 11: 3-14.
- Cooper, R. G. 1998. *Winning at new products: accelerating the process from idea to launch* (2 ed.). London: Addison Wesley Publishing Company.
- Cooper, R. G. 1999. From Experience: The Invisible Success Factors in Product Innovation. *Journal of Product Innovation Management*, 16: 115-133.
- Cooper, R. G. 2008. Perspective: The Stage-Gate Idea-to-Launch Process -Update, What's New, and NexGen Systems. *Journal of Product Innovation Management*, 25: 213-232.
- Cooper, R. G., & Kleinschmidt, E. J. 1987. New Products What Separates Winners from Losers. Journal of Product Innovation Management, 4(3): 169-184.
- Cooper, R. G., & Kleinschmidt, E. J. 1995. Benchmarking the Firm's Critical Success Factors in New Product Development. *Journal of Product Innovation Management*, 12(5): 374-391.

- Cormican, K., & O'Sullivan, D. 2004. Auditing Best Practice for Effective Product Innovation Management. *Technovation*, 24: 819-829.
- Crittenden, V. L., & Woodside, A. G. 2006. Mapping Strategic Decision Making in Cross-functional Contexts. *Journal of Business Research*, 59: 360-364.
- Dalkey, N., & Helmer, O. 1963. An Experimental Application of the Delphi Method to the Use of Experts. *Management Science*, 9(3): 458-467.
- Dane, E., & Pratt, M. G. 2007. Exploring Intuition and its Role in Managerial Decision Making. Academy of Management Review, 32(1): 33-54.
- Davila, T. 2000. An empirical study on the drivers of management control systems' design in new product development. *Accounting, Organizations and Society*(25): 383-409.
- Day, G., & Schoemaker, P. 2004. Driving through the fog: managing at the edge. *Long Range Planning*, 37: 127-142.
- Dayan, M., & Di Benedetto, C. A. 2011. Team intuition as a continuum construct and new product creativity:The role of environmental turbulence, team experience and stress. *Research Policy*, 40: 276-286.
- Dean, J. W., & Sharfman, M. P. 1996. Does Decision Process Matter? A Study of Strategic Decision-Making Effectiveness. Academy of Management Journal, 39(2): 368-396.
- Denzin, N., & Lincoln, Y. S. (Eds.). 1994. *Handbook of qualitative research*. Thousand Oaks, CA: Sage.
- Doctor, R. N., Newton, D. P., & Pearson, A. 2001. Managing uncertainty in research and development. *Technovation*(21): 79-90.
- Dougherty, D., & Hardy, C. 1996. Sustained Product Innovation in Large, Mature Organizations: Overcoming Innovation-to-Organization Problems. Academy of Management Journal, 39(5): 1120-1153.
- Drejer, A. 2002. Situations for innovation management: towards a contingency model. *European Journal of Innovation Management*, 5(1): 4-17.
- Drucker, P. 1985. *Innovation and Entrepreneurship*. New York: Harper&Row.
- Dufva, M. 2008. Emotions, Decision Making and Systems Intelligence. In R. P. Hämäläinen, & E. Saarinen (Eds.), Systems Intelligence: A new lens on human engagement and action. Espoo: Helsinki University of Technology, Systems Analysis Laboratory.
- Eisenhardt, K. 1989a. Agency Theory: An Assessment and Review. *The Academy of Management Review*, 14(1): 57-74.

- Eisenhardt, K. 1989b. Building Theories from Case Study Research. *The Academy of Management Review*, 14(4): 532-550.
- Eisenhardt, K. 1989c. Making Fast Strategic Decisions in High-Velocity Environments. *Academy of Management Journal*, 32(3): 543-576.
- Eisenhardt, K. 1999. Strategy as Strategic Desicion Making. *Sloan Management Review*, Spring 1999: 65-72.
- Eisenhardt, K., & Bourgeois, L. J. 1988. Politics of strategic making in highvelocity environments. Academy of Management Journal, 31(4): 737-770.
- Eisenhardt, K., & Zbaracki, M. J. 1992. Strategic Decision Making. *Strategic Management Journal*, 13: 17-37.
- Eisenhardt, K. M., & Martin, J. A. 2000. Dynamic Capabilities: What Are They? *Strategic Management Journal*, 21: 1105-1121.
- Epstein, S. 1990. Cognitive-experiential self-theory. In L. Pervin (Ed.), *Handbook of personality theory and research*: 165-192. New York: Guilford Press.
- Epstein, S. 1994. Integration of the cognitive and the psychodynamic unconscious. *American Psychologist*, 49: 709-724.
- Epstein, S., Pacini, R., Denes-Raj, V., & Heier, H. 1996. Individual Differences in Intuitive-Experiential and Analytical-Rational Thinking Styles. *Journal of Personality and Social Psychology*, 71(2): 390-405.
- Ferguson, G. 1999. Cracking the Intuition Code: Understanding and Mastering Your Intuitive Power. Chicago: Contemporary Books.
- Flanagan, J. C. 1954. The Critical Incident Technique. *Psychological Bulletin*, 51(4): 327-358.
- Forgas, J. P. 1995. Mood and judgement: The Affect Infusion Model (AIM). *Psychological Bulletin*, 117: 39-66.
- Francis, D., & Bessant, J. 2005. Targeting innovation and implications for capability development. *Technovation*, 25(3): 171-183.
- Frantz, R. 2003. Herbert Simon: Artificial Intelligence as A Framework for Understanding Intuition. Journal of Economic Psychology, 24: 265-277.
- Garcia, R., & Calantone, R. 2002. A Critical Look at Technological Innovation Typology and Innovativeness Terminology: A Literature Review. Journal of Product Innovation Management, 19(2): 110-132.
- Gardner, W. L., & Martinko, M. J. 1996. Using the MBTI to study managers: A literature review and research agenda. *Journal of Management*, 22(1): 45-83.

- Garfield, E. 1983. Citation data as science indicators. *Essays of an Information Scientist*, 6: 580.
- Gemunden, H. G., Salomo, S., & Hölzle, K. 2007. Role Models for Radical Innovations in Times of Open Innovation. *Creativity and Innovation Management*, 16(4).
- Gigerenzer, G. 2007. *Gut Feelings: Short Cuts to Better Decision Making*. London: Penguin Books.
- Gilovich, T., Griffin, D., & Kahnemann, D. (Eds.). 2006. *Heuristics and Biases: The Psychology of Intuitive Judgment*. Cambridge: Cambridge University Press.
- Glaser, B. G. 1978. *Theoretical sensitivity*. Mill Valley, CA: Sociology Press.
- Glaser, B. G., & Strauss, A. 1965. *Awareness of dying*. Chicago: Aldine Transaction.
- Glaser, B. G., & Strauss, A. L. 1967. *The Discovery of Grounded Theory: Strategies for Qualitative Research*. Chicago: Aldine Transaction.
- Glaser, M. 1995. Measuring intuition. *Research Technology Management*(38): 43-46.
- Goldberg, L. 1970. Man versus a moidel of man: A rationale plus some evidence, for a method of improving clinical inferences. *Psychological Bulletin*, 73(4): 422-432.
- Grant, R. 1996. Toward a Knowledge-Based Theory of the firm. *Strategic Management Journal*, 17(Winter Special Issue): 109-122.
- Hambrick, D. C., & Mason, P. A. 1984. Upper Echelons: The Organization as a Reflection of Its Top Managers. *Academy of Management Review*, 9(2): 193-206.
- Harper, S. C. 1988. Intuition: What Separates Executives from Managers. *Business Horizons*, September-October 1988: 13-19.
- Hart, S., Hultink, E. J., Tzokas, N., & Commandeur, H. R. 2003. Industrial Companies' Evaluation Criteria in New Product Development Gates. *Journal of Product Innovation Management*, 20: 22-36.
- Hayashi, A. M. 2001. When to Trust Your Gut. *Harvard Business Review*, February 2001: 59-65.
- Higgins, J. M. 1995. Innovation: The Core Competence. *Planning Review*, 23(6): 32-35.
- Hippel, E. v. 1988. *The Sources of Innovation*. Cambridge: Oxford University Press.

- Hippel, E. v. 2006. *Democritizing Innovation* (1 ed.). Cambridge: MIT Press Books.
- Hodgkinson, G. P., Langan-Fox, J., & Sadler-Smith, E. 2008. Intuition: A fundamental bridging construct in the behavioural sciences. *British Journal of Psychology*, 99: 1-27.
- Hodgkinson, G. P., & Sadler-Smith, E. 2003a. Complex or unitary? A critique and empirical re-assessment of the Allinson-Hayes Cognitive Style Index. *Journal of Occupational and Organizational Psychology*, 76: 243-268.
- Hodgkinson, G. P., & Sadler-Smith, E. 2003b. Reflections on reflections: On the nature of intuition, analysis and the construct validity of the Cognitive Style Index. *Journal of Occupational and Organizational Psychology*, 76: 279-281.
- Hodgkinson, G. P., Sadler-Smith, E., & Sinclair, M. 2006. More than Meets the Eye? Intuition and Analysis Revisited, Academy of Management Annual Meeting. Atlanta, Georgia.
- Hogarth, R. M. 2001. *Educating Intuition*. Chicago: University of Chicago Press.
- Howell, J. M., & Higgins, C. A. 1990. Champions of Technological Innovation. *Administrative Science Quarterly*, 35: 317-341.
- Hultink, E. J., Hart, S., Robben, H. S. J., & Griffin, A. 2000. Launch Decisions and New Product Success: An Empirical Comparison of Consumer and Industrial Products. *Journal of Product Innovation Management*, 17: 5-23.
- Hämäläinen, R. P., & Saarinen, E. 2008a. Systems Intelligence The Way Forward? A Note On Ackoff's "Why Few Organizations Adopt Systems Thinking". Systems Research and Behavioral Science, 25: 821-825.
- Hämäläinen, R. P., & Saarinen, E. (Eds.). 2008b. Systems Intelligence: A New Lens on Human Engagement and Action. Espoo: Helsinki University of Technology, Systems Analysis Laboratory.
- Isenberg, D. J. 1984. How Senior Managers Think. *Harvard Business Review*, November-December 1984: 81-90.
- Jones, R. W. 1958. Management of New Products. *The Journal of Industrial Engineering*, IX(5): 429-435.
- Jung, C. G. 1926,1989. Psychological types. London: Routledge.
- Kahneman, D. 2003a. Maps of Bounded Rationality: Psychology of Behavioral Economics. *The American Economic Review*, 93(5, December 2003): 1449-1475.
- Kahneman, D. 2003b. A perspective on judgment and choice Mapping bounded rationality. *American Psychologist*, 58(9): 697-720.

- Kahneman, D. 2010. Strategic decisions: When can you trust your gut?, *McKinsey Quarterly*, Vol. March: McKinsey Quarterly.
- Kahneman, D., & Klein, G. A. 2009. Conditions for Intuitive Expertise. *American Psychologist*, 64(6): 515-526.
- Kahneman, D., Slovic, P., & Tversky, A. (Eds.). 1982. Judgment Under Uncertainty: Heuristics and Biases. Cambridge: Cambridge University Press.
- Kahneman, D., & Tversky, A. 1982. Intuitive prediction: Biases and corrective procedures, Judgment under uncertainty: Heuristics and biases. New York: Cambridge University Press.
- Keen, P. G. W. 1973. *The implications of cognitive style for individual decision making*. Doctoral dissertation, Harvard University Graduate School of Business
- Khatri, N., & Ng, H. A. 2000. The Role of Intuition in Strategic Decision Making. *Human Relations*, 53(1): 57-86.
- Khurana, A., & Rosenthal, S. R. 1997. Integrating the Fuzzy Front End of New Product Development. *Sloan Management Review*(Winter): 103-120.
- Khurana, A., & Rosenthal, S. R. 1998. Towards Holistic "Front Ends" In New Product Development. Journal of Product Innovation Management, 15: 57-74.
- Kim, J., & Wilemon, D. 2002a. Focusing the Fuzzy Front-end in New Product Development. *R&D Management*, 32(4): 269-279.
- Kim, J., & Wilemon, D. 2002b. Strategic issues in managing innovation's fuzzy front end. *European Journal of Innovation Management*, 5(1): 27-39.
- Kim, W. C., & Maubourgne, R. 1998. Procedural Justice, Strategic Decision Making, and The Knowledge Economy. *Strategic Management Journal*, 19: 323-338.
- King, L., & Appleton, J. V. 1997. Intuition: A critical review of the research and rhetoric. *Journal of Advanced Nursing*, 26: 194-202.
- Kleinschmidt, E. J., De Brentani, U., & Salomo, S. 2007. Performance of Global New Product Development Programs: A Resource-Based View. Journal of Product Innovation Management, 24: 419-441.
- Koen, P., Ajamian, G., Burkart, R., Clamen, A., Davidson, J., D'Amore, R., Elkins, C., Herald, K., Incorvia, M., Johnson, A., Karol, R., Seibert, R., Slavejkov, A., & Wagner, K. 2001. Providing Clarity and a Common Language to the "Fuzzy Front End". *Research Technology Management*(March-April): 46-55.
- Kogut, B., & Zander, U. 1992. Knowledge of the Firm, Combinative Capabilities, And The Replication Of Technology. *Organization Science*, 3(3): 383-397.

- Koriat, A. 1993. How do we know that we know? The accessibility model of the feeling of knowing. *Psychological review*, 100: 609-639.
- Krishnan, V., & Ulrich, K. T. 2001. Product Development Decisions: A Review of the Literature. *Management Science*, 47(1): 1-21.
- Lamond, D., & Thompson, C. 2000. Intuition and Analysis in Decision Making and Choice. Journal of Nursing Scholarship, 32(3): 411-414.
- Leifer, R., McDermott, C., O'Conner, G., Peters, L. S., Rice, M., & Veryzer, R. V. 2000. *Radical Innovation*. Boston: Harvard Business School Press.
- Leners, D. W. 1992. Intuition in Nursing Practice. *Journal of Holistic Nursing*, 10: 137-153.
- Leybourne, S., & Sadler-Smith, E. 2006. The role of intuition and improvization in project management. *International Journal of Project Management*(24): 483-492.
- Liberatone, M. J., & Stylianou, A. C. 1995. Expert support systems for new product development decision making: a modeling framework and applications. *Management Science*, 41: 1296-1316.
- Lieberman, M. D. 2000. Intuition: A social cognitive neuroscience approach. *Psychological Bulletin*, 126: 109-137.
- Magee, J. F. 1964. Decsion trees for decision making. *Harvard Business Review*, 42(4): 126-138.
- March, J. G. 1994. *A Primer on Decision Making: How Decisions Happen*. New York: The Free Press.
- Matzler, K., Bailom, F., & Mooradian, T. A. 2007. Intuitive Decision Making. *MIT Sloan Management Review*, 49(1): 12-15.
- McCarthy, I. P., Tsinopoulos, C., Allen, P., & Rose-Anderssen, C. 2006. New Product Development as a Complex Adaptive System of Decisions. *Journal of Product Innovation Management*, 23: 437-456.
- McGrath, R. G. 2001. Exploratory Learning, Innovative Capacity and Managerial Oversight. *Academy of Management Journal*, 44(1): 118-131.
- Meehl, P. E. 1954. *Clinical Versus Statistical Prediction*. Minneapolis: University of Minnesota Press.
- Meehl, P. E. 1965. Clinical versus statistical prediction. *Journal of Experimental Research in Personality*, 63(1): 81-97.
- Miles, M. B., & Huberman, A. M. 1994. *An Expanded Sourcebook: Qualitative Data Analysis* (2 ed.). Thousand oaks: Sage.

- Miller, C. C., & Ireland, D. R. 2005. Intuition in Strategic Decision Making: Friend or Foe in the Fast-Paced 21st Century? *Academy of Management Executive*, 19(1): 19-30.
- Mintzberg, H. 1973. *The Nature of Managerial Work*. New York: Harper&Row.
- Mintzberg, H., Raisinghani, D., & Théorêt, A. 1976. The Structure of "Unstructured" Decision Processes. *Administrative Science Quarterly*, 21: 246-275.
- Mitleton-Kelly, E. (Ed.). 2003. Complex Systems and Evolutionary Perspectives on Organizations. The Application of Complexity Theory to Organizations. Amsterdam: Pergamon.
- Montoya-Weiss, M. M., & Calantone, R. J. 1994. Determinants of new product performance: A review and meta analysis. *Journal of Product Innovation Management*(11): 397-417.
- Myers, D. G. 2004. *Intuition: Its Powers and Perils*. New Haven: Yale University Press.
- Nobelius, D., & Trygg, L. 2002. Stop chasing the Front End process management of the early phases in product development projects. *International Journal of Project Management*(20): 331-340.
- Nonaka, I., & Takeuchi, H. 1995. *The Knowledge-Creating Company: How Japanese Companies Create the Dynamics of Innovation*. New York: Oxford University Press.
- Nummenmaa, L. 2004. *Käyttäytymistieteiden tilastolliset menetelmät*. Helsinki: Kustannusosakeyhtiö Tammi.
- Nutt, P. C. 1984. Types of Organizational Decision Processes. *Administrative Science Quarterly*, 29(September): 414-450.
- O'Connor, G. C., & Veryzer, R. V. 2001. The Nature of Market Visioning for Technology Based Radical Innovation. *Journal of Product Innovation Management*, 18(4): 231-246.
- Oakey, R. 1991. High technology small firms; their potential for rapid industrial growth. *International Small Business Journal*, 9: 30-42.
- Ortt, J. R., & Smits, R. 2006. Innovation management: different approaches to cope with the same trends. *Int. J. Technology Management*, 34(3/4): 296-318.
- Osareh, F. 1996. Bibliometrics, citation analysis and co-citation analysis: A review of literature 1. *Libri*, 46(3): 149-158.
- Ozer, M. 1999. A Survey of New Product Evaluation Models. *Journal of Product Innovation Management*(16): 77-94.
- Pacini, R., & Epstein, S. 1999. The relation of rational and experiential information processing styles to personality, basic beliefs, and the

ratio-bias phenomenon. *Journal of Personality and Social Psychology*, 76(6): 972-987.

- Papadakis, V., Lioukas, S., & Chambers, D. 1998. Strategic Decision-Making Processes: The Role of Management and Context. *Strategic Management Journal*, 19: 115-147.
- Parikh, J. 1994. *Intuition: The new frontier of management*. Oxford: Blackwell Business.
- Park, M. H., Lim, J. W., & Birnbaum-More, P. H. 2009. The Effect of Multiknowledge Individuals on Performance in Cross-Functional New Product Development Teams. *Journal of Product Innovation Management*, 26: 86-96.
- Perminova, O., Gustafsson, M., & Wikström, K. 2008. Defining uncertainty in projects - a new perspective. *International Journal of Project Management*, 26: 73-79.
- Polanyi, M. 1999. Intuition. In M. Runco, & S. R. Pritzker (Eds.), *Encyclopedia of creativity*, Vol. 2: 89-93. San Diego, CA: Academic Press.
- Porter, M. 1980. Competive Strategy: Technigues for Analysing Industries and Competitors. New York: Free Press.
- Porter, M. 1990. *The Competitive Advantage of Nations*. London: Macmillan.
- Poskela, J. 2009. Management Control in the Front End of Innovation. Doctoral Dissertation, Helsinki University of Technology, Department of Industrial Engineering and Management, Doctoral Dissertation Series 2009/11. Espoo, Finland: Helsinki University of Technology.
- Poskela, J., & Martinsuo, M. 2009. Management Control and Strategic Renewal in the Front End of Innovation. *Journal of Product Innovation Management*, 26: 671-684.
- Prahalad, C. K., & Hamel, G. 1990. The Core Competence of the Corporation. *Harvard Business Review*, 68(3): 79-91.
- Pretz, J. E., & Totz, K. S. 2007. Measuring individual differences in affective, heuristic, and holistic intuition. *Personality and Individual Differences*, 43: 1247-1257.
- Raami, A., Mielonen, S., & Keinänen, M. 2010. **Designers' Experiences** of Intuition: Coaching intuitive skills as a part of creative design process. Helsinki: University of Art&Design, Media Lab.
- Reid, S. E., & de Brentani, U. 2004. The Fuzzy Front End of New Product Development for Discontinuous Innovations: A Theoretical Model. *Journal of Product Innovation Management*, 21: 170-184.
- Reinertsen, D. G. 1985. Blitzkrieg product development: Cut development time in half. *Electronic Business*, January(15).

- Reinertsen, D. G. 1999. Taking the Fuzziness Out of the Fuzzy Front End **Research-Technology** Management, 42(November-December): 25-31.
- Rew, L. 2000. Acknowledging intuition in clinical decision making. *Journal of Holistic Nursing*, 18: 94-108.
- Rogers, E. M. 1995. *Diffusion on Innovations* (4 ed.). New York: The Free Press.
- Rosenau Jr., M. D., Griffin, A., Castellion, G., & Anschuetz, N. 1996. *The PDMA handbook of New Product Development*. New York: John Wiley & Sons.
- Rothwell, R. 1983. Innovation and firm size: a case of dynamic complementarity. *Journal of General Management*, 8(3): 362-370.
- Rothwell, R. 1992. Successful Industrial Innovation: Critical Success Factors for the 1990s. *R&D Management*, 22(3): 221-239.
- Rothwell, R. 1993. SMEs: their role in industrial and economic change. *International Journal of Technology Management*(Special issue on small firms): 8-22.
- Rothwell, R., & Gardiner, P. 1985. Invention, innovation, re-innovation and the role of the user: A case study of British hovercraft development. *Technovation*, 3(3): 167-186.
- Saarinen, E., & Hämäläinen, R. P. 2004. Systems Intelligence: Connecting Engineering Thinking with Human Sensitivity. In E. Saarinen, & R. P. Hämäläinen (Eds.), Systems Intelligence: Discovering a Hidden Competence in Human Action and Organizational Life: 9-38. Espoo: Helsinki University of Technology, Systems Analysis Laboratory Research Report A88.
- Saaty, T. L. 1977. A Scaling method for priorities in hierarchical structures. *Journal of Mathematical Psychology*(15): 234-281.
- Sadler-Smith, E., & Shefy, E. 2004. The Intuitive Executive: Understanding and Applying "gut feel" in Decision-Making. Academy of Management Executive, 18(4): 76-91.
- Schildt, H. A. 2005. SITKIS: Software for Bibliometric Data Management and Analysis, v1.5 ed. Helsinki: Helsinki University of Technology: Institute of Strategy and International Business.
- Schmidt, J. B., & Calantone, R. 1998. Are really new product development projects harder to shut down? *Journal of Product Innovation Management*, 15(2): 111-123.
- Schmidt, J. B., Montoya-Weiss, M. M., & Massey, A. P. 2001. New product development decision-making effectiveness: comparing individuals, face-to-face teams, and virtual teams. *Decision Sciences*, 32(4): 575-600.

- Schoemaker, P. 1993. Strategic Decisions in Organizations: Rational and Behavioural Views. *Journal of Management Studies*, 30(1): 107-129.
- Schumpeter, J. 1934. *The Theory of Economic Development*. Cambridge: Harvard University Press.
- Schumpeter, J. 1939. Business Cycles: A Theoretical, Historical, and Statistical Analysis. New York: McGraw-Hill.
- Scott, S. G., & Bruce, R. A. 1995. Decision-Making Style: The Development and Assessment of A New Measure. *Educational and Psychological Measurement*, 55(5): 818-831.
- Shamian, J. 1991. Effect of teaching decision analysis on student nurses' clinical intervention decision making. *Research in Nursing and Health*, 14: 59-66.
- Shapiro, S., & Spence, M. T. 1997. Managerial Intuition: A Conceptual and Operational Framework. *Business Horizons*(January-February): 63-68.
- Shirley, D. A., & Langan-Fox, J. 1996. Intuition: A review of the literature. *Psychological Reports*, 79: 563-584.
- Simon, H. A. 1987. Making Management Decisions: The Role of Intuition and Emotion. Academy of Management Executive, February 1987: 57-64.
- Sinclair, M., & Ashkanasy, N. M. 2005. Intuition: Myth or a Decision-Making tool? *Management Learning*, 36(3): 353-370.
- Sinclair, M., Sadler-Smith, E., & Hodgkinson, G. P. 2009. The Role of Intuition in Strategic Decision Making. In L. A. Costanzo, & R. B. MacKay (Eds.), *The Handbook of Research on Strategy and Foresight*. Cheltenham: Edward Elgar Publishing.
- Sloman, S. 1996. The Empirical Case for Two Systems of Reasoning. *Psychological Bulletin*(119): 3-22.
- Sloman, S. A. 2006. Two Systems of Reasoning. In T. Gilovich, D. Griffin, & D. Kahneman (Eds.), *Heuristics and Biases: The Psychology* of Intuitive Judgment: 379-396. New York: Cambridge University Press.
- Small, H. 1973a. Co-citation in the scientific literature. A new measure of the relationship between two documents. Journal of the American Society for Information Science(July-August): 265-269.
- Small, H. 1973b. Co-citation in the Scientific Literature: A New Measure of the Relationship Between Two Documents. Journal of the American Society for Information Science, 24(4): 265-269.
- Smiths, R. 2002. Innovation Studies in the 21st Century: Questions from a user's perspective. *Technological Forecasting & Social Change*, 69: 861-883.

- Smits, R. 2002. Innovation Studies in the 21st Century: Questions from a user's perspective. *Technological Forecasting & Social Change*, 69: 861-883.
- Souder, W. E., & Mandakovic, T. 1986. R&D project selection models. *Research Management*(29): 36-42.
- SPSS. PASW Statistics, 18 ed.: SPSS Inc.
- Stevens, G. A., & Burley, J. 2003. Piloting the Rocket of Radical Innovation. *Research-Technology Management*(March-April): 16-25.
- Strauss, A., & Corbin, J. 1998. Basics of Qualitative Research; Techniques and Procedures for Developing Grounded Theory (2 ed.): SAGE Publications Ltd.
- Teece, D. J. 2007. Explicating Dynamic Capabilities: The Nature and Microfoundations of (Sustainable) Enterprice Performance. *Strategic Management Journal*, 28: 1319-1350.
- Teece, D. J., Pisano, G., & Shuen, A. 1997. Dynamic Capabilities and Strategic Management. Strategic Management Journal, 18(7): 509-533.
- Thamhain, H., & Wilemon, D. 1987. Building high performing engineering project teams. *IEEE Transactions on Engineering Management*, EM-34(3): 130-137.
- Thomke, S. H., & Reinertsen, D. G. 1998. Agile Product Development. *California Management Review*, 41(1): 8-30.
- Tidd, J., & Bessant, J. 2009. *Managing Innovation: Integrating Technological, Market and Organizational Change* (4 ed.). Chichester: John Wiley & Sons, Ltd.
- Trott, P. 2005. *Innovation Management and New Product Development* (3 ed.). Harlow: Prentice Hall.
- Tushman, M. L. 1977. Special Boundary Roles in the Innovation Process. Administrative Science Quarterly, 22: 587-605.
- Tushman, M. L., & Anderson, P. 1987. Technological discontinuities and organizational environments. Administrative Science Quarterly, 31(3): 439-465.
- Tushman, M. L., & Katz, R. 1980. External Communication and Project Performance: An Investigation into the Role of Gatekeepers. *Management Science*, 26(11): 1071-1085.
- Tversky, A., & Kahneman, D. 1974. Judgment Under Uncertainty: Heuristics and Biases. *Science*(185): 1124-1131.
- Tzokas, N., Hultink, E. J., & Hart, S. 2004. Navigating the new product development process. *Industrial Marketing Management*, 33: 619-626.

- Utterback, J. 1994. *Mastering the Dynamics of Innovation*. Boston, MA: Harvard Business School Press.
- Wally, S., & Baum, R. J. 1994. Personal and Structural Determinants of the Pace of Strategic Decision Making. Academy of Management Journal, 37(4): 932-956.
- Van de Ven, A. H. 1986. Central Problems in the Management of Innovation. *Management Science*, 32(5): 590-607.
- Van de Ven, A. H., Polley, D. G., Garud, R., & Venkataraman, S. 1999. *The Innovation Journey*. New York: Oxford University Press.
- Van Riel, A. C. R., Semeijn, J., Hammedi, W., & Henseler, J. 2011. Technology-based service proposal screening and decision-making effectiveness. *Management Decision*, 49(5): 762-783.
- Vaughan, F. E. 1979. Awakening Intuition. New York: Anchor Books.
- Weick, K. E. 1995. *Sensemaking in Organizations*. Thousand Oaks: Sage.
- Verganti, R. 1997. Leveraging on systemic learning to manage the early phases of product innovation projects. *R&D Management*, 27(4): 377-392.
- Wheelwright, S. C., & Clark, K. B. 1992. *Revolutionizing Product Development*. New York: The Free Press.
- Yahaya, S.-Y., & Abu-Bakar, N. 2007. New product development management issues and decision-making approaches. *Management Decision*, 45(7): 1123-1142.
- Zadeh, L. A. 1965. Fuzzy Sets. Information and Control, 8(3): 338-353.
- Zhang, Q., & Doll, W. J. 2001. The fuzzy front end and success of new product development: a causal model. *European Journal of Innovation Management*, 4(2): 95-112.
- Zirger, B. J., & Maidique, M. A. 1990. A Model of New Product Development: An Empirical Test. *Management Science*, 36(7): 867-883.

7 Appendices

7.1 Appendix A: Bibliometric analysis

To support the literature review and to verify that relevant theory discussions have been taken into account for this study a bibliometric analysis was done. A bibliometric analysis is a semi-automated mathematical methodology used to analyze scientific literature. It is based on the argument that citations and co-citations can be used as indications of present and past activities of scientific work (Garfield, 1983; Osareh, 1996; Small, 1973a). Citation analysis assumes that authors are citing those documents that they consider important in the development of their research. Therefore frequently cited documents are likely to have greater influence on the development of that discipline than those that are cited less frequently. Cocitation analysis records the number of papers that have cited particular pairs of documents.

A bibliometric analysis was done using SITKIS: Software for bibliometric data management and analysis (Schildt, 2005). The analyzed data was exported for further processing in UCINET (Borgatti et al., 2002) and for network drawing to NetDraw (Borgatti, 2002). The time of the analysis was January 2008.

SITKIS is software which uses data imported from ISI Web of Knowledge databases. In this bibliometric analysis the data only contains a social science citation index and used the ISI Web of Science database with the maximum time range available from Helsinki University of Technology (1986-2008).

The co-citation network that was used for the main analysis, and some keyword relatedness analysis was used to verify the most common keywords used in the research area. Co-citation is the frequency with which two documents are cited together by the later literature (Small, 1973b). To be strongly co-cited, a large number of authors must cite the two earlier papers. The measure is the degree of the relationship between papers as perceived by the population of citing authors. These patterns change over time as the subject field evolves. When two papers are frequently co-cited, they are also frequently cited individually as well. Frequently cited papers represent the key concepts or ideas in a research field. Co-cited patterns map the relationships between these key ideas.

7.1.1 Search patterns

First search pattern

The first search pattern was "decision making AND innovation management". The search pattern resulted 445 articles. The most cited articles were two seminal articles one from Grant on knowledge based theory of the firm (Grant, 1996) and another from Eisenhardt on dynamic capabilities (Eisenhardt & Martin, 2000).

A co-citation network analysis was done using the parameter for "minimum citations to reference" set as 20. This was done in order to make the network somehow usable, otherwise the results would have been unreadable. No normalization was used for network data creation.

A network graph was drawn using a tie strength value between nodes of 3. The value was set to 3 in order to highlight only the most relevant linkages. The plotted graph is shown in Figure 7.1. The size of the node represents the number of citations.

The most co-cited article was the article on absorptive capacity by Cohen and Levinthal (Cohen & Levinthal, 1990). Other articles with many cocitations were Eisenhardt's article on agency theory (Eisenhardt, 1989a), Nonaka and Takeuchi's book on knowledge creating company (Nonaka & Takeuchi, 1995), Rogers articles on diffusion of innovations (Rogers, 1995), Barney's article on firms' resources and competitive advantage (Barney, 1991), and Kogut and Zander's article on the knowledge of the firm(Kogut & Zander, 1992).



Figure 7.1: Co-citation network of "decision making AND innovation management"

A keyword relatedness network was also processed with the same parameter settings as the co-citation network. The network is shown in Figure 7.2. Strong linkages between the keywords were obvious.



Figure 7.2: Keyword relatedness in search pattern "decision making AND innovation management"

Second search pattern

Second search pattern was "strategic decision making AND innovation management". The search pattern resulted in 122 articles. The most frequently cited article was again Eisenhardt's paper on dynamic capabilities (Eisenhardt & Martin, 2000).

A co-citation network analysis was done using the parameter for "minimum citations to reference" set as 10. No normalization was used for network data creation.

A network graph was drawn using a tie strength value between nodes of 1. The plotted graph is shown in Figure 7.3.

Appendices

The most co-cited article was Barney's article on firms resources and competitive advantage which was among the top co-cited articles in search pattern "decision making AND intuition" (Barney, 1991). The new articles and books not mentioned in the first search pattern are Porter's seminal book on competitive strategy (Porter, 1980), Prahalad and Hamel's article on core competences (Prahalad & Hamel, 1990), Hambrick and Mason's article on upper echelons theory (Hambrick & Mason, 1984) and Bantel and Jackson's paper on top management teams (Bantel & Jackson, 1989).



Figure 7.3: Co-citation network of "strategic decision making AND innovation management"

Third search pattern

The third search pattern was "decision making AND intuition". The search pattern resulted 237 articles. The most cited articles was Kahneman's article on judgment and choice (Kahneman, 2003b).

A co-citation network analysis was done using the parameter for "minimum citations to reference" set as 10. No normalization was used for network data creation.

A network graph was drawn using a tie strength value between nodes of 1. The plotted graph is shown in Figure 7.4.

The most co-cited article was Benner and Tanner's article on intuition use by expert nurses (Benner & Tanner, 1987). The other co-cited article was Tversky and Kahneman's article on judgment under uncertainty (Tversky & Kahneman, 1974).



Figure 7.4: Co-citation network of "decision making AND intuition"

7.1.2 Summary

Many of the most recognized studies about competing in the market have been done by Michael Porter. Porter's book on competitive strategy (Porter,

Appendices

1980) highlights the importance of innovations when competing in the marketplace.

Prahalad and Hamel's article on core competences (Prahalad & Hamel, 1990) describes the competences inside the company as one of the key differentiators between successful and unsuccessful companies. Innovations and innovation management can be one of those core competencies.

The absorptive capacity of the companies has been researched by Cohen and Levinthal (Cohen & Levinthal, 1990). Barney's article on firms' resources and competitive advantage (Barney, 1991), and Kogut and Zander's article on knowledge of the firm (Kogut & Zander, 1992) as well as Nonaka's research on knowledge creating companies (Nonaka & Takeuchi, 1995) all touch on the way in which companies create and absorb a competitive advantage by creating new knowledge and innovations.

These paths have been further studied by Grant in his knowledge based theory of the firm (Grant, 1996) and Teece (Teece, 2007; Teece et al., 1997) as well as by Eisenhardt and Martin in their dynamic capabilities research (Eisenhardt & Martin, 2000).

The bibliometric analysis did not produce any surprising results. It further strengthened the importance and relevance of the key documents and discussions in innovation management, strategic decision making, decision making and the intuition field. The results also revealed that a considerable part of intuitive decision making research that is done is related to nursing and medicine.

The limitations of this bibliometric analysis are mostly related to the bibliometric data used in the analysis. The database that was used (ISI Web of Knowledge) is very extensive, but it includes only one set of literature for the whole research area. The marginal publications are not typically present in ISI. However the ISI database includes all of the major quality publications and the results can be considered trustworthy.

7.2 Appendix B: Survey questionnaire used to collect data on experienced and inexperienced decision makers' use of intuition

Survey questionnaire modified based on Epstein et al. (Epstein et al., 1996).

Completely₂ Completely 3 Δ false 1 true 5 () () () ()()1. I have a logical mind 2. I prefer complex problems to simple problems 3. I believe in trusting my hunches 4. I am not a very analytical thinker 5. I trust my initial feelings about people 6. I try to avoid situations that require thinking in depth about somethina 7. I like to rely on my intuitive impressions 8. I don't reason well under pressure 9. I don't like situations in which I have to rely on intuition 10. Thinking hard and for a long time about something gives me little satisfaction 11. Intuition can be a very useful way to solve problems 12. I would not want to depend on anyone who described himself or herself as intuitive 13. I am much better at figuring things out logically than most people 14. I usually have clear, explainable reasons for my decisions 15. I don't think it is a good idea to rely on one's intuition for important decisions 16. Thinking is not my idea of an enjoyable activity 17. I have no problem thinking things through carefully 18. When it comes to trusting people, I can usually rely on my gut feelinas 19. I can usually feel when a person is right or wrong, even if I can't explain how I know 20. Learning new ways to think would be very appealing to me 21. I hardly ever go wrong when I listen to my deepest gut feelings to find an answer 22. I think it is foolish to make important decisions based on feelings 23. I tend to use my heart as a guide for my actions 24. I often go by my instincts when deciding on a course of action 25. I'm not that good at figuring out complicated problems 26. I enjoy intellectual challenges 27. Reasoning things out carefully is not one of my strong points 28. I enjoy thinking in abstract terms 29. I generally don't depend on my feelings to help me make decisions 30. Using logic usually works well for me in figuring out problems in my life 31. I think there are times when one should rely on one's intuition 32. I don't like to have to do a lot of thinking 33. Knowing the answer without having to understand the reasoning behind it is good enough for me

34. Using my gut feelings usually works well for me in figuring out prob-

lems in my life

35. I don't have a very good sense of intuition

36. If I were to rely on my gut feelings, I would often make mistakes

37. I suspect my hunches are inaccurate as often as they are accurate

38. My snap judgments are probably most as good as most people's

39. I am not very good at solving problems that require careful logical analysis

40. I enjoy solving problems that require hard thinking

41. What is your accumulated working experience?

42. How many years have you been in your current organization?

43. What is your highest education?

- 44. What is your education discipline?
- 45. What is the area in which you are currently functioning?

46. What is your job title?



7.3 Appendix C: Coding network which resulted after the first encounter of the data

C1P3	C1P2	C1P1	C2P5	C2P4	C2P3	C2P2	C2P1	C3P5	C3P4	СЗРЗ	C3P2	C3P1	C4P6	C4P5	C4P4	C4P3	C4P2	C4P1	
×																			Cross func
×	×	×																	Criteria based
						×													Back-up
×																			Finding
							×												Need fro room
							×												Personal dedic
							×		×										Problem solving
		×					×												Technological
		×				×													Compl. Int.
				×															Forcing dec
					×	×													Cost reduction
	×		×																Sponsor need
							×												Innovativeness
×	×	×			×			×											Clear way
	×			×															Good relations
		×	×		×														Trends
×									х	х									Learning by doing
									х	Х									Simple
		×	х	×			×	×				×							Experience
		×			×		×					×							Analysis
×	×	×	×		×			×	×		×	×							Feeling
×					×				х	Х	×		×				×		Process
			х	×	×		×	×	х		×	×	×		×				Commersializ.
×	×	×	×	×			×						×		×				Compl. knowl.
			×					×	×	×	×								Ind. decision
			×			×			×										Prioritization
			×			×				×	×	×				×	×		Spontaneous
					×													×	Competng
					×			×	×	×	×	×	×	×	×	×	×	×	Productizing k.
	×	×	×		×			×	×	×	×			×		×	×	×	Group dec.
×						×				×	×	×			×	×		×	Common und.
×	×	×	×	×	×		×	×	×	×	×	×	×	×	×	×	×	×	Customer foc
			×			×	×			×					×		×		New tech
			×			×			×	×	×	×							Own core
			×	×	×		×			×						×	×	×	Enabling tech
												×			×			×	Expanding
			×	×	×					×	×		×				×		Changing env
			×	×	×				×	×	×	×	×		×		×	×	Productization
				×	×				×	×	×	×					×		Technology
×				×					×	×							×		Bringing new

7.4 Appendix D: Cross tabulation of the interviewed persons and the codes used in the analysis



ISBN 978-952-60-5029-4 ISBN 978-952-60-5030-0 (pdf) ISSN-L 1799-4934 ISSN 1799-4934 ISSN 1799-4942 (pdf)

Aalto University School of Science Department of Industrial Engineering and Management www.aalto.fi BUSINESS + ECONOMY

ART + DESIGN + ARCHITECTURE

SCIENCE +

CROSSOVER

DOCTORAL DISSERTATIONS